

FACULTY OF MATHEMATICS AND PHYSICS Charles University

DOCTORAL THESIS

Denys Bulavka

Algebraic Tools in Combinatorial Geometry and Topology

Departement of Applied Mathematics

Supervisor of the doctoral thesis: doc. RNDr. Martin Tancer, Ph.D Study programme: Computer Science Study branch: Theory of Computing, Discrete Models and Optimization

Prague 2023

I declare that I carried out this doctoral thesis independently, and only with the cited sources, literature and other professional sources. It has not been used to obtain another or the same degree.

I understand that my work relates to the rights and obligations under the Act No. 121/2000 Sb., the Copyright Act, as amended, in particular the fact that the Charles University has the right to conclude a license agreement on the use of this work as a school work pursuant to Section 60 subsection 1 of the Copyright Act.

In date

Author's signature

First, I would like to thank Martin Tancer for his guidance throughout this journey and for sharing with me his intuition as well as many interesting problems.

I would like to thank my collaborators, Mykhaylo Tyomkyn, Eran Nevo, Yuval Peled, Russ Woodroofe, Éric Colin de Verdière, and Niloufar Fuladi.

I would also like to express my gratitude to Eran Nevo for hosting me at Hebrew University; Russ Woodroofe for hosting me twice at the University of Primorska; Uli Wanger for hosting me at IST Austria; Éric Colin de Verdière for hosting me at Gustave Eiffel University; and Xavier Goaoc for hosting me at Lorraine University. I thoroughly enjoyed each of these visits.

I would also like to thank Michael Skotnica for helping me navigate any bureaucracy involving the Czech language; Petra Příhodová for all her infinite patience and help with all the paperwork during my doctoral studies; the PhD students from KAM+IUUK, without whom this journey would have been less fun.

Finally, I would like to say thank you to Sara for always being there for me and for her constant support. To my mother, who never stopped supporting me even with me being so far away. I wouldn't be where I am today if it wasn't for her.

The work on which this thesis is based was supported by the following projects:

- Grant no. 19-27871X of the Czech Science Foundation (GACR).
- Charles University project PRIMUS/21/SCI/014.
- Grant Schemes at CU, reg. no. CZ.02.2.69/0.0/0.0/19_073/0016935.
- SVV-2020-260578.
- Grant no. 21-32817S of the Czech Science Foundation (GACR).

Title: Algebraic Tools in Combinatorial Geometry and Topology

Author: Denys Bulavka

Department: Departement of Applied Mathematics

Supervisor: doc. RNDr. Martin Tancer, Ph.D, Departement of Applied Mathematics

Abstract: In this thesis we study combinatorial problems through the lenses of the exterior algebra. This algebra is a natural object to model set systems as well as simplicial complexes. Moreover, it is possible to translate the classical operations from the combinatorial setting to the setting of exterior algebra. For example, set intersection and the classical boundary map from topology. Often this point of view makes it possible to translate the combinatorial problem to a problem regarding the dimension of certain vector space. The latter one might turn up to be easier since we can study the dimension of a vector space with linear maps. We follow such approach in this thesis.

In the first part we study the weak saturation problem introduced by Bollobás in the 60's. This problem consists in, given a host graph F and a pattern graph H, to determine the minimum number of infected edges of F one has to start with in order for the infection to spread, according to the pattern H, to the whole host graph F. We study this problem when the host and the pattern are complete uniform multipartite hypergraphs.

Next, we work on a generalization of a theorem by Helly regarding intersecting patterns of convex sets. Concretely, given a finite family of convex sets in \mathbb{R}^d partitioned into d + 1 colors classes such that a fraction α of the colorful (d + 1)-tuples intersect, what is the size of the largest monochromatic intersecting subfamily that we can guaranty? We answer this question.

In the third part we study the notion of volume-rigidity for simplicial complexes. This is a generalization of (generic) rigidity for graphs to higher dimensional objects. We relate volume-rigidity with exterior algebraic shifting and show that compact surfaces of small genus without boundary are volume-rigid.

In the last part we study a generalization of a classical theorem by Erdős, Ko and Rado on the maximal size of a pairwise-intersecting family. Concretely we are interested in charaterizing the family achieving the maximal size when it is restricted to a simplicial complex.

Keywords: Helly theorem Erdős-Ko-Rado rigidity simplicial complex algebraic shifting exterior algebra weak saturation

Contents

1	Intr	oduction	2
2	Pre 2.1 2.2	liminariesSimplicial complexExterior algebra2.2.1Left interior product	4 4 5 6
3	Wea	ak saturation	9
	3.1	The upper bound	11
	3.2	Colorful exterior algebra.	12
	3.3	The lower bound	15
	3.4	Weak saturation in the clique	19
4	Col	orful Fractional Helly Theorem	23
	4.1	d-representable and d -collapsible complexes	25
		4.1.1 Kim's refined conjecture and further generalization	26
	4.2	Colorful exterior algebra	27
	4.3	\mathbf{k} -colorful fractional Helly theorem $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	29
	4.4	A topological version?	35
5	Volu	ıme rigidity	$ \begin{array}{c} 6 \\ 9 \\ $
		5.0.1 Main results	
		5.0.2 Relation to previous works.	40
	5.1	Volume rigidity and $\Delta^p(\cdot)$	40
		5.1.1 Basic properties of $\Delta^p(\cdot)$	40
	50	5.1.2 Volume rigidity and $\Delta^{p}(\cdot)$	41
	5.2	Volume rigidity, local moves and homology	42
	5.9	5.2.1 Volume rigidity of surfaces	44
	0.0 5 4	Concluding remarks	40
	0.4		47
6	Erd	ős-Ko-Rado for Simplicial Complexes	48
	6.1	Applications	51
	6.2	Preliminaries	55
		6.2.1 Combinatorial shifting	55
		6.2.2 <i>t</i> -fold near-cone	57
		6.2.3 Depth and the sequentially Cohen-Macaulay property	59 60
	C 9	b.2.4 Exterior algebraic shifting	60 60
	0.3 6 4	Strict Endőz Ko Dodo	03 66
	0.4 6 F	SURICI EROS-KO-KAGO	00 60
	0.0	Cross-intersecting families	00
Bi	bliog	raphy	70
\mathbf{Li}	st of	publications	76

1. Introduction

In combinatorics one is usually interested in counting, or enumerating, objects that satisfy certain properties. For example, what is the number of trees on n vertices? What is the largest size of a pairwise-intersecting family of k-subsets from a ground set with n elements? One of the possible ways to answer these and similar questions is with the aid of linear algebra. In this setting counting the desired quantity translates into computing the dimension of a certain vector space. It often turns out that the later one is easier to compute since one can study the dimension of a vector space through linear maps which capture the properties satisfied by the objects of interest.

In this thesis we will mainly work with the exterior algebra of a vector space since, among other properties, it is a natural framework to model the set intersection property as well as the classical boundary map from topology. Moreover, one can associate elements of a set family with basis elements in this algebra in a very simple way. This process associates a combinatorial object with an algebraic object. We will be also interested in the opposite direction, that is from a vector space we obtain a combinatorial object which is in some sense, to be made precise later, the simplest object satisfying the desired properties. This last part refers to a compression operation, so called algebraic shifting, introduced by Kalai [50] and it turns out that one can read structural and geometric information from the compressed object.

Results of the thesis. Bollobás [15] introduced the problem of weak saturation on graphs. Given a pattern graph H and a host graph F, the objective is to determine the minimum number of infected edges that one has to start with in order to propagate the infection according to H to all the edges of F. In Chapter 3 we solve this problem for the case when the pattern and the host graphs are complete uniform multipartite hypergraphs.

The celebrated theorem of Helly [38] states that every finite family of convex sets in \mathbb{R}^d with every (d + 1)-tuple of its elements having non-empty intersection satisfies the condition that all of its members have non-empty intersection. Kalai [46] and Eckhoff [26] showed independently the optimal size of an intersecting subfamily if one has only that a certain fraction of the (d + 1)-tuples is intersecting. In Chapter 4 we extend this result for the case of colorful families of convex sets.

Asimow and Roth [5, 6] introduced the notion of graph rigidity. Lee [57, 67] showed that the membership of a certain edge in the symmetric algebraic shift of the corresponding graph is equivalent with the graph being (generically) rigid. In Chapter 5 we establish an analogous link between volume-rigidity and the membership of a face in the exterior algebraic shifting of the corresponding simplicial complex. Moreover, we show that any triangulation of a small genus surface is volume-rigidity.

What is the largest cardinality of a family of pairwise-intersecting sets? The celebrated theorem of Erdős, Ko and Rado [28] states that the family achieving the maximal size is given by subsets all containing a fixed element. In Chapter 6 we show that the statement is still true when the pairwise-intersecting family is restricted to a sequentially Cohen-Macaulay near-cone. This family of simplicial

complexes is quite large, for example it contains the simplicial complex which faces are given by the independence sets of a chordal graph that contains an isolated vertex.

Bibliographic remarks. The results of this thesis have been published or are being considered for publication. The relation of the chapters with the published work is as follows. In Chapter 2 we establish the necessary preliminaries about simplicial complexes and exterior algebra. This chapter is based on the respective section from the articles [19, 22]. Chapter 3 is based on a joint work with Mykhaylo Tyomkyn and Martin Tancer [22]. Chapter 4 is based on a joint work with Afshin Goodarzi and Martin Tancer [19]. Chapter 5 is based on a joint work with Eran Nevo and Yuval Peled [20]. Chapter 6 is based on a joint work with Russ Woodroofe and some of the results appeared in [21]. The presentation of this last chapter is different from the already published work and it contains unpublished results which will be part of the full version.

2. Preliminaries

2.1 Simplicial complex

A (finite abstract) simplicial complex is a set system K on a finite set of vertices N such that whenever $A \in K$ and $B \subseteq A$, then $B \in K$. The elements of N represent vertices and we will typically denote them by letters such as v or w. The elements of K are faces (a.k.a., simplices) of K. The dimension of a face $A \in K$ is defined as dim A = |A| - 1; this corresponds to representing A as an (|A| - 1)-dimensional simplex. The dimension of K, denoted dim K, is the maximum of the dimensions of faces in K. A face of dimension k is a k-face in short. Vertices of K are usually identified with 0-faces, that is, $v \in N$ is identified with $\{v\} \in K$. (Though the definition of simplicial complex allows that $\{v\} \notin K$ for $v \in N$, in our applications we will always have $\{v\} \in K$ for $v \in N$.) The k-skeleton of K, denoted by $K^{(k)}$, is the simplicial complex formed by the set of faces from K whose dimension is at most k. The f-vector of K is defined by $f(K) = (f_{-1}(K), f_0(K), \ldots, f_d(K))$ where $f_k(K)$ is the number of k-faces in K with the convention that $f_{-1}(K) = 1$ if K is not the empty complex.

Example 1. Let K be the simplicial complex depicted in Figure 2.1, that is

$$K = \{ \\ \emptyset, \\ \{a\}, \{b\}, \{c\}, \{d\}, \{e\}, \{f\} \text{ }0\text{-}faces, a.k.a., vertices,} \\ \{a, b\}, \{a, d\}, \{b, c\}, \{b, d\}, \{c, d\}, \{c, e\} \text{ }1\text{-}faces, a.k.a., edges,} \\ \{a, b, d\} \text{ }2\text{-}faces \\ \}.$$

Since the simplicial complex is determined by its maximal faces, called facets, we will say that the simplicial complex is generated by its set of facets. In this example, K is generated by $\{\{f\}, \{b, c\}, \{c, d\}, \{c, e\}, \{a, b, d\}\}$. Its f-vector is f(K) = (1, 6, 6, 1).



Figure 2.1: Example of a simplicial complex.

2.2 Exterior algebra

Let N be a set of size n, ordered with a total order < and let $V = \mathbb{R}^N$ be an n-dimensional real vector space with a basis $(e_v)_{v \in N}$. The *exterior algebra* of V, denoted by $\wedge V$, is a 2^n -dimensional vector space with basis $(e_S)_{S \subseteq N}$ and an associative bilinear product operation, denoted by \wedge , that satisfies

- (i) e_{\emptyset} is the neutral element, i.e., $e_{\emptyset} \wedge e_S = e_S e_{\emptyset}$;
- (ii) $e_S = e_{s_1} \wedge \cdots \wedge e_{s_k}$ for $S = \{s_1 < \cdots < s_k\} \subseteq N;$
- (iii) $e_v \wedge e_w = -e_w \wedge e_v$ for all $v, w \in N$.

For $0 \leq k \leq n$ we denote by $\bigwedge^k V$ the subspace of $\bigwedge V$ with basis $(e_S)_{S \in \binom{N}{k}}$. Denote by $\langle \cdot, \cdot \rangle$ the standard inner product (dot product) on V as well as on $\bigwedge V$ with respect to the basis $(e_v)_{v \in N}$ and $(e_S)_{S \subseteq N}$ respectively; that is, for every pair of sets $S, T \subseteq N$, the inner product $\langle e_S, e_T \rangle$ is 1 if S = T and 0 otherwise.

If $(f_v)_{v \in N}$ is another basis of V, then $(f_S)_{S \subseteq N}$ is a new basis of $\bigwedge V$, where f_S stands for $f_{s_1} \land \cdots \land f_{s_k}$ for $S = \{s_1 < \cdots < s_k\} \subseteq N$. Similarly, $(f_S)_{S \in \binom{N}{k}}$ is a basis of $\bigwedge^k V$ for $k \in \{0, \ldots, n\}$. The formulas (i), (ii) and (iii) remain valid for the basis $(f_v)_{v \in N}$ due to the definition of f_S and bilinearity of \land .

In particular, $\bigwedge V$ and $\bigwedge^k V$ do not depend on the initial choice of the basis. Using (ii) and (iii) iteratively, for $S, T \subseteq N$ we get

$$f_S \wedge f_T = \begin{cases} \operatorname{sgn}(S,T) f_{S \cup T} & \text{if } S \cap T = \emptyset \\ 0 & \text{if } S \cap T \neq \emptyset, \end{cases}$$
(2.1)

where $\operatorname{sgn}(S, T)$ is the sign of the permutation of $S \cup T$ obtained by first placing the elements of S (in our total order <) and then the elements of T. Equivalently, $\operatorname{sgn}(S,T) = (-1)^{\alpha(S,T)}$ where $\alpha(S,T) = |\{(s,t) \in S \times T : t < s\}|$ is the number of transpositions.

Let $A = (a_{vw})_{v,w\in N}$ be the transition matrix from $(e_v)_{v\in N}$ to $(f_v)_{v\in N}$, meaning that $f_v = \sum_{w\in N} a_{vw}e_w$. Then, for $S \subseteq N$ of size k, f_S can be expressed as

$$f_S = \sum_{T \in \binom{N}{k}} \det(A_{S|T}) e_T, \qquad (2.2)$$

where $A_{S|T}$ is the submatrix of A formed by rows in S and columns in T, i.e., $A_{S|T} = (a_{vw})_{v \in S, w \in T}$.

As noted in [46], it follows from the Cauchy-Binet formula that if the basis $(f_v)_{v \in N}$ is orthonormal then $(f_S)_{S \subseteq N}$ is orthonormal as well. For completeness, we provide a short explanation. Let $S, L \subseteq N$ be a pair of subsets. If $|S| \neq |L|$, then f_S and f_L belong to two orthogonal subspaces of $\bigwedge V$, namely $\bigwedge^{|S|} V$ and $\bigwedge^{|L|} V$, and so $\langle f_S, f_L \rangle = 0$. On the other hand, if |S| = |L| =: k, then by writing f_S and f_L in the standard basis $(e_T)_{T \subseteq N}$ we have that

$$\langle f_S, f_L \rangle = \sum_{T \in \binom{N}{k}} \det(A_{S|T}) \det(A_{L|T}^t) = \det(A_{S|N} A_{L|N}^t),$$

where B^t stands for the transpose matrix of B (and expressions like $A^t_{L|T}$ stand for $(A_{L|T})^t$), and the last equality holds by the Cauchy-Binet formula (see e.g. Section

1.2.4 of [37]). Notice that for any $u \in S$ and $w \in L$ we have $(A_{S|N}A_{L|N}^t)_{u,w} = \langle f_u, f_w \rangle$, and since $(f_v)_{v \in N}$ is orthonormal this is 1 if u = w and 0 otherwise. Therefore, if S = L, the product $A_{S|N}A_{L|N}^t$ is the identity matrix and consequently the determinant will be 1. On the other hand, if $S \neq L$, the product $A_{S|N}A_{L|N}^t$ will have a zero column, and so the determinant will be 0. The above claim follows.

We say that the change of basis from $(e_v)_{v \in N}$ to $(f_v)_{v \in N}$ is generic if $\det(A_{S|T}) \neq 0$ for every $S, T \subseteq N$ of the same size; that is, every square submatrix of A has full rank. It is known (see e.g. [46, Section 2]) that $(f_v)_{v \in N}$ can be chosen to be both generic and orthonormal. For a basis $(f_v)_{v \in N}$ generic with respect to $(e_v)_{v \in N}$ and a pair of sets $S, T \in {N \choose k}$ we have

$$\langle f_S, e_T \rangle \stackrel{(2.2)}{=} \langle \sum_{T' \in \binom{N}{k}} \det(A_{S|T'}) e_{T'}, e_T \rangle = \sum_{T' \in \binom{N}{k}} \det(A_{S|T'}) \langle e_{T'}, e_T \rangle = \det A_{S|T} \neq 0.$$

$$(2.3)$$

The following lemma is implicitly contained in [46].

Lemma 2. If the columns of an $m \times n$ matrix A are linearly independent, then the columns of $C_k(A)$ are linearly independent as well.

Proof. If columns of A are linearly independent, then $n \leq m$. Consider an arbitrary square submatrix B of rank n. Considering B as a transition matrix from $(e_i)_{i\in N}$ to $(f_i)_{i\in N}$, we get that $C_k(B)$ is a transition matrix from $(e_S)_{S\in\binom{N}{k}}$ to $(f_S)_{S\in\binom{N}{k}}$, thus $C_k(B)$ has full rank. However, $C_k(B)$ is also a submatrix of $C_k(A)$ with all $\binom{n}{k}$ columns.

2.2.1 Left interior product

In this subsection we recall the *left interior product* $g \downarrow f$ of g and f. We refer to Section 2.2.6 of [72] for a more extensive coverage of the topic.

Lemma 3. For any $f, g \in \bigwedge V$ there exists a unique element $g \llcorner f \in \bigwedge V$ that satisfies

$$\langle h, g \llcorner f \rangle = \langle h \land g, f \rangle \text{ for all } h \in \bigwedge V.$$
 (2.4)

Furthermore, assuming $f \in \bigwedge^{s} V$ and $g \in \bigwedge^{t} V$, if t > s then $g \downarrow f = 0$, while if $t \leq s$ then $g \downarrow f \in \bigwedge^{s-t} V$.

Proof. For $f, g \in \bigwedge V$ we set

$$g\llcorner f := \sum_{S \subseteq N} \langle e_S \land g, f \rangle e_S.$$

To verify that this satisfies (2.4) let $h \in \bigwedge V$ be arbitrary. By bilinearity of $\langle \cdot, \cdot \rangle$ and \wedge , and orthonormality of $(e_S)_{S \subseteq N}$ we have

$$\begin{split} \langle h, g \llcorner f \rangle &= \langle h, \sum_{S \subseteq N} \langle e_S \land g, f \rangle e_S \rangle = \sum_{S \subseteq N} \langle e_S \land g, f \rangle \langle h, e_S \rangle \\ &= \Big\langle \sum_{S \subseteq N} \langle h, e_S \rangle (e_S \land g), f \Big\rangle = \Big\langle \Big(\sum_{S \subseteq N} \langle h, e_S \rangle e_S \Big) \land g, f \Big\rangle \\ &= \langle h \land g, f \rangle. \end{split}$$

To show uniqueness, suppose that z is an element in $\bigwedge V$ that satisfies (2.4). Then for each $T \subseteq N$ we have

$$\langle e_T, z \rangle \stackrel{(2.4)}{=} \langle e_T \wedge g, f \rangle \stackrel{(2.4)}{=} \langle e_T, g \llcorner f \rangle.$$

Therefore z and $g \downarrow f$ are identical, as their inner products with all basis elements coincide.

Now assume that $f \in \bigwedge^s V$ and $g \in \bigwedge^t V$, and let $S \subseteq N$ be arbitrary. By (2.4) we have

$$\langle e_S, g \llcorner f \rangle = \langle e_S \land g, f \rangle.$$

Observe that $e_S \wedge g \in \bigwedge^{|S|+t}$ while $f \in \bigwedge^s V$ and these spaces are orthogonal unless |S| + t = s. Hence, $g \llcorner f = 0$ for t > s and $g \llcorner f \in \bigwedge^{s-t} V$ otherwise. \Box

It is straightforward to check from the definition that the left interior product is bilinear:

- $(f+g) \llcorner h = (f \llcorner h) + (g \llcorner h),$
- $f \llcorner (g+h) = (f \llcorner g) + (f \llcorner h),$

and satisfies

$$h\llcorner (g\llcorner f) = (h \land g)\llcorner f. \tag{2.5}$$

Lemma 4. Let $(f_v)_{v \in N}$ be an orthonormal basis of V. Then, for any $S, T \subseteq N$ we have

$$f_T \llcorner f_S = \begin{cases} \operatorname{sgn}(S \setminus T, T) f_{S \setminus T} & \text{if } T \subseteq S, \\ 0 & \text{otherwise.} \end{cases}$$

Proof. Put s := |S| and t := |T|. If t > s then by Lemma 3 we have $f_{T \sqcup} f_S = 0$ and the conclusion follows. So we may assume that $s \ge t$, and by the same lemma it follows that $f_{T \sqcup} f_S \in \bigwedge^{s-t} V$. Since the basis $(f_v)_{v \in N}$ is orthonormal, so is the basis $(f_L)_{L \in \binom{N}{s-t}}$ of $\bigwedge^{s-t} V$. Expressing $f_{T \sqcup} f_S$ in this basis and using (2.4), we obtain

$$f_T \sqcup f_S = \sum_{L \in \binom{N}{s-t}} \langle f_L, f_T \sqcup f_S \rangle f_L = \sum_{L \in \binom{N}{s-t}} \langle f_L \wedge f_T, f_S \rangle f_L$$

Due to (2.1) and orthonormality of $(f_v)_{v \in N}$ we have $\langle f_L \wedge f_T, f_S \rangle = 0$ unless $T \subseteq S$ and $L = S \setminus T$. Therefore, using (2.1) again we get

$$f_{T} \llcorner f_{S} = \begin{cases} \langle f_{S \backslash T} \land f_{T}, f_{S} \rangle f_{S \backslash T} = \operatorname{sgn}(S \setminus T, T) f_{S \backslash T} & \text{if } T \subseteq S, \\ 0 & \text{if } T \not\subseteq S. \end{cases}$$

Lemma 5. Let $(f_v)_{v \in N}$ be a generic orthonormal basis of V with respect to $(e_v)_{v \in N}$. For a pair of sets $T, R \subseteq N$ of sizes t and r, respectively, such that $r \geq t$ we have

$$f_T \llcorner e_R = \sum_{S \in \binom{N \setminus T}{r-t}} \lambda_S f_S,$$

where all the coefficients λ_S are non-zero.

Proof. By Lemma 3 we have that $f_T \llcorner e_R \in \bigwedge^{r-t} V$. Since $(f_S)_{S \in \binom{N}{r-t}}$ is an orthonormal basis of $\bigwedge^{r-t} V$, we can write

$$f_{T \sqcup} e_R = \sum_{S \in \binom{N}{r-t}} \langle f_S, f_T \llcorner e_R \rangle f_S.$$

Applying (2.4) and (2.1) gives

$$\langle f_S, f_T \llcorner e_R \rangle = \langle f_S \land f_T, e_R \rangle = \begin{cases} \pm \langle f_{S \cup T}, e_R \rangle & \text{if } S \cap T = \emptyset, \text{, i.e., if } S \in \binom{N \setminus T}{r-t}, \\ 0 & \text{otherwise.} \end{cases}$$

Setting $\lambda_S = \langle f_S \wedge f_T, e_R \rangle$ for $S \in \binom{N \setminus T}{r-t}$, we thus obtain

$$f_T \llcorner e_R = \sum_{S \in \binom{N \setminus T}{r-t}} \lambda_S f_S,$$

as claimed. In addition, since we assumed that $(f_v)_{v \in N}$ is generic with respect to $(e_v)_{v \in N}$, we have $\lambda_S = \pm \langle f_{S \cup T}, e_R \rangle \neq 0$ by (2.3) for all $S \in \binom{N \setminus T}{r-t}$.

3. Weak saturation

Let F and H be q-uniform hypergraphs (q-graphs for short); we identify hypergraphs with their edge sets. We say that a subgraph $G \subseteq F$ is weakly H-saturated in F if the edges of $F \setminus G$ can be ordered as e_1, \ldots, e_k such that for all $i \in [k]$ the hypergraph $G \cup \{e_1, \ldots, e_i\}$ contains an isomorphic copy of H which in turn contains the edge e_i . We call such e_1, \ldots, e_k an H-saturating sequence of G in F. The weak saturation number of H in F, weat(F, H) is the minimum number of edges in a weakly H-saturated subgraph of F. When F is complete of order n, we simply write weat(n, H).

Weak saturation was introduced by Bollobás [15] in 1968 and is related to (strong) graph saturation: G is H-saturated in F if adding any edge of $F \setminus G$ would create a new copy of H. However, a number of properties of weak saturation make it a more natural object of study. Firstly, it follows from the definition that any graph G achieving wsat(F, H) has to be H-free (we could otherwise remove an edge from a copy of H in G resulting in a smaller example), while for strong saturation H-freeness may or may not be imposed, resulting in two competing notions (see [64] for a discussion). Secondly, a short subadditivity argument originally due to Alon [2] shows that for every 2-uniform H, $\lim_{n\to\infty} wsat(n, H)/n$ exists. Whether the same holds for strong saturation is a longstanding conjecture of Tuza [83]. And thirdly, weak saturation lends itself to be studied via algebraic methods, thus offering insight into algebraic and matroid structures underlying graphs and hypergraphs.

The most natural case when F and H are cliques was the first to be studied. Let K_r^q denote the complete q-graph of order r. Confirming a conjecture of Bollobás, Frankl [34], and Kalai [47, 48] independently proved that wsat $(n, K_r^q) = \binom{n}{q} - \binom{n-r+q}{q}$. Another proof has been given by Alon [2] and in hindsight this conjecture could be also derived from an earlier paper of Lovász [60]. While the upper bound is a construction that is easy to guess (a common feature in weak saturation problems), all of the above lower bound proofs rely on algebraic or geometric methods, and no purely combinatorial proof is known to this date.

In the subsequent years weak saturation has been studied extensively [2, 85, 27, 69, 84, 64, 70, 73, 18, 74, 30, 8, 9, 63]. Despite this, our understanding of weak saturation numbers is still rather limited. For instance we do not know whether for $q \geq 3$ we have a similar limiting behavior as in the graph case, in that $\lim_{n\to\infty} \operatorname{wsat}(n, H)/n^{q-1}$ always exists; this has been conjectured by Tuza [85].

In this chapter we address the case when $H = K_{r_1,\ldots,r_d}^q$ is a complete *d*-partite *q*-graph for arbitrary $d \ge q > 1$. That is, V(H) is a disjoint union of sets R_1, \ldots, R_d with $|R_i| = r_i$ and

$$E(H) = \left\{ e \in \binom{V(H)}{q} : |e \cap R_i| \le 1 \text{ for all } i \in [d] \right\},\$$

in particular, for q = 2 we recover the usual complete multipartite graphs. This is perhaps the next most natural class of hypergraphs to consider after the cliques.

For the host graph F, besides the clique it is natural to consider a larger complete *d*-partite *q*-graph K_{n_1,\ldots,n_d}^q . In the latter case we have a choice between the *undirected* and *directed* versions of the problem. The former follows the definition of weak saturation given at the beginning, while in the latter we additionally impose that the new copies of H in F created in every step "point the same way", i.e. have r_i vertices in the *i*-th partition class for all $i \in [d]$ (see below for a formal definition).

All three above versions have been studied in the past. For q = 2, Kalai [48] determined wsat $(n, K_{r,r})$ for large enough n. Kronenberg, Martins and Morrison [54] recently extended it to wsat $(n, K_{r,r-1})$ and asymptotically to all wsat $(n, K_{s,t})$. No other values wsat $(n, K_{r_1,...,r_d}^q)$ are known except for $r_1 = \cdots = r_d = 1$ when H is a clique and a handful of closely related cases, e.g., when all r_i but one are 1 [70]. When both H and F are complete d-partite, for d = q Alon [2] solved the problem in the directed setting. Moshkovitz and Shapira [64], building on Alon's work, settled the undirected case, determining wsat $(K_{n_1,...,n_d}^d, K_{r_1,...,r_d}^d)$. There has been no progress for d > q.

In our main contribution in this chapter we settle completely the directed case for all q and d. To state the problem formally, let $\mathbf{r} = (r_1, \ldots, r_d)$ and $\mathbf{n} = (n_1, \ldots, n_d)$ be integer vectors such that $1 \leq r_i \leq n_i$. Suppose $N = N_1 \sqcup \cdots \sqcup N_d$ where $|N_i| = n_i$ and \sqcup denotes a disjoint union. Let $K_{\mathbf{n}}^q$ be the complete d-partite q-graph on N whose partition classes are the N_i , and let $K_{\mathbf{r}}^q$ be an unspecified complete d-partite q-graph on the same partition classes, with r_i vertices in each N_i . Given a subgraph G of $K_{\mathbf{n}}^q$, a sequence of edges e_1, \ldots, e_k in $K_{\mathbf{n}}^q$ is a (directed) $K_{\mathbf{r}}^q$ -saturating sequence of G in $K_{\mathbf{n}}^q$ if: (i) $K_{\mathbf{n}}^q \setminus G = \{e_1, \ldots, e_k\}$; (ii) for every $j \in [k]$ there exists $H_j \subseteq G \cup \{e_1, \ldots, e_j\}$ isomorphic to $K_{\mathbf{r}}^q$ such that $e_j \in H_j$ and $|V(H_j) \cap N_i| = r_i$ for all $i \in [d]$. The q-graph G is said to be (directed) weakly $K_{\mathbf{r}}^q$ -saturated in $K_{\mathbf{n}}^q$ if it admits a $K_{\mathbf{r}}^q$ -saturating sequence in the latter. The (directed) weak saturation number of $K_{\mathbf{r}}^q$ in $K_{\mathbf{n}}^q$, in notation $w(K_{\mathbf{n}}^q, K_{\mathbf{r}}^q)$, is the minimal number of edges in a weakly $K_{\mathbf{r}}^q$ -saturated subgraph of $K_{\mathbf{n}}^q$.

Theorem 6. For all $d \ge q \ge 2$, **n** and **r** we have

$$w(K_{\mathbf{n}}^{q}, K_{\mathbf{r}}^{q}) = \sum_{I \in \binom{[d]}{q}} \prod_{i \in I} n_{i} - \sum_{I \in \binom{[d]}{\leq q}} \prod_{i \in I} (n_{i} - r_{i}).$$

In the above formula $\binom{[d]}{\leq q}$ stands for the set of all subsets of [d] of size at most q, and we use the convention that $\prod_{i \in \emptyset} (n_i - r_i) = 1$.

As mentioned, the d = q case of Theorem 6 was proved by Alon [2]. Hence our result generalizes Alon's theorem to arbitrary $d \ge q$. When H is balanced, that is when $r_1 = \cdots = r_d$, there is no difference between the directed and undirected partite settings. Writing $K^q(r; d)$ for $K^q_{r,\ldots,r}$ (d times), Theorem 6 thus determines the weak saturation number of $K^q(r; d)$ in complete d-partite q-graphs.

Corollary 7. For all $d \ge q \ge 2$ and $n_1, \ldots, n_d \ge r \ge 1$ we have

$$\operatorname{wsat}(K_{n_1,\dots,n_d}^q, K^q(r; d)) = \sum_{I \in \binom{[d]}{q}} \prod_{i \in I} n_i - \sum_{I \in \binom{[d]}{\leq q}} \prod_{i \in I} (n_i - r).$$

Our proof of Theorem 6 combines exterior algebra techniques in the spirit of [48] with a new ingredient: the use of the colorful exterior algebra inspired by the recent work of Bulavka, Goodarzi and Tancer on the colorful fractional Helly theorem [19].

Kronenberg, Martins and Morrison ([54], Section 5) remarked that while the values wsat $(n, K_{t,t})$ and wsat $(K_{\ell,m}, K_{t,t})$ for $\ell + m = n$, which were determined in separate works, are of the same order of magnitude, it is not obvious if there is any direct connection. In our second contribution in this chapter we establish such a connection using a tensoring trick. As we have mentioned earlier, 2-graphs H satisfy wsat $(n, H) = c_H n + o(n)$, and Alon's proof of this fact [2] can be straightforwardly adjusted to show that wsat $(K_{n,n}, H) = c'_H \cdot 2n + o(n)$ when H is bipartite. We show that in fact $c_H = c'_H$. A minor adjustment to our proof gives that, for any rational $0 < \alpha < 1$, the quantities wsat(n, H) and wsat $(K_{\alpha n,(1-\alpha)n}, H)$, when $\alpha n \in \mathbb{Z}$, are of the same order of magnitude. Setting $H = K_{t,t}$ answers the above question of [54].

For $q \geq 3$ while we do not have (yet) the same knowledge of limiting constants, a similar method determines asymptotically the weak saturation number of complete *d*-partite *d*-graphs in the clique, generalizing Theorem 4 of [54].

Theorem 8. For every bipartite 2-uniform graph H we have

$$\lim_{n \to \infty} \frac{\operatorname{wsat}(n, H)}{n} = \lim_{n \to \infty} \frac{\operatorname{wsat}(K_{n,n}, H)}{2n}.$$
(3.1)

Furthermore, for any $d \geq 2$ and $1 \leq r_1 \leq \cdots \leq r_d$ we have

wsat
$$(n, K^{d}_{r_1, \dots, r_d}) = \frac{r_1 - 1}{(d - 1)!} n^{d - 1} + O(n^{d - 2}).$$
 (3.2)

The rest of the chapter is organized as follows. In Section 3.1 we give a construction for the upper bound in Theorem 6. In Section 3.2 we provide the necessary background for this chapter. In Section 3.3 we provide the proof for the lower bound in Theorem 6. In Section 3.4 we discuss weak saturation in the clique and prove Theorem 8.

3.1 The upper bound

In this section we prove the upper bound in Theorem 6 by exhibiting a weakly $K^q_{\mathbf{r}}$ -saturated q-graph G. Fix a subset $R \subseteq N$ such that $|R \cap N_i| = r_i$ for every $i \in [d]$ and set

$$\Sigma := \left\{ S \in \binom{N \setminus R}{\leq q} : |S \cap N_i| \leq 1 \text{ for each } i \in [d] \right\}.$$

We define G via its complement in $K_{\mathbf{n}}^q$ as follows. For every $S \in \Sigma$ choose an edge $\lambda(S) \in K_{\mathbf{n}}^q[R \cup S]$ satisfying $S \subseteq \lambda(S)$. Note that the assignment λ is injective, as $\lambda(S) \cap (N \setminus R) = S$. Recall that we associate hypergraphs with their edge sets. Define

$$G := K^q_{\mathbf{n}} \setminus \bigcup_{S \in \Sigma} \lambda(S),$$

so that

$$|E(G)| = \sum_{I \in \binom{[d]}{q}} \prod_{i \in I} n_i - \sum_{I \in \binom{[d]}{\leq q}} \prod_{i \in I} (n_i - r_i).$$

Notice that the choices of $\lambda(S)$ are not unique, but as the next lemma shows, each of them yields a weakly $K_{\mathbf{r}}^q$ -saturated q-graph. Such non-uniqueness is a common occurrence in weak saturation: for instance, every *n*-vertex tree is an extremal example for weak triangle saturation in K_n .

Lemma 9. The q-graph G defined above is weakly $K^q_{\mathbf{r}}$ -saturated. Therefore,

$$w(K_{\mathbf{n}}^{q}, K_{\mathbf{r}}^{q}) \leq |E(G)| = \sum_{I \in \binom{[d]}{q}} \prod_{i \in I} n_{i} - \sum_{I \in \binom{[d]}{\leq q}} \prod_{i \in I} (n_{i} - r_{i}).$$

Proof. For each $0 \le k \le q$ let

$$G_k := G \cup \{T \in K^q_{\mathbf{n}} \colon |T \setminus R| \le k\},\$$

and put $G_{-1} := G$. We claim that adding any new edge $L \in K_{\mathbf{n}}^{q}$ with $|L \setminus R| = k$ to G_{k-1} creates a new copy of $K_{\mathbf{r}}^{q}$ containing L. This gives rise to a $K_{\mathbf{r}}^{q}$ -saturating sequence between G_{k-1} and G_{k} and, by extension, between $G = G_{-1}$ and $G_{q} = K_{\mathbf{n}}^{q}$.

First, notice that G_0 is obtained from G_{-1} by adding the sole missing edge $\lambda(\emptyset)$. Doing so creates a new copy of $K^q_{\mathbf{r}}$, namely $K^q_{\mathbf{n}}[R]$. For an arbitrary k, suppose that L is a missing edge in G_{k-1} such that $S := L \setminus R$ is of size k. Observe that every $T \in K^q[R \cup S]$ is an edge in G_{k-1} unless T = L. Indeed, if $|T \setminus R| < k$ then this holds by definition of G_{k-1} . While otherwise we have $T \setminus R = S$. Hence, by the definition of G, we have $L = \lambda(S)$, so that either T = L or $T \in G \subseteq G_{k-1}$. Therefore, adding L to G_{k-1} creates a new copy of $K^q_{\mathbf{n}}[R \cup S]$ containing L and a fortiori also a new copy of $K^q_{\mathbf{r}}$ containing L, as desired. \Box

3.2 Colorful exterior algebra.

In this section we introduce the linear algebra tools needed for the proof of the lower bound in Theorem 6.

Before we start explaining the algebraic background, we will try to sketch why algebraic tools can be useful in this context. This sketch should be understood loosely—we do not provide any guarantees for the claims in this sketch. In particular, many important technical details are skipped in the sketch. Understanding this sketch is not required in the following text, thus it can be skipped.

Consider first the somewhat trivial case of providing the lower bound on wsat (n, K_3) , the weak saturation number of the complete graph K_3 in K_n . Consider a subgraph G of K_n and a saturating sequence $e_1, \ldots e_k$ of edges in $E(K_n) \setminus E(G)$. Let $G_i := G \cup \{e_1, \ldots, e_i\}$. Because the sequence is saturating, we know that G_i contains a copy of K_3 containing e_i . This means that the dimension of the cycle space of G_i is strictly larger than the dimension of the cycle space of G_{i-1} . Because the final dimension of the cycle space of K_n equals $\binom{n-1}{2}$, we may perform at most $\binom{n-1}{2}$ such steps. In other words $k \leq \binom{n-1}{2}$ and thus $|E(G)| \geq \binom{n}{2} - \binom{n-1}{2}$ as required.

In the language of algebraic topology (which we however do not use in the proofs, no topological background is required), the property that the dimension of the cycle space increases can be phrased so that a new copy of K_3 in each

step belongs to the kernel of the standard boundary operator. For more complicated (hyper)graphs than K_3 it is actually useful to use several independent boundary operators in order to generalize the aforementioned approach. Using such independent operators can be actually efficiently phrased in terms of exterior algebra (without mentioning algebraic topology). They correspond to the *left interior product*, which we discussed in subsection 2.2.1, subject to some suitable independence (genericity) condition.¹

Let $V = \mathbb{R}^N$, in Section 2.2 we have introduced the exterior algebra $\wedge V$. As we are interested in multipartite hypergraphs it is natural to assume in addition that the set N is partitioned as a disjoint union $N = N_1 \sqcup N_2 \sqcup \cdots \sqcup N_d$; consistently with the beginning of this chapter $n_i := |N_i|$. Here each N_i is ordered by a total order $<_i$. We extend these orders to the whole N as follows, for $x \in N_i$ and $y \in N_j$, we say that

$$x < y$$
 if $i < j$ or if $i = j$ and $x <_i y$.

Given the standard basis $(e_v)_{v \in N}$ of V we say that a basis $(f_v)_{v \in N}$ is colorful with respect to this partition if $(f_v)_{v \in N_i}$ generates the same subspace of $V = \mathbb{R}^N$ as $(e_v)_{v \in N_i}$ for every $i \in [d]$; we denote this subspace V_i . Put differently, the transition matrix A from $(e_v)_{v \in N}$ to $(f_v)_{v \in N}$ is a block-diagonal matrix with blocks $N_i \times N_i$ for $i \in [d]$. We also say that $(f_v)_{v \in N}$ is colorful generic (with respect to this partition) if the basis change from $(e_v)_{v \in N_i}$ to $(f_v)_{v \in N_i}$ is generic for every $i \in [d]$. Remember that $(e_v)_{v \in N_i}$ to $(f_v)_{v \in N_i}$ is generic if $\langle f_S, e_T \rangle \neq 0$ for every $S, T \subseteq N_i$ with |S| = |T|. It is possible to choose a basis which is simultaneously colorful generic with respect to a given partition and orthonormal by choosing each change of basis from $(e_v)_{v \in N_i}$ to $(f_v)_{v \in N_i}$ generic and orthonormal.

By $\bigwedge V_i$ we denote the subalgebra of $\bigwedge V$ generated by e_S for $S \subseteq N_i$ and by $\bigwedge^k V_i$ the subspace of $\bigwedge V_i$ with basis $(e_S)_{S \in \binom{N_i}{k}}$; that is, $\bigwedge^k V_i = \bigwedge^k V \cap \bigwedge V_i$.

The following formula expresses a colorful product as a linear combination of colorful elements. Let s_1, \ldots, s_d be integers with $0 \le s_i \le |N_i|$. Suppose that for each $i \in [d]$ we are given

$$h_i = \sum_{S_i \in \binom{N_i}{s_i}} \lambda_{S_i} f_{S_i}$$

for $\lambda_{S_i} \in \mathbb{R}$ (so that $h_i \in \bigwedge^{s_i} V$). Then by bilinearity of \wedge and (2.1) we get

$$h_1 \wedge \dots \wedge h_d = \sum_{\substack{(S_1, \dots, S_d) \in \\ \binom{N_1}{s_1} \times \dots \times \binom{N_d}{s_d}}} \left(\prod_{i \in [d]} \lambda_{S_i}\right) f_{S_1} \wedge \dots \wedge f_{S_d}$$
(3.3)

$$= \sum_{\substack{(S_1,\dots,S_d)\in\\\binom{N_1}{s_1}\times\cdots\times\binom{N_d}{s_d}}} \pm \left(\prod_{i\in[d]}\lambda_{S_i}\right) f_{S_1\cup\cdots\cup S_d}.$$
(3.4)

¹Perhaps the closest relation between the boundary operators and the left interior product can be seen in Lemma 5 interpreting e_R as a simplex with set of vertices R, and $f_{T \sqcup}$ as an operator removing t times the top-dimensional simplices, yielding a linear combination of simplices f_S with r - t vertices. (However, for this relation, it would be even better to express the right hand side using e_S so that all possible e_S would appear.) Adding a colorful aspect (in our case) then makes it easier to work with multipartite (hyper)graphs rather than complete ones.

We claim that the left interior product behaves nicely with respect to a colorful partition. To see this, we first need an auxiliary lemma about signs.

Lemma 10. Let U and T be disjoint subsets of N and for all $i \in [d]$ let $U_i := U \cap N_i$, $T_i := T \cap N_i$, $u_i := |U_i|$ and $t_i := |T_i|$. Then

$$\operatorname{sgn}(U,T) = (-1)^c \operatorname{sgn}(U_1,T_1) \cdots \operatorname{sgn}(U_d,T_d),$$

where c depends only on u_1, \ldots, u_d and t_1, \ldots, t_d .

Proof. The value sgn(U, T) is -1 to the number of transpositions in the permutation π of $U \cup T$ where we first place the elements of U (in our given order on N) and then the elements of T (in the same order). Considering that for i < j, U_i precedes U_j and T_i precedes T_j , the order of the blocks $U_1, \ldots, U_d, T_1, \ldots, T_d$ in π is

$$(U_1,\ldots,U_d,T_1,\ldots,T_d).$$

After c transpositions where c depends only on $u_1, \ldots, u_d, t_1, \ldots, t_d$, we get a permutation π' with the following order of blocks

$$(U_1, T_1, U_2, T_2, \ldots, U_d, T_d).$$

By the above, the sign of π' equals $(-1)^c \operatorname{sgn}(U, T)$. On the other hand, as T_i precedes U_j for i < j in our order on N, the sign of π' is also equal the product $\operatorname{sgn}(U_1, T_1) \cdots \operatorname{sgn}(U_d, T_d)$. Equating these two expressions gives the desired identity.

In the following proposition, the f_i are not necessarily coming from a colorful generic basis. However, we intend to apply it in this setting. With a slight abuse of notation, we use \wedge both for the exterior algebra as well as for the wedge product of multiple elements. (This can be easily distinguished from the context.)

Proposition 11. Suppose that s_1, \ldots, s_d and t_1, \ldots, t_d are nonnegative integers with $t_i \leq s_i \leq n_i$ for every $i \in [d]$. Suppose further that $f_i \in \bigwedge^{t_i} V_i$ and $h_i \in \bigwedge^{s_i} V_i$ for all $i \in [d]$. Then

$$\left(\bigwedge_{i=1}^{d} f_{i}\right) \llcorner \left(\bigwedge_{i=1}^{d} h_{i}\right) = \pm \bigwedge_{i=1}^{d} (f_{i} \llcorner h_{i}).$$

Proof. We will show that

$$\left(\bigwedge_{i=1}^{d} f_{i}\right) \llcorner \left(\bigwedge_{i=1}^{d} h_{i}\right) = (-1)^{c} \bigwedge_{i=1}^{d} (f_{i} \llcorner h_{i})$$
(3.5)

where c comes from Lemma 10; in particular, it depends only on t_1, \ldots, t_d and s_1, \ldots, s_d .

By bilinearity of $\[\] and \land it is sufficient to prove (3.5) in the case when the <math>f_i$ and the h_i are basis elements of $\land^{t_i} V_i$ and $\land^{s_i} V_i$ respectively. So, assume for each $i \in [d]$ that $f_i = e_{T_i}$ and $h_i = e_{S_i}$ where $T_i \in \binom{N_i}{t_i}$ and $S_i \in \binom{N_i}{s_i}$, and let $T := T_1 \cup \cdots \cup T_d$ and $S := S_1 \cup \cdots \cup S_d$. Then $\land^d_{i=1} f_i = e_T$ and $\land^d_{i=1} h_i = e_S$ by the definition of the exterior product \land . If $T_i \not\subseteq S_i$ for some $i \in [d]$, then $T \not\subseteq S$ and both sides of (3.5) vanish by Lemma 4. Therefore, it remains to check the

case that $T_i \subseteq S_i$ for every $i \in [d]$. Here by Lemma 10 (with $U = S \setminus T$) and Lemma 4 we get

$$e_{T \sqcup} e_{S} = \operatorname{sgn}(S \setminus T, T) e_{S \setminus T}$$

= $(-1)^{c} \operatorname{sgn}(S_{1} \setminus T_{1}, T_{1}) \cdots \operatorname{sgn}(S_{d} \setminus T_{d}, T_{d}) e_{S_{1} \setminus T_{1}} \wedge \cdots \wedge e_{S_{d} \setminus T_{d}}$
= $(-1)^{c} (e_{T_{1} \sqcup} e_{S_{1}}) \wedge \cdots \wedge (e_{T_{d} \sqcup} e_{S_{d}}),$

as required.

In this section we prove the lower bound in Theorem 6. Our proof follows a strategy similar to [8] and [48]. Viewing the edges of $K_{\mathbf{n}}^q$ as elements of the exterior algebra of \mathbb{R}^N , we will define a linear mapping closely related to the weak saturation process and lower-bound $w(K_{\mathbf{n}}^q, K_{\mathbf{r}}^q)$ by the rank of the corresponding matrix.

As outlined in Section 2.2, let V be an n-dimensional real vector space with a basis $(e_v)_{v \in N}$, equipped with a standard inner product $\langle \cdot, \cdot \rangle$ with respect to this basis, that is, $(e_v)_{v \in N}$ is orthonormal. Using the exterior product notation of Section 2.2, define

$$\operatorname{span} K_{\mathbf{n}}^q := \operatorname{span} \{ e_T \colon T \in E(K_{\mathbf{n}}^q) \} \subseteq \bigwedge^q V.$$

For an element $m \in \bigwedge^k V$ the support of m is the set

$$\operatorname{supp}(m) = \left\{ S \in \binom{N}{k} : \langle e_S, m \rangle \neq 0 \right\}.$$

The following lemma, which converts the problem at hand into a constructive question in linear algebra, is analogous to Lemma 3 in [8].²

Lemma 12. Let Y be a real vector space and Γ : span $K_{\mathbf{n}}^q \to Y$ a linear map such that for every subset $R \subseteq N$ with $|R \cap N_i| = r_i$ for all $i \in [d]$ there exists an element $m \in \ker \Gamma$ with $\operatorname{supp}(m) = E(K_{\mathbf{n}}^q[R])$. Then

$$w(K_{\mathbf{n}}^q, K_{\mathbf{r}}^q) \ge \operatorname{rank} \Gamma.$$

Proof. Suppose the q-graph G_0 is weakly $K_{\mathbf{r}}^q$ -saturated in $K_{\mathbf{n}}^q$ and $|E(G_0)| = w(K_{\mathbf{n}}^q, K_{\mathbf{r}}^q)$. Denote by $\{L_1, \ldots, L_k\}$ a corresponding saturating sequence and by H_i a new copy of $K_{\mathbf{r}}^q$ that appears in $G_i = G_0 \cup \{L_1, \ldots, L_i\}$ with $L_i \in E(H_i)$. Let $Y_i = \operatorname{span}\{\Gamma(e_T): T \in E(G_i)\}$, and note that $Y_k = \Gamma(\operatorname{span} K_{\mathbf{n}}^q)$. By assumption, for each $i = 1, \ldots, k$ there exist non-zero coefficients $\{c_T: T \in E(H_i)\}$ such that $\sum_{T \in E(H_i)} c_T \Gamma(e_T) = 0$. Therefore,

$$\Gamma(e_{L_i}) = -\frac{1}{c_{L_i}} \sum_{T \in E(H_i) \setminus L_i} c_T \Gamma(e_T) \in Y_{i-1}.$$

г	_	_	ъ
L			т
L			т
-	-	-	-

²Put equivalently in the language of [8], we map each edge of K_n^q to vector in a certain vector space \tilde{W} , so that for each copy of $K_{\mathbf{r}}^q$ in K_n^q the underlying vectors are linearly dependent with all coefficients involved being non-zero. This implies $w(K_n^q, K_{\mathbf{r}}^q) \geq \dim \tilde{W}$.

We conclude that $Y_i = Y_{i-1}$. By repeating this procedure we obtain

$$w(K_{\mathbf{n}}^q, K_{\mathbf{r}}^q) = |E(G_0)| \ge \dim Y_0 = \dim Y_k = \operatorname{rank} \Gamma.$$

Our goal now is to define a linear map Γ as in Lemma 12. For this purpose let us fix an orthonormal colorful generic basis $(f_v)_{v \in N}$ of V with respect to the partition of N, as described in Section 3.2. Next, for each $i \in [d]$ choose a set $J_i \subseteq N_i$ with $|J_i| = r_i - 1$ and a vertex $w_i \in N_i \setminus J_i$. Put $J := \bigcup_{i \in [d]} J_i$ and $W := \{w_i : i \in [d]\}$. Finally, set s := d - q and

$$g := \sum_{T \in \binom{W}{s}} f_T. \tag{3.6}$$

We can now state the following auxiliary lemma.

Lemma 13. Let z be an integer with $d \ge z \ge s$ and let $Z \in \binom{N}{z}$. Then

(*i*)
$$g \llcorner f_Z = 0$$
 if $|Z \cap W| < s$.

(*ii*) If
$$z = s$$
, then $\langle g, f_Z \rangle = \begin{cases} \pm 1 & \text{if } Z \subseteq W, \\ 0 & \text{if } Z \not\subseteq W. \end{cases}$

Proof. By (3.6), bilinearity of \lfloor , and Lemma 4 we get

$$g\llcorner f_Z = \sum_{W' \in \binom{W}{s}} f_{W'} \llcorner f_Z = \sum_{W' \in \binom{W \cap Z}{s}} \pm f_{Z \setminus W'}.$$
(3.7)

The last expression is 0 if $|Z \cap W| < s$; this shows (i).

Now, assume that z = s. Then

$$\langle g, f_Z \rangle = \langle f_{\emptyset} \wedge g, f_Z \rangle = \langle f_{\emptyset}, g \llcorner f_Z \rangle \stackrel{(3.7)}{=} \sum_{W' \in \binom{W \cap Z}{s}} \pm \langle f_{\emptyset}, f_{Z \backslash W'} \rangle.$$
(3.8)

If $Z \not\subseteq W$, then $|Z \cap W| < z = s$, so $g \downarrow f_Z = 0$ from (i), and thus (3.8) evaluates to 0. On the other hand, if $Z \subseteq W$, then $\binom{W \cap Z}{s} = \{Z\}$. It follows that

$$\langle g, f_Z \rangle \stackrel{(3.8)}{=} \pm \langle f_{\emptyset}, f_{\emptyset} \rangle = \pm 1,$$

yielding (ii).

We define the subspace

$$U := \operatorname{span}\{g \llcorner f_T \colon T \in E(K^d_{\mathbf{n}}[N \setminus J]), |T \cap W| \ge s\},\tag{3.9}$$

and observe first that $U \subseteq \operatorname{span} K_{\mathbf{n}}^q$. Indeed, for each T in (3.9) and $W' \in \binom{W}{s}$, we have by Lemma 4 that $f_{W'} \sqcup f_T = 0$ if $W' \not\subseteq T$ and $f_{W'} \sqcup f_T = \pm f_{T \setminus W'}$ if $W' \subseteq T$. In the latter case note that $T \setminus W' \in E(K_{\mathbf{n}}^q)$, and the claim follows by bilinearity of \sqcup .

Let Y be the orthogonal complement of U in span $K_{\mathbf{n}}^q$ and let Γ : span $K_{\mathbf{n}}^q \rightarrow$ span $K_{\mathbf{n}}^q$ be the orthogonal projection on Y. Our main technical lemma states that Γ satisfies the assumptions of Lemma 12.

Lemma 14. Suppose that $R \subseteq N$ satisfies $|R \cap N_i| = r_i$ for every $i \in [d]$. Then, there exists $m \in \ker \Gamma$ such that $\operatorname{supp}(m) = E(K_{\mathbf{n}}^q[R])$.

Deferring the proof of Lemma 14, let us first compute rank Γ and conclude the proof of Theorem 6 assuming Lemma 14.

Notice that the sets $T \in K^d_{\mathbf{n}}[N \setminus J]$ with $|T \cap W| \geq s$ are in bijective correspondence with the sets $T \setminus W \in K^p_{\mathbf{n}}[N \setminus (J \cup W)]$ with $p \leq q$. Using this bijection,

$$\dim U \stackrel{(3.9)}{\leq} |\{T \in K^{d}_{\mathbf{n}}[N \setminus J] \colon |T \cap W| \geq s\}| = \sum_{\substack{I \subseteq [d] \\ |I| \leq q}} \prod_{i \in I} (n_{i} - r_{i}).$$

Consequently,

$$\operatorname{rank} \Gamma = \operatorname{dim}(\operatorname{span} K_{\mathbf{n}}^{q}) - \operatorname{dim} U \ge \sum_{I \in \binom{[d]}{q}} \prod_{i \in I} n_{i} - \sum_{\substack{I \subseteq [d] \\ |I| \le q}} \prod_{i \in I} (n_{i} - r_{i}).$$
(3.10)

Proof of Theorem 6. On the one hand, by Lemma 14 the map Γ satisfies the assumptions of Lemma 12. Therefore,

$$w(K_{\mathbf{n}}^{q}, K_{\mathbf{r}}^{q}) \ge \operatorname{rank} \Gamma \overset{(3.10)}{\ge} \sum_{I \in \binom{[d]}{q}} \prod_{i \in I} n_{i} - \sum_{\substack{I \subseteq [d] \\ |I| \le q}} \prod_{i \in I} (n_{i} - r_{i}).$$

On the other hand, Lemma 9 gives the same upper bound.

Proof of Lemma 14. We claim that

$$m = (g \wedge f_J) \llcorner e_R$$

is the desired element.³ Let $R_i := R \cap N_i$ for each $i \in [d]$.

First, we verify that $m \in \ker \Gamma = U$. By Proposition 11 we have

$$f_{J} \llcorner e_R = \pm (f_{J_1} \llcorner e_{R_1}) \land \dots \land (f_{J_d} \llcorner e_{R_d})$$

By Lemma 5 we can write each of these terms as

$$f_{J_i \sqcup} e_{R_i} = \sum_{v \in N_i \setminus J_i} \lambda_v f_v \quad \text{with all} \quad \lambda_v \neq 0.$$
(3.11)

Combining this with (3.3) gives

$$f_{J \sqcup} e_R = \sum_{Z \in E(K_{\mathbf{n}}^d[N \setminus J])} \pm (\prod_{v \in Z} \lambda_v) f_Z.$$
(3.12)

³Let us briefly sketch the topological idea hidden behind this choice: As it can be easily deduced from the computations below, m can be also expressed as $\pm g_{\perp}((f_{J_1 \perp} e_{R_1}) \wedge \cdots \wedge (f_{J_d \perp} e_{R_d}))$. In the terminology of simplicial complexes interpreting loosely (i) e_{R_i} as a full simplex on the vertex set R_i , (ii) \wedge as a join of simplicial complexes and (iii) \perp as an operator taking the skeleton of appropriate dimension, we gradually get the following: $f_{J_i \perp} e_{R_i}$ corresponds to the 0-skeleton of the simplex on R_i , that is, the vertices of R_i . Then $(f_{J_1 \perp} e_{R_1}) \wedge \cdots \wedge (f_{J_d \perp} e_{R_d})$ corresponds to the join of the sets R_i , that is, the complete *d*-partite complex on R_1, \ldots, R_d . Finally, applying g_{\perp} to this element takes the skeleton again reducing the dimension so that the corresponding hypergraph is the required $K_n^{\mathbf{q}}[R]$.

Therefore, we get

$$m = (g \wedge f_J) \llcorner e_R \stackrel{(2.5)}{=} g \llcorner (f_J \llcorner e_R) \stackrel{(3.12)}{=} \sum_{\substack{Z \in E(K_{\mathbf{n}}^d[N \setminus J]) \\ |Z \cap W| \ge s}} (\prod_{v \in Z} \lambda_v) g \llcorner f_Z,$$

where the last equality follows by Lemma 13(i) with z = d. Thus $m \in U$ as wanted.

Next, we show that $\operatorname{supp}(m) = E(K_{\mathbf{n}}^{q}[R])$. As we just have shown, $m \in U \subseteq \operatorname{span} K_{\mathbf{n}}^{q}$, i.e. $\operatorname{supp}(m) \subseteq E(K_{\mathbf{n}}^{q})$. Now, for $T \in E(K_{\mathbf{n}}^{q})$ we have

$$\langle e_T, m \rangle \stackrel{(2.4)}{=} \langle e_T \wedge (g \wedge f_J), e_R \rangle = \pm \langle (g \wedge f_J) \wedge e_T, e_R \rangle \stackrel{(2.4)}{=} \pm \langle g \wedge f_J, e_T \llcorner e_R \rangle.$$
(3.13)

If $T \notin E(K_{\mathbf{n}}^{q}[R])$, then $T \nsubseteq R$ and by Lemma 4 we have $e_{T \sqcup} e_{R} = 0$, and consequently $\langle e_{T}, m \rangle = 0$. Hence, $T \notin \operatorname{supp}(m)$.

Now assume that $T \in E(K_{\mathbf{n}}^{q}[R])$, i.e., $T \subseteq R$. By (3.13) and Lemma 4 we have

$$\langle e_T, m \rangle = \pm \langle g \wedge f_J, e_{R \setminus T} \rangle \stackrel{(2.4)}{=} \pm \langle g, f_{J \sqcup} e_{R \setminus T} \rangle.$$
 (3.14)

Let $P := \{i \in [d] : T \cap N_i \neq \emptyset\}$ and $P' := [d] \setminus P$. Using this notation we can write

$$e_{R\setminus T} = \pm \left(\bigwedge_{i\in P} e_{R_i\setminus \tau_i}\right) \wedge \left(\bigwedge_{i\in P'} e_{R_i}\right),$$

where for each $i \in P$ the set $\tau_i = T \cap N_i$ contains a single vertex. Applying Proposition 11, we deduce

$$f_{J} \llcorner e_{R \setminus T} = \pm \left(\bigwedge_{i \in P} f_{J_i} \llcorner e_{R_i \setminus \tau_i} \right) \land \left(\bigwedge_{i \in P'} f_{J_i} \llcorner e_{R_i} \right).$$
(3.15)

Since $|J_i| = r_i - 1 = |R_i \setminus \tau_i|$, by Lemma 3 for every $i \in P$ we have $f_{J_i \sqcup} e_{R_i \setminus \tau_i} \in \bigwedge^0 V$. Thus

$$f_{J_i \sqcup} e_{R_i \setminus \tau_i} = \langle e_{\emptyset}, f_{J_i \sqcup} e_{R_i \setminus \tau_i} \rangle e_{\emptyset} = \langle e_{\emptyset} \wedge f_{J_i}, e_{R_i \setminus \tau_i} \rangle e_{\emptyset} = \langle f_{J_i}, e_{R_i \setminus \tau_i} \rangle e_{\emptyset},$$

and notice that $\langle f_{J_i}, e_{R_i \setminus \tau_i} \rangle \neq 0$ because $(f_v)_{v \in N_i}$ is generic with respect to $(e_v)_{v \in N_i}$. Plugging it into (3.15) yields

$$f_{J} \llcorner e_{R \setminus T} = \pm \left(\bigwedge_{i \in P} \langle f_{J_i}, e_{R_i \setminus \tau_i} \rangle e_{\emptyset} \right) \land \left(\bigwedge_{i \in P'} f_{J_i} \llcorner e_{R_i} \right)$$
(3.16)

$$= \pm \left(\prod_{i \in P} \langle f_{J_i}, e_{R_i \setminus \tau_i} \rangle\right) \bigwedge_{i \in P'} f_{J_i \sqcup} e_{R_i}.$$
(3.17)

Turning to P', denote $N' := \bigcup_{i \in P'} N_i \setminus J_i$. We have

$$\bigwedge_{i \in P'} f_{J_i \sqcup} e_{R_i} \stackrel{(3.11)}{=} \bigwedge_{i \in P'} \left(\sum_{v \in N_i \setminus J_i} \lambda_v f_v \right) \stackrel{(3.3)}{=} \sum_{Z \in E(K_{\mathbf{n}}^s[N'])} \pm \left(\prod_{v \in Z} \lambda_v \right) f_Z.$$
(3.18)

Therefore,

<

$$g, \bigwedge_{i \in P'} f_{J_i \sqcup} e_{R_i} \rangle = \sum_{Z \in E(K_{\mathbf{n}}^s[N'])} \pm (\prod_{v \in Z} \lambda_v) \langle g, f_Z \rangle = \pm \prod_{v \in W \cap N'} \lambda_v, \tag{3.19}$$

where the second equality is due to Lemma 13(ii), using that there is exactly one $Z \in E(K_{\mathbf{n}}^{s}[N'])$ with $Z \subseteq W$, namely $Z = W \cap N'$. Putting it all together,

$$\langle e_T, m \rangle \stackrel{(3.14)}{=} \pm \langle g, f_{J \sqcup} e_{R \setminus T} \rangle \stackrel{(3.16)}{=} \pm (\prod_{i \in P} \langle f_{J_i}, e_{R_i \setminus \tau_i} \rangle) \langle g, \bigwedge_{i \in P'} f_{J_i \sqcup} e_{R_i} \rangle$$

$$\stackrel{(3.19)}{=} \pm (\prod_{i \in P} \langle f_{J_i}, e_{R_i \setminus \tau_i} \rangle) \prod_{v \in W \cap N'} \lambda_v \neq 0,$$

and consequently $T \in \operatorname{supp}(m)$.

3.4 Weak saturation in the clique

In this section we prove Theorem 8. Let H be a q-graph where $q \ge 2$ without isolated vertices. We recall the notion of a *link hypergraph* of a vertex $v \in V(H)$: it is the (q-1)-graph (possibly with isolated vertices) defined via

$$L_H(v) := \{e \setminus \{v\} \colon e \in E(H), v \in e\}.$$

The *co-degree* of a set W of q - 1 vertices in H is

$$d_H(W) := |\{e \in E(H) : W \subset e\}|.$$

Define the minimum positive co-degree of H, in notation $\delta^*(H)$, as

$$\delta^*(H) := \min\left\{d_H(W) \colon W \in \binom{V(H)}{q-1}, d_H(W) > 0\right\}.$$

Notice that $\delta^*(H) \leq \delta^*(L_H(v))$ for all $v \in V(H)$, and equality holds for some v. Lemma 15. wsat $(n, H) \leq (\delta^*(H) - 1) \binom{n}{q-1} + O_H(n^{q-2})$.

Proof. We apply induction on q. For q = 2 this is a well-known fact ([31], Theorem 4). Suppose now that $q \ge 3$ and the statement holds for all smaller values. Let H be a q-graph and let $W = \{v_1, \ldots, v_{q-1}\}$ be a set satisfying $d_H(W) = \delta^*(H)$. Let $H_1 = L_H(v_1)$ be the link hypergraph of v_1 , and observe that $\delta^*(H_1) = \delta^*(H)$. A weakly H-saturated q-graph on [n] is obtained as follows. Take a minimum weakly H_1 -saturated (q-1)-graph on [n-1] and insert n into each edge; take a union of the resulting q-graph with a minimum weakly Hsaturated q-graph on [n-1]. We therefore obtain

$$\operatorname{wsat}(n, H) \leq \operatorname{wsat}(n-1, H) + \operatorname{wsat}(n-1, H_1).$$

Iterating and applying the induction hypothesis,

$$wsat(n, H) \le wsat(|V(H)|, H) + \sum_{m=|V(H)|}^{n-1} wsat(m, H_1)$$
$$\le (\delta^*(H_1) - 1) \sum_{m=q-2}^{n-1} {m \choose q-2} + O_H(n^{q-2})$$
$$= (\delta^*(H) - 1) {n \choose q-1} + O_H(n^{q-2}).$$

The tensor product of two q-graphs G and J, $G \times J$ is defined having the vertex set $V(G) \times V(J)$ and the edge set

$$E(G \times J) = \{\{(v_1, w_1), \dots, (v_q, w_q)\} : \{v_1, \dots, v_q\} \in E(G), \{w_1, \dots, w_q\} \in E(J)\}.$$

(Note that every pair of edges in the original graphs produces q! edges in the product.)

Lemma 16. Let $H = K^d_{r_1,...,r_d}$, and let F^d_n be the copy of $K^d(n;d)$ between the vertex sets $[n] \times \{1\}, \ldots, [n] \times \{d\}$. Then there exists a d-graph $E^d(n,H) \subseteq F^d_n \setminus (K^d_{[n]} \times K^d_{[d]})$ of size $O_H(n^{d-2})$ such that

$$G(n,H) := (K^d_{[n]} \times K^d_{[d]}) \sqcup E^d(n,H)$$

is weakly H-saturated in F_n^d .

Proof. It suffices to prove the above statement when $r_1 = \cdots = r_d =: r$, i.e. when $H = K^d(r; d)$, as every edge creating a new copy of $K^d(\max\{r_1, \ldots, r_d\}; d)$ creates in particular a new copy of $K^d_{r_1, \ldots, r_d}$.

We apply induction on d and n. For d = 2 and any $n \ge |V(H)|$ the graph $K_{[n]} \times K_{[2]}$ misses only a matching from F_n^2 , making it already H-saturated in F_n^2 , as can be easily checked. Moreover, for every fixed H we can assume the statement to hold for all n less than some large C(H).

For the induction step, fix (n, d) and suppose that the statement holds for all (n', d') with d' < d and all (n'', d) with n'' < n. It suffices to show that $O_H(n^{d-3})$ edges can be added to G(n-1, H) to satisfy the assertion; these edges will be as follows.

For each $i \in [d]$ let the (d-1)-graph E'_i be an isomorphic copy of $E^{d-1}(n-1, K^{d-1}(r; d-1))$ between the sets $[n-1] \times \{j\}$ for $j \in [d] \setminus \{i\}$, such that $(K^{d-1}_{[n-1]} \times K^{d-1}_{[d] \setminus \{i\}}) \sqcup E'_i$ is weakly $K^{d-1}(r; d-1)$ -saturated in the complete (d-1)-partite (d-1)-graph between the sets $[n-1] \times \{j\}$ for $j \in [d] \setminus \{i\}$. Let

$$E_i := \{ e \sqcup \{ (n, i) \} : e \in E'_i \}.$$

By the induction hypothesis $|E_i| = |E'_i| = O_H(n^{d-3})$.

Similarly, for each $\{i_1, i_2\} \in {\binom{[d]}{2}}$ apply Corollary 7 to obtain a (d-2)-graph E'_{i_1,i_2} of size $O_H(n^{d-3})$ which is weakly $K^{d-2}(r; d-2)$ -saturated in the copy of $K^{d-2}(n-1; d-2)$ between the sets $[n-1] \times \{j\}$ for $j \in [d] \setminus \{i_1, i_2\}$ (for d=3 take any r-1 vertices in $[n-1] \times [d] \setminus \{i_1, i_2\}$). As above, insert (n, i_1) and (n, i_2) into each edge of E'_{i_1,i_2} ; let the resulting edge set be called E_{i_1,i_2} .

Finally, take all edges of F_n^d containing at least three vertices with n as their first coordinate, and let E_0 be this edge set; clearly $|E_0| = O_H(n^{d-3})$ as well. Put

$$G(n,H) := G(n-1,H) \cup \bigcup_{i \in [d]} E_i \cup \bigcup_{\{i_1,i_2\} \in \binom{[d]}{2}} E_{i_1,i_2} \cup E_0,$$

and

$$E^d(n,H) := G(n,H) \setminus (K^d_{[n]} \times K^d_{[d]}).$$

By the induction hypothesis and the bounds on the $|E_i|$, the $|E_{i_1,i_2}|$ and $|E_0|$, we have $|E^d(n, H)| = O_H(n^{d-2})$. To see that G(n, H) is weakly H-saturated, first note that by induction hypothesis G(n-1, H) is weakly *H*-saturated in F_{n-1}^d , hence the *d*-graph $G(n-1, H) \cup (K_{[n]}^d \times K_{[d]}^d) \subseteq G(n, H)$ is weakly *H*-saturated in $J_0 := F_{n-1}^d \cup (K_{[n]}^d \times K_{[d]}^d)$. Furthermore, let

$$K_1 := \{ e \in F_n^d \colon |e \cap (\{n\} \times [d])| = 1 \},\$$

and

$$K_2 := \{ e \in F_n^d \colon |e \cap (\{n\} \times [d])| = 2 \}.$$

Let $J_1 := J_0 \cup K_1$ and $J_2 := J_1 \cup K_2$. By construction, $J_0 \cup \bigcup_{i \in [d]} E_i$ is weakly *H*-saturated in $J_1, J_1 \cup \bigcup_{\{i_1, i_2\} \in \binom{[d]}{2}} E_{i_1, i_2}$ is weakly *H*-saturated in J_2 and $J_2 \cup E_0 = F_n^d$. Thus, G(n, H) is weakly *H*-saturated in F_n^d as desired. This proves the induction step, and the statement of the lemma follows. \Box

Proof of Theorem 8. For the first statement, suppose that $G \subseteq K_{n,n}$ is weakly *H*-saturated in $K_{n,n}$. Placing two |V(H)|-cliques on the parts of *G* is easily seen to produce a weakly *H*-saturated graph in K_{2n} . Therefore,

$$wsat(2n, H) \le wsat(K_{n,n}, H) + |V(H)|^2.$$
 (3.20)

Conversely, suppose that $G = G_0$ is weakly *H*-saturated in $K_{[n]}$ via a saturating sequence $e_1 = \{i_1, j_1\}, \ldots, e_k = \{i_k, j_k\}$. For $1 \le \ell \le k$ let $G_\ell = G_0 \cup \{e_1, \ldots, e_\ell\}$, and let H_ℓ be a copy of *H* in G_ℓ containing e_ℓ .

Let $G^{bip} = G \times K_{[2]}$, i.e., $V(G^{bip}) = [n] \times \{1, 2\}$ and

$$E(G^{bip}) = \{\{(i,1), (j,2)\} : \{i,j\} \in E(G)\}.$$

We claim that G^{bip} is weakly *H*-saturated in $K_{[n]}^{bip} = K_{[n]} \times K_{[2]}$ via the *H*-saturating sequence $f_1, f'_1, \ldots, f_k, f'_k$, where, for each $\ell \in [k], f_\ell = \{(i_\ell, 1), (j_\ell, 2)\}$ and $f'_\ell = \{(i_\ell, 2), (j_\ell, 1)\}$, and that $G_{\ell-1}^{bip} \cup \{f_\ell, f'_\ell\} = G_\ell^{bip}$ for all $\ell \in [k]$ (where G_ℓ^{bip} is defined analogously, i.e., $G_\ell^{bip} = G_\ell \times K_{[2]}$). Indeed, let (A, B) be a bipartition of $V(H_\ell)$ with $i_\ell \in A$ and $j_\ell \in B$, and consider the analogous graph H_ℓ^b between $A \times \{1\}$ and $B \times \{2\}$, i.e., for every $(i, j) \in A \times B$ we have $\{(i, 1), (j, 2)\} \in E(H_\ell^b)$ if and only if $\{i, j\} \in E(H_\ell)$. Note that $f_\ell \in E(H_\ell^b)$ is the only edge of H_ℓ^b not already present in $G_{\ell-1}^{bip}$, therefore we can add it to the latter creating a new copy of H, namely H_ℓ^b . Symmetrically, taking a graph $H_\ell^{\prime b}$ between $A \times \{2\}$ and $B \times \{1\}$ allows to add f'_ℓ . Since $G_\ell = G_{\ell-1} \cup e_\ell$, we have $G_{\ell-1}^{bip} \cup \{f_\ell, f'_\ell\} = G_\ell^{bip}$. Finally, note that $G^{bip} \cup \{f_1, \ldots, f'_k\} = G_k^{bip} = K_{[n]}^{bip}$.

Note that $K_{[n]}^{bip}$ is isomorphic to $K_{n,n}$ minus a perfect matching, and it is a straightforward check that this graph is *H*-saturated in $K_{n,n}$ (we can assume that $|V(H)| \leq n$). We have thus shown

$$wsat(K_{n,n}, H) \le 2 wsat(n, H).$$
(3.21)

Combining (3.20) and (3.21) gives

$$\frac{\operatorname{wsat}(2n,H)}{2n} - o(1) \le \frac{\operatorname{wsat}(K_{n,n},H)}{2n} \le \frac{\operatorname{wsat}(n,H)}{n},$$

and taking the limit, (3.1) follows readily.

For the second statement, denote $H = K^d_{r_1,\dots,r_d}$ where $1 \leq r_1 \leq \cdots \leq r_d$. Observe that the upper bound in (3.2) holds by Lemma 15, as $\delta^*(H) = r_1$. To prove the lower bound, suppose G is weakly H-saturated in $K^d_{[n]}$, and that $|E(G)| = \operatorname{wsat}(n, H)$. Let $G^{mult} = G \times K^d_{[d]}$, that is, $V(G^{mult}) = [n] \times [d]$ and

$$E(G^{mult}) = \{\{(i_1, 1), \dots, (i_d, d)\} : \{i_1, \dots, i_d\} \in E(G)\}.$$

Essentially the same argument as for G^{bip} before shows that G^{mult} is weakly H-saturated in $K^d_{[n]} \times K^d_{[d]}$. By Lemma 16 adding further $O_H(n^{d-2})$ edges creates a weakly H-saturated d-graph in $K^d(n; d)$. Hence,

$$\operatorname{wsat}(K^d(n;d),H) \le |E(G^{mult})| + O(n^{d-2}) = d! \operatorname{wsat}(n,H) + O(n^{d-2}).$$
 (3.22)

On the other hand, Moshkovitz and Shapira [64] proved that wsat $(K^d(n; d), H) = d(r_1 - 1)n^{d-1} + O(n^{d-2})$. Combining this with (3.22) yields the lower bound in (3.2).

4. Colorful Fractional Helly Theorem

The target of this chapter is to provide optimal bounds for the colorful fractional Helly theorem first stated by Bárány, Fodor, Montejano, Oliveros, and Pór [11], and then improved by Kim [53]. In order to explain the colorful fractional Helly theorem, let us briefly survey the preceding results.

The starting point, as usual in this context, is the Helly theorem:

Theorem 17 (Helly's theorem [38]). Let \mathcal{F} be a finite family of at least d + 1 convex sets in \mathbb{R}^d . Assume that every subfamily of \mathcal{F} with exactly d + 1 members has a nonempty intersection. Then all sets in \mathcal{F} have a nonempty intersection.

Helly's theorem admits numerous extensions and two of them, important in our context, are the fractional Helly theorem and the colorful Helly theorem. The fractional Helly theorem of Katchalski and Liu covers the case when only some fraction of the d + 1 tuples in \mathcal{F} has a nonempty intersection.

Theorem 18 (The fractional Helly theorem [52]). For every $\alpha \in (0, 1]$ and every non-negative integer d, there is $\beta = \beta(\alpha, d) \in (0, 1]$ with the following property. Let \mathcal{F} be a finite family of $n \geq d+1$ convex sets in \mathbb{R}^d such that at least $\alpha \binom{n}{d+1}$ of the subfamilies of \mathcal{F} with exactly d+1 members have a nonempty intersection. Then there is a subfamily of \mathcal{F} with at least βn members with a nonempty intersection.

An interesting aspect of the fractional Helly theorem is not only to show the existence of $\beta(\alpha, d)$ but also to provide the largest value of $\beta(\alpha, d)$ with which the theorem is valid. This has been resolved independently by Eckhoff [26] and by Kalai [46] showing that the fractional Helly theorem holds with $\beta(\alpha, d) = 1 - (1 - \alpha)^{1/(d+1)}$; yet another simplified proof of this fact has been subsequently given by Alon and Kalai [3]. It is well known that this bound is sharp by considering a family \mathcal{F} consisting of $\approx (1 - (1 - \alpha)^{1/(d+1)})n$ copies of \mathbb{R}^d and $\approx (1 - \alpha)^{1/(d+1)}n$ hyperplanes in general position; see, e.g., the introduction of [46].

The colorful Helly theorem of Lovász covers the case where the sets are colored by d + 1 colors and only the 'colorful' (d + 1)-tuples of sets in \mathcal{F} are considered. Given families $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$ of sets in \mathbb{R}^d a family of sets $\{F_1, \ldots, F_{d+1}\}$ is a colorful (d + 1)-tuple if $F_i \in \mathcal{F}_i$ for $i \in [d + 1]$, where $[n] := \{1, \ldots, n\}$ for a nonnegative integer $n \geq 1$. (The reader may think of \mathcal{F} from preceding theorems decomposed into color classes $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$.)

Theorem 19 (The colorful Helly theorem [59, 10]). Let $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$ be finite nonempty families of convex sets in \mathbb{R}^d . Let us assume that every colorful (d+1)tuple has a nonempty intersection. Then one of the families $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$ has a nonempty intersection.

Both the colorful Helly theorem and the fractional Helly theorem with optimal bounds imply the Helly theorem. The colorful one by setting $\mathcal{F}_1 = \cdots = \mathcal{F}_{d+1} = \mathcal{F}$ and the fractional one by setting $\alpha = 1$ giving $\beta(1, d) = 1$.

The preceding two theorems can be merged into the following colorful fractional Helly theorem: **Theorem 20** (The colorful fractional Helly theorem [11]). For every $\alpha \in (0, 1]$ and every non-negative integer d, there is $\beta_{col} = \beta_{col}(\alpha, d) \in (0, 1]$ with the following property. Let $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$ be finite nonempty families of convex sets in \mathbb{R}^d of sizes n_1, \ldots, n_{d+1} respectively. If at least $\alpha n_1 \cdots n_{d+1}$ of the colorful (d+1)-tuples have a nonempty intersection, then there is $i \in [d+1]$ such that \mathcal{F}_i contains a subfamily of size at least $\beta_{col}n_i$ with a nonempty intersection.

Bárány et al. proved the colorful fractional Helly theorem with the value $\beta_{col}(\alpha, d) = \frac{\alpha}{d+1}$ and they used it as a lemma [11, Lemma 3] in a proof of a colorful variant of a (p, q)-theorem. Despite this, the optimal bound for β_{col} seems to be of independent interest. In particular, the bound on β_{col} has been subsequently improved by Kim [53] who showed that the colorful fractional Helly theorem is true with $\beta_{col}(\alpha, d) = \max\{\frac{\alpha}{d+1}, 1 - (d+1)(1-\alpha)^{1/(d+1)}\}$. On the other hand, the value of $\beta_{col}(\alpha, d)$ cannot go beyond $1 - (1-\alpha)^{1/(d+1)}$ because essentially the same example as for the standard fractional Helly theorem applies in this setting as well—it is sufficient to set $n_1 = n_2 = \cdots = n_{d+1}$ and take $\approx (1 - (1-\alpha)^{1/(d+1)})n_i$ copies of \mathbb{R}^d and $\approx (1-\alpha)^{1/(d+1)}n_i$ hyperplanes in general position in each color class.¹ (Kim [53] provides a slightly different upper bound example showing the same bound.)

Coming back to the lower bound on $\beta_{col}(\alpha, d)$, Kim explicitly conjectured that $1 - (1 - \alpha)^{1/(d+1)}$ is also a lower bound, thereby an optimal bound for the colorful fractional Helly theorem. He also provides a more refined conjecture, that we discuss slightly later on (see Conjecture 24), which implies this lower bound. We prove the refined conjecture, and therefore the optimal bounds for the colorful fractional Helly theorem.

Theorem 21 (The optimal colorful fractional Helly theorem). Let $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$ be finite nonempty families of convex sets in \mathbb{R}^d of sizes n_1, \ldots, n_{d+1} respectively. If at least $\alpha n_1 \cdots n_{d+1}$ of the colorful (d+1)-tuples have a nonempty intersection, for $\alpha \in (0,1]$, then there is $i \in [d+1]$ such that \mathcal{F}_i contains a subfamily of size at least $(1 - (1 - \alpha)^{1/(d+1)})n_i$ with a nonempty intersection.

In the proof we follow the exterior algebra approach which has been used by Kalai [46] in order to provide optimal bounds for the standard fractional Helly theorem. We have to upgrade Kalai's proof to the colorful setting. This requires guessing the right generalization of several steps in Kalai's proof (in particular guessing the statement of Theorem 26 below). However, we honestly admit that after making these 'guesses' we follow Kalai's proof quite straightforwardly.

Let us also compare one aspect of our proof with the previous proof of the weaker bound by Kim [53]: Kim's proof uses the colorful Helly theorem as a blackbox while our proof includes the proof of the colorful Helly theorem.

Last but not least, the exterior algebra approach actually allows to generalize Theorem 21 in several different directions. The extension to so called d-collapsible complexes is essentially mandatory for the well working proof while the other generalizations that we will present just follow from the method. We will discuss this in detail in forthcoming sections.

¹At the end of Section 4.3 we discuss this example in full detail in more general context. However, in this special case, it is perhaps much easier to check directly that β_{col} cannot be improved due to this example.

4.1 *d*-representable and *d*-collapsible complexes

The nerve and *d*-representable complexes. The important information in Theorems 17, 18, 19, 20, and 21 is which subfamilies have a nonempty intersection. This information can be efficiently stored in a simplicial complex called the nerve.

Given a family of sets \mathcal{F} , the *nerve* of \mathcal{F} , is the simplicial complex whose vertex set is \mathcal{F} and whose faces are subfamilies with a nonempty intersection. A simplicial complex is *d*-representable if it is the nerve of a finite family of convex sets in \mathbb{R}^d .

As a preparation for the *d*-collapsible setting, we now restate Theorem 21 in terms of *d*-representable complexes. For this we need two more notions. Given a simplicial complex K and a subset U of the vertex set N, the *induced subcomplex* K[U] is defined as $K[U] := \{A \in K : A \subseteq U\}$. Now, let us assume that the vertex set N is split into d + 1 pairwise disjoint subsets $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ (we can think of this partition as coloring each vertex of N with one of the d + 1 possible colors). Then a *colorful d-face* is a *d*-face A, such that $|A \cap N_i| = 1$ for every $i \in [d+1]$.

Theorem 22 (Theorem 21 reformulated). Let K be a d-representable simplicial complex with the set of vertices $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ divided into d+1 disjoint subsets. Let $n_i := |N_i|$ for $i \in [d+1]$ and assume that K contains at least $\alpha n_1 \cdots n_{d+1}$ colorful d-faces for some $\alpha \in (0,1]$. Then there is $i \in [d+1]$ such that dim $K[N_i] \ge (1 - (1 - \alpha)^{1/(d+1)})n_i - 1$.

Theorem 22 is indeed just a reformulation of Theorem 21: Considering \mathcal{F} as disjoint union² $\mathcal{F} = \mathcal{F}_1 \sqcup \cdots \sqcup \mathcal{F}_{d+1}$, then K corresponds to the nerve of \mathcal{F} , colorful d-faces correspond to colorful (d + 1)-tuples with nonempty intersection and the dimension of $K[N_i]$ corresponds to the size of largest subfamily of \mathcal{F}_i with nonempty intersection minus 1. (The shift by minus 1 between size of a face and dimension of a face is a bit unpleasant; however, we want to follow the standard terminology.)

d-collapsible complexes. In [87] Wegner introduced an important class of simplicial complexes, called d-collapsible complexes. They include all d-representable complexes, which is the main result of [87], while they admit quite simple combinatorial description which is useful for induction.

Given a simplicial complex K, we say that a simplicial complex K' arises from K by an *elementary d-collapse*, if there are faces $L, M \in K$ with the following properties: (i) dim $L \leq d - 1$; (ii) M is the unique inclusion-wise maximal face which contains L; and (iii) $K' = K \setminus \{A \in K : L \subseteq A\}$. A simplicial complex K is *d-collapsible* if there is a sequence of simplicial complexes K_0, \ldots, K_ℓ such that $K = K_0$; K_i arises from K_{i-1} by an elementary *d*-collapse for $i \in [\ell]$; and K_ℓ is the empty complex.

We will prove the following generalization of Theorem 22 (equivalently of Theorem 21).

²If there are any repetitions of sets in \mathcal{F} , which we generally allow for families of sets, then each repetition creates a new vertex in the nerve.

Theorem 23 (The optimal colorful fractional Helly theorem for *d*-collapsible complexes). Let *K* be a *d*-collapsible simplicial complex with the set of vertices $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ divided into d+1 disjoint subsets. Let $n_i := |N_i|$ for $i \in [d+1]$ and assume that *K* contains at least $\alpha n_1 \cdots n_{d+1}$ colorful *d*-faces for some $\alpha \in$ (0,1]. Then there is $i \in [d+1]$ such that dim $K[N_i] \ge (1 - (1 - \alpha)^{1/(d+1)})n_i - 1$.

4.1.1 Kim's refined conjecture and further generalization

As a tool for a possible proof of Theorem 21, Kim [53, Conjecture 4.2] suggested the following conjecture. (The notation k_i in Kim's statement of the conjecture is our $r_i + 1$.)

Conjecture 24 ([53]). Let n_i be positive and r_i non-negative integers for $i \in [d+1]$ with $n_i \geq r_i + 1$. Let $\mathcal{F}_1, \ldots, \mathcal{F}_{d+1}$ be families of convex sets in \mathbb{R}^d such that $|\mathcal{F}_i| = n_i$ and there is no subfamily of \mathcal{F}_i of size $r_i + 1$ with non-empty intersection for every $i \in [d+1]$. Then the number of colorful (d+1)-tuples with nonempty intersection is at most

$$n_1 \cdots n_{d+1} - (n_1 - r_1) \cdots (n_{d+1} - r_{d+1}).$$

We explicitly prove this conjecture in a slightly more general setting for *d*-collapsible complexes. (Note that the condition 'no subfamily of size $r_i + 1$ ' translates as 'no r_i -face', that is, 'the dimension is at most $r_i - 1$ '.)

Proposition 25. Let n_i be positive and r_i non-negative integers for $i \in [d+1]$ with $n_i \geq r_i + 1$. Let K be a d-collapsible simplicial complex with the set of vertices $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ divided into d+1 disjoint subsets. Assume that $|N_i| = n_i$ and that dim $K[N_i] \leq r_i - 1$ for every $i \in [d+1]$. Then K contains at most

$$n_1 \cdots n_{d+1} - (n_1 - r_1) \cdots (n_{d+1} - r_{d+1}).$$

colorful d-faces.

Our main technical result. Now, let us present our main technical tool for a proof of Proposition 25 and consequently for a proof of Theorem 23 as well.

We denote by N the set of positive integers whereas \mathbb{N}_0 is the set of nonnegative integers. Let us consider $c \in \mathbb{N}$ and vectors $\mathbf{k} = (k_1, \ldots, k_c), \mathbf{r} = (r_1, \ldots, r_c) \in \mathbb{N}_0^c$ and $\mathbf{n} = (n_1, \ldots, n_c) \in \mathbb{N}^c$ such that $\mathbf{k}, \mathbf{r} \leq \mathbf{n}$. (Here the notation $\mathbf{a} \leq \mathbf{b}$ means that \mathbf{a} is less or equal to \mathbf{b} in every coordinate.) We will also use the notation $k := k_1 + \cdots + k_c$, $n := n_1 + \cdots + n_c$, and $r := r_1 + \cdots + r_c$. Let N be a set with n elements partitioned as $N = N_1 \sqcup \cdots \sqcup N_c$ where $|N_i| = n_i$ for $i \in [c]$. By $\binom{N}{\mathbf{k}}$ we denote the set of all subsets A of N such that $|A \cap N_i| = k_i$ for every $i \in [c]$. Note that $\binom{N}{\mathbf{k}} \subseteq \binom{N}{k}$ where $\binom{N}{k}$ denotes the set of all subsets of N of size k.

Let K be a simplicial complex with the vertex set N as above. We say that a face A of K is **k**-colorful if $A \in \binom{N}{\mathbf{k}}$, that is, $|A \cap N_i| = k_i$ for every $i \in [c]$. The earlier notion of colorful face corresponds to setting c = d + 1 and $\mathbf{k} = \mathbf{1} := (1, \ldots, 1) \in \mathbb{N}^c$. By $f_{\mathbf{k}} = f_{\mathbf{k}}(K)$ we denote the **k**-colorful f-vector of K, that is, the number of **k**-colorful faces in K. Let us further assume that we are given sets $R_i \subseteq N_i$ with $|R_i| = r_i$ for every $i \in [c]$. Let $R = R_1 \sqcup \cdots \sqcup R_c$ and $\overline{R} := N \setminus R$. Then, we define the set system

$$P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) = \left\{ S \in \binom{N}{\mathbf{k}} : |S \cap \overline{R}| \le d \right\}.$$

We remark that $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$ is not a simplicial complex, as it contains only sets in $\binom{N}{\mathbf{k}}$. However, this set system is useful for estimating the number of **k**-colorful faces in a *d*-collapsible complex. By $p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$ we denote the size of $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$, that is, $p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) := |P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})|$.

Theorem 26. For integers $c, d \ge 1$, let K be a d-collapsible simplicial complex with vertex partition $N = N_1 \sqcup \cdots \sqcup N_c$ and let $\mathbf{n} = (n_1, \ldots, n_c) \in \mathbb{N}^c$ be the vector with $n_i = |N_i|$. For $\mathbf{r} = (r_1, \ldots, r_c) \in \mathbb{N}^c$ such that dim $K[N_i] \le r_i - 1$ for $i \in [c]$ and $\mathbf{k} \in \mathbb{N}_0^c$ such that $\mathbf{k} \le \mathbf{n}$ it follows that

$$f_{\mathbf{k}}(K) \leq p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}).$$

Theorem 26 is proved in Section 4.2. Here we show the implications Theorem $26 \Rightarrow$ Proposition 25 and Proposition $25 \Rightarrow$ Theorem 23. In addition, we advertise that Theorem 26 yields further generalizations of Theorem 23. We explain this last part in Section 4.3.

Proof of Proposition 25 modulo Theorem 26. We use Theorem 26 with c = d + 1and $\mathbf{k} = \mathbf{1}$. Then it is sufficient to compute $p_{\mathbf{1}}(\mathbf{n}, d, \mathbf{r})$. On the one hand, the size of $\binom{N}{\mathbf{1}}$ is $n_1 \dots n_{d+1}$. On the other hand, A belongs to $\binom{N}{\mathbf{1}} \setminus P_{\mathbf{1}}(\mathbf{n}, d, \mathbf{r})$ if and only if $|A \cap (N_i \setminus R_i)| = 1$ for every $i \in [d+1]$. Then, the number of such A is $(n_1 - r_1) \cdots (n_{d+1} - r_{d+1})$. Combining these observations we obtain the required formula

$$p_1(\mathbf{n}, d, \mathbf{r}) = n_1 \dots n_{d+1} - (n_1 - r_1) \dots (n_{d+1} - r_{d+1}).$$

Proof of Theorem 23 modulo Proposition 25. By contradiction, let us assume that for every $i \in [d+1]$ we get dim $K[N_i] < (1 - (1 - \alpha)^{1/(d+1)})n_i - 1$. Let us set $r_i := \dim K[N_i] + 1 < (1 - (1 - \alpha)^{1/(d+1)})n_i$. Then Proposition 25 gives that the number of colorful *d*-faces is at most

$$\prod_{i=1}^{d+1} n_i - \prod_{i=1}^{d+1} (n_i - r_i) < \prod_{i=1}^{d+1} n_i - (1 - (1 - (1 - \alpha)^{1/(d+1)}))^{d+1} \prod_{i=1}^{d+1} n_i = \alpha \prod_{i=1}^{d+1} n_i$$

which is a contradiction due to the strict inequality on the first line.

4.2 Colorful exterior algebra

Now we extend the tools introduced in Section 2.2 to the colorful setting. Set $V = \mathbb{R}^N$. From now on, let us assume that N is an n-element set decomposed into c-color classes, $N = N_1 \sqcup \cdots \sqcup N_c$. (The total order on N in this case starts with elements of N_1 , then continues with elements of N_2 , etc.) Let A

be an $N \times N$ transition matrix from $(e_j)_{j \in N}$ to $(g_j)_{j \in N}^3$. We pick A so that it is a block-diagonal matrix with blocks corresponding to individual N_i . That is, $A_{N_i|N_j}$ is a zero matrix whenever $i \neq j$. On the other hand, as shown by Kalai [46, Section 2], it is possible to pick each $A_{N_i|N_i}$ so that $(g_j)_{j \in N_i}$ is an orthonormal basis of the subspace of V generated by $(e_j)_{j \in N_i}$ and each square submatrix of $A_{N_i|N_i}$ has full rank. Therefore, from now on, we assume that we picked A and the vectors g_j this way. (Such a block matrix, for c = 2, is previously mentioned in [66].)

Similarly as in the beginning of the chapter, let us set $\mathbf{n} = (n_1, \ldots, n_c)$ so that $n_i = |N_i|$ for $i \in [c]$; for simplicity, let us assume that each N_i is nonempty—that is, \mathbf{n} is a *c*-tuple of positive integers. Let us also consider another *c*-tuple $\mathbf{k} = (k_1, \ldots, k_c)$ of non-negative integers such that $\mathbf{k} \leq \mathbf{n}$ and we set $k := k_1 + \cdots + k_c$. Then by $\bigwedge^{\mathbf{k}} V$ we mean the subspace of $\bigwedge V$ generated by $(e_S)_{S \in \binom{N}{\mathbf{k}}}$; recall that $\binom{N}{\mathbf{k}}$ is the set of all subsets A of N such that $|A \cap N_i| = k_i$ and that $\binom{N}{\mathbf{k}} \subseteq \binom{N}{k}$. Thus we also get that $\bigwedge^{\mathbf{k}} V$ is a subspace of $\bigwedge^k V$. In addition, due to our choice of $(g_j)_{j \in N}$ we get that $g_S \in \bigwedge^{\mathbf{k}} V$ if $S \in \binom{N}{\mathbf{k}}$. In addition det $A_{S|T} = 0$ if $T \in \binom{N}{k} \setminus \binom{N}{\mathbf{k}}$ because $A_{S|T}$ is in this case a block matrix such that some of the blocks is not a square. Thus the formula (2.2) simplifies to

$$g_S = \sum_{T \in \binom{N}{\mathbf{k}}} \det A_{S|T} e_T. \tag{4.1}$$

Proof of Theorem 26. For $\mathbf{k} \in \mathbb{N}^c$ such that $k \leq d$ we have that $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) = \binom{N}{\mathbf{k}}$, thus the theorem follows trivially. On the other hand, if k > r, then $k_i > r_i$ for some *i* and consequently $f_{\mathbf{k}}(K) = 0$ due to our assumption dim $K[N_i] \leq r_i - 1$; therefore the theorem again follows trivially. From now on we assume $d + 1 \leq k \leq r$. (We also use the notation for the sets R, \overline{R} and R_i with $|R_i| = r_i$ as in the definition of $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$.)

Let us define the following subspaces of $\bigwedge^{\mathbf{k}} V$

$$A_{\mathbf{k}} := \left\{ m \in \bigwedge^{\mathbf{k}} V : \left(\forall T \in \binom{R}{k-d} \right) g_{T} \llcorner m = 0 \right\},$$

and

$$W_{\mathbf{k}} := \operatorname{span} \left\{ e_S \in \bigwedge^{\mathbf{k}} V : S \in \binom{N}{\mathbf{k}} \text{ and } S \in K \right\}$$

from the definition it follows that the colorful *f*-vector and the dimension of $W_{\mathbf{k}}$ coincide, i.e. $f_{\mathbf{k}} = \dim(W_{\mathbf{k}})$.

We claim that

$$\dim(A_{\mathbf{k}}) \ge \left| \binom{N}{\mathbf{k}} \right| - p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}).$$

Indeed, if $S \in \binom{N}{\mathbf{k}}$ such that $S \notin P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$, then $|S \cap \overline{R}| > d$. As $S \subseteq R \sqcup \overline{R} = N$ and |S| = k we have that $|S \cap R| < k - d$. If $T \in \binom{R}{k-d}$ we have that $S \not\supseteq T$; therefore $g_T \sqcup g_S = 0$. From this it follows that $g_S \in A_{\mathbf{k}}$ and finally the claim because $g_S \in \bigwedge^{\mathbf{k}} V$.

³Although in Section 2.2 we have chosen to use f's to denote a new basis, in this chapter we will use g's to avoid possible confusion with the f-vector.

The core of the proof is to show $A_{\mathbf{k}} \cap W_{\mathbf{k}} = \{0\}$. Once we have this, we get $f_k(K) = \dim(W_k) \leq \dim \bigwedge^{\mathbf{k}} V - \dim A_{\mathbf{k}} \leq p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$ which proves the theorem.

For contradiction, let $m \in A_k \cap W_k$ be a non-zero element. Because $m \in W_k$, it can be written as $m = \sum \alpha_S e_S$ where the sum is over all $S \in \binom{N}{k}$ such that $S \in K$. Let K_0, \ldots, K_ℓ be a sequence of simplicial complexes showing *d*-collapsibility of *K* as in the definition of *d*-collapsible complex. In addition, due to [46, Lemma 3.2], it is possible to assume that K_i arises from K_{i-1} by so called *special* elementary *d*-collapse which is either a removal of a maximal face of dimension at most d-1or the minimal face (the face *L* in the definition) has dimension exactly d-1. Now let us consider the first step from K_{i-1} to K_i such that a face $U \in \binom{N}{k}$ with nonzero α_U is eliminated. Denote by *L* and *M* the faces determining the collapse as in the definition. We have $L \subseteq U \subseteq M$, $|M| \ge |U| = k > d$ and therefore |L| = d(equivalently, dim L = d - 1), because the collapse is special. For $T \in \binom{R}{k-d}$ let $\mathbf{t} = (t_1, \ldots, t_c) \in \mathbb{N}^c$ be such that $t_i = |T \cap N_i|$. Then $g_T = \sum_{P \in \binom{N}{k}} \det(A_{T|P})e_P$ via (4.1). We also need to simplify the expression $\langle e_L, g_T \sqcup e_S \rangle$ for $S \in \binom{N}{k}$. We obtain

$$\langle e_L, g_T \llcorner e_S \rangle = \langle e_L \land g_T, e_S \rangle = \sum_{P \in \binom{N}{\mathbf{t}}} \det(A_T|_P) \langle e_L \land e_P, e_S \rangle$$
(4.2)

If $S \not\supseteq L$ then $\langle e_L \wedge e_P, e_S \rangle = 0$ for all P, and therefore $\langle e_L, g_T \llcorner e_S \rangle = 0$. If $S \supseteq L$ then $\langle e_L \wedge e_P, e_S \rangle = 0$ unless $P = S \setminus L$ and therefore $\langle e_L, g_T \llcorner e_S \rangle = \langle e_L \wedge e_S \backslash L, e_S \rangle \det(A_{T|S \setminus L})$.

Since $m \in A_k$, for arbitrary $T \in \binom{R}{k-d}$ we get

$$0 = \langle e_L, g_T \llcorner m \rangle = \sum_{S \in \binom{N}{\mathbf{k}} : S \in K} \alpha_S \langle e_L, g_T \llcorner e_S \rangle = \sum_{S \in \binom{N}{\mathbf{k}} : S \in K_{i-1}} \alpha_S \langle e_L, g_T \llcorner e_S \rangle$$
$$= \sum_{S \in \binom{N}{\mathbf{k}} : S \supseteq L} \alpha_S \langle e_L, g_T \llcorner e_S \rangle = \sum_{S \in \binom{N}{\mathbf{k}} : M \supseteq S \supseteq L} \alpha_S \langle e_L \land e_S \land L, e_S \rangle \det(A_{T|S \setminus L})$$

where the third equality follows from the fact that $\alpha_S = 0$ for $S \in K \setminus K_{i-1}$ due to our choice of K_{i-1} and the last two equalities follow from our earlier simplification of $\langle e_L, g_T \llcorner e_S \rangle$. (We also use that the expressions $S \supseteq L$ and $M \supseteq S \supseteq L$ are equivalent as M is the unique maximal face containing L.) We also have $U \in \binom{N}{\mathbf{k}}$ with $M \supseteq U \supseteq L$ for which $\alpha_U \neq 0$ as well as

We also have $U \in \binom{N}{\mathbf{k}}$ with $M \supseteq U \supseteq L$ for which $\alpha_U \neq 0$ as well as $\langle e_L \wedge e_{U \setminus L}, e_U \rangle$ is nonzero (the latter one equals ± 1). Therefore the expression above is a linear dependence of the columns of $C_{k-d}(A_{R|M \setminus L})$. However, we will also show that the columns of $C_{k-d}(A_{R|M \setminus L})$ are linearly independent, thereby getting a contradiction. Via Lemma 2, it is sufficient to check that the columns of $A_{R|M \setminus L}$ are linearly independent. Because A is a block-matrix with blocks $A_{N_i|N_i}$, we get that $A_{R|M \setminus L}$ is a block matrix with blocks $A_{R_i|(M \setminus L) \cap N_i}$. Thus it is sufficient to check that the columns are independent in each block. But this follows from our assumptions of how we picked A in each block, using that $|R_i| = r_i \ge |(M \setminus L) \cap N_i|$ as $|M \cap N_i| \le r_i$ due to our assumption dim $K[V_i] \le r_i - 1$.

4.3 k-colorful fractional Helly theorem

Theorem 26 allows to generalize Theorem 23 in two more directions.

The first generalization of Theorem 23 is already touched in the beginning of the chapter. We can deduce an analogy of Theorem 23 for **k**-colorful faces (instead of just colorful *d*-faces) where $\mathbf{k} = (k_1, \ldots, k_c) \in \mathbb{N}_0^c$ is some vector with $c \geq 1$. For example, if d = 2, $\mathbf{k} = (2, 1, 1)$ and we understand the partition of $N = N_1 \sqcup N_2 \sqcup N_3$ as coloring the vertices of K red, green, or blue. Then we seek for number of faces that contain two red vertices, one green vertex and one blue vertex.

For the second generalization, let us first observe that in the conclusion of Theorem 23 there is the same coefficient $1 - (1 - \alpha)^{1/(d+1)}$ independently of *i*. However, in the notation of Theorem 23, we may also seek for *i* such that dim $K[N_i] \geq \beta_i n_i + 1$ where $\boldsymbol{\beta} = (\beta_1, \ldots, \beta_c) \in (0, 1]^c$ is some fixed vector. Then for given $\boldsymbol{\beta}$, we want to find the lowest $\alpha \in (0, 1]$ with which we reach the conclusion analogous as in Theorem 23. This is a natural analogy of various Ramsey type statements: for example, if the edges of a complete graph G with at least 9 vertices are colored blue or red, then the graph contains either a blue copy of the complete graph on 3 vertices or a red copy of the complete graph on 4 vertices.

For the purpose of stating the generalization, let us set

$$L_{\mathbf{k}}(d) := \{ \boldsymbol{\ell} = (\ell_1, \cdots \ell_c) \in \mathbb{N}_0^c \colon \ell_1 + \cdots + \ell_c \le d \text{ and } \ell_i \le k_i \text{ for } i \in [c] \}$$
(4.3)

and

$$\alpha_{\mathbf{k}}(d,\boldsymbol{\beta}) := \sum_{\boldsymbol{\ell}=(\ell_1,\dots,\ell_c)\in L_{\mathbf{k}}(d)} \prod_{i=1}^c \binom{k_i}{\ell_i} (1-\beta_i)^{\ell_i} (\beta_i)^{k_i-\ell_i}.$$
(4.4)

Theorem 27. Let $c, d \ge 1$ and $\mathbf{k} = (k_1, \ldots, k_c) \in \mathbb{N}_0^c$ be such that $k := k_1 + \cdots + k_c \ge d + 1$. Let K be a d-collapsible simplicial complex with the set of vertices $N = N_1 \sqcup \cdots \sqcup N_c$ divided into c disjoint subsets. Let $n_i := |N_i|$ for $i \in [c]$ and assume that K contains at least $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) | \binom{N}{\mathbf{k}} | \mathbf{k}$ -colorful faces for some $\boldsymbol{\beta} = (\beta_1, \ldots, \beta_c) \in (0, 1]^c$. Then there is $i \in [c]$ such that dim $K[N_i] \ge \beta_i n_i - 1$.

The formula (4.4) for $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta})$ in Theorem 27 is, unfortunately, a bit complicated. However, this is the optimal value for α in the theorem. We first prove Theorem 27 and then we will provide an example showing that for every d, \mathbf{k} and $\boldsymbol{\beta}$ as in the theorem, the value for α cannot be improved. The remark below is a probabilistic interpretation of (4.4). (This, for example, easily reveals that $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) \in (0, 1]$ for given parameters and will help us with checking monotonicity in $\boldsymbol{\beta}$.)

Remark 28. Consider a random experiment where we gradually for each *i* pick k_i numbers $x_1^i, \ldots, x_{k_i}^i$ in the interval [0, 1] independently at random (with uniform distribution). Let ℓ_i be the number of x_j^i which are greater than β_i and let us consider the event $A_{\mathbf{k}}(d, \boldsymbol{\beta})$ expressing that $\ell_1 + \cdots + \ell_c \leq d$. Then $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta})$ is the probability $\mathbf{P}[A_{\mathbf{k}}(d, \boldsymbol{\beta})]$.

Indeed, the probability that the number of x_j^i which are greater than β_i is exactly ℓ_i is given by the expression beyond the sum in (4.4). Therefore, we need to sum this over all options giving $\ell_1 + \cdots + \ell_c \leq d$ and $\ell_i \leq k_i$.

In the proof of Theorem 27 we will need the following slightly modified proposition. We relax 'at least' to 'more than' while we aim at strict inequality in the conclusion—this innocent change will be a significant advantage in the proof. On the other hand, after this change we can drop the assumption $k \ge d + 1$. But this is only a cosmetic change, because the proposition below is vacuous if $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) = 1$ which in particular happens if k < d + 1.

Proposition 29. Let $c, d \ge 1$ and $\mathbf{k} = (k_1, \ldots, k_c) \in \mathbb{N}_0^c$. Let K be a d-collapsible simplicial complex with the set of vertices $N = N_1 \sqcup \cdots \sqcup N_c$ divided into c disjoint subsets. Let $n_i := |N_i|$ for $i \in [c]$ and assume that K contains more than $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) | \binom{N}{\mathbf{k}} | \mathbf{k}$ -colorful faces for some $\boldsymbol{\beta} = (\beta_1, \ldots, \beta_c) \in (0, 1]^c$. Then there is $i \in [c]$ such that dim $K[N_i] > \beta_i n_i - 1$.

First we show how Theorem 27 follows from Proposition 29 by a limit transition. Then we prove Proposition 29.

Proof of Theorem 27 modulo Proposition 29. Let us consider $\varepsilon > 0$ such that $\beta - \varepsilon \in (0, 1]^c$ for $\varepsilon = (\varepsilon, \ldots, \varepsilon) \in (0, 1]^c$.

First, we need to check $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) > \alpha_{\mathbf{k}}(d, \boldsymbol{\beta}-\boldsymbol{\varepsilon})$. For this we will use Remark 28 and we also use $k \geq d+1$. It is easy to check $A_{\mathbf{k}}(d, \boldsymbol{\beta}) \supseteq A_{\mathbf{k}}(d, \boldsymbol{\beta}-\boldsymbol{\varepsilon})$ which gives $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) \geq \alpha_{\mathbf{k}}(d, \boldsymbol{\beta}-\boldsymbol{\varepsilon})$. In order to show the strict inequality, it remains to show that $A_{\mathbf{k}}(d, \boldsymbol{\beta}) \setminus A_{\mathbf{k}}(d, \boldsymbol{\beta}-\boldsymbol{\varepsilon})$ has positive probability. Consider the output of the experiment when each $x_i^j \in (\beta_i - \varepsilon, \beta_i)$. This output has positive probability ε^k . In addition, this output belongs to $A_{\mathbf{k}}(d, \boldsymbol{\beta}-\boldsymbol{\varepsilon})$ whereas it does not belong to $A_{\mathbf{k}}(d, \boldsymbol{\beta}-\boldsymbol{\varepsilon})$ (because $k \geq d+1$) as required.

This means, that we can apply Proposition 29 with $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta} - \boldsymbol{\varepsilon})$ as we know that K has at least $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) |\binom{N}{\mathbf{k}}|$ **k**-colorful faces by assumptions of Theorem 27 which is more than $\alpha_{\mathbf{k}}(d, \boldsymbol{\beta} - \boldsymbol{\varepsilon}) |\binom{N}{\mathbf{k}}|$. We obtain dim $K[N_i] > (\beta_i - \boldsymbol{\varepsilon})n_i - 1$. By letting $\boldsymbol{\varepsilon}$ to tend to 0, we obtain the required dim $K[N_i] \geq \beta_i n_i - 1$. \Box

Boosting the complex. In the proof of Proposition 29, we will need the following procedure for boosting the complex. For a given complex K with vertex set $N = N_1 \sqcup \cdots \sqcup N_c$ partitioned as usual, and a non-negative integer m we define the complex $K_{\langle m \rangle}$ as a complex with the vertex set $N \times [m] = N_1 \times [m] \sqcup \cdots \sqcup N_c \times [m]$ whose maximal faces are of the form $S \times [m]$, where S is a maximal face of K. We will also use the notation $\delta_{\mathbf{k}}(K) := f_{\mathbf{k}}(K)/|\binom{N}{\mathbf{k}}|$ for the density of \mathbf{k} -colorful faces of K.

Lemma 30. Let K be a simplicial complex with vertex partition $N = N_1 \sqcup \cdots \sqcup N_c$ and $\mathbf{k} = (k_1, \ldots, k_c) \in \mathbb{N}_0^c$, then

- (i) $\delta_{\mathbf{k}}(K_{\langle m \rangle}) \geq \delta_{\mathbf{k}}(K)$; and
- (ii) if K is d-collapsible, then $K_{\langle m \rangle}$ is d-collapsible as well.

Proof. Let us start with the proof of (i). If $\delta_{\mathbf{k}}(K) = 0$ there is nothing to prove. Thus we may assume that $\delta_{\mathbf{k}}(K) > 0$ (equivalently $f_{\mathbf{k}}(K) > 0$) and consequently we have that $|N_i| \ge k_i$. Let us interpret $\delta_{\mathbf{k}}(K)$ as the probability that a random \mathbf{k} -tuple of vertices in N is a simplex of K, and we interpret $\delta_{\mathbf{k}}(K_{\langle m \rangle})$ analogously. Let $\pi \colon N \times [m] \to N$ be the projection to the first coordinate. Now, let U be a \mathbf{k} -tuple of vertices in $N \times [m]$ taken uniformly at random. Considering the set $\pi(U) \subseteq N$, it need not be a \mathbf{k} -tuple (this happens exactly when two points in U have the same image under π) but it can be extended to a **k**-tuple W using that $|N_i| \ge k_i$ for every *i*. Let W be an extension of $\pi(U)$ to a **k**-tuple, taken uniformly at random among all possible choices. Because of the choices we made, W is in fact a **k**-tuple of vertices in N taken uniformly at random. (Note that the choices done in each N_i or $N_i \times [m]$ are independent of each other.) Altogether, using **P** for probability, we get

$$\delta_{\mathbf{k}}(K_{\langle m\rangle}) = \mathbf{P}[U \in K_{\langle m\rangle}] = \mathbf{P}[\pi(U) \in K] \ge \mathbf{P}[W \in K] = \delta_{\mathbf{k}}(K).$$

This shows (i).

For (ii), we follow the idea of splitting a vertex from [4, Proposition 14(i)] which proves a similar statement for *d*-Leray complexes. For a complex *K* and a vertex $v \in K$ let $K^{v \to v_1, v_2}$ be a complex obtained from *K* by splitting the vertex *v* into two newly introduced vertices v_1 and v_2 . That is, if *V* is the set of vertices of *K*, then the set of vertices of $K^{v \to v_1, v_2}$ is $(V \cup \{v_1, v_2\}) \setminus \{v\}$ assuming $v_1, v_2 \notin V$. The maximal simplices of $K^{v \to v_1, v_2}$ are obtained from maximal simplices *S* of *K* by replacing *v* with v_1 and v_2 , if *S* contains *v* (otherwise *S* is kept as it is). Our aim is to show that if *K* is *d*-collapsible, then $K^{v \to v_1, v_2}$ is *d*-collapsible as well. This will prove (ii) because $K_{\langle m \rangle}$ can be obtained from *K* by repeatedly splitting some vertex. For the proof, we extend the notation $K^{v \to v_1, v_2}$ by setting $K^{v \to v_1, v_2} = K$ if *v* does not belong to *K*.

Let $K_0 = K, K_1, \ldots, K_{\ell} = \emptyset$ be a sequence such that K_i arises from K_{i-1} by an elementary *d*-collapse. Our task is to show that $K_{i-1}^{v \to v_1, v_2}$ *d*-collapses to $K_i^{v \to v_1, v_2}$ for $i \in [\ell]$. This will show the claim as $K_{\ell}^{v \to v_1, v_2} = \emptyset$. For simplicity of the notation, we will treat only the elementary *d*-collapse from *K* to K_1 as other steps are analogous. We will assume $v \in K$, as there is nothing to do if $v \notin K$.

Let L and M be the faces from the elementary d-collapse. That is, dim $L \leq d-1$; M is the unique maximal face in K which contains L and K_1 is obtained from K by removing all faces that contain L, including L. We will distinguish three cases according to whether $v \in L$ or $v \in M$.

If $v \notin M$ (which implies $v \notin L$), then M is the unique maximal face containing L in $K^{v \to v_1, v_2}$ and the elementary d-collapse removing L and all its superfaces yields $K_1^{v \to v_1, v_2}$.

If $v \in M$ while $v \notin L$, then $(M \cup \{v_1, v_2\}) \setminus \{v\}$ is the unique maximal face containing L in $K^{v \to v_1, v_2}$ and the elementary d-collapse removing L and all its superfaces yields $K_1^{v \to v_1, v_2}$.

Finally, if $v \in M$ and $v \in L$, then we need to perform the *d*-collapse from $K^{v \to v_1, v_2}$ to $K_1^{v \to v_1, v_2}$ by two elementary steps; see Figure 4.1. First we realize that $(M \cup \{v_1, v_2\}) \setminus \{v\}$ is the unique maximal face containing $(L \cup \{v_1\}) \setminus \{v\}$ in $K^{v \to v_1, v_2}$. Because dim $(L \cup \{v_1\}) \setminus \{v\} = \dim L$, we can perform an elementary *d*-collapse removing $(L \cup \{v_1\}) \setminus \{v\}$ and all its superfaces obtaining a complex K'. In K' we have that $(M \cup \{v_2\}) \setminus \{v\}$ is the unique maximal face containing $(L \cup \{v_2\}) \setminus \{v\}$. After removing $(L \cup \{v_2\}) \setminus \{v\}$ and all its superfaces, we get desired $K_1^{v \to v_1, v_2}$ (note that in this case $K_1^{v \to v_1, v_2}$ is indeed obtained from $K^{v \to v_1, v_2}$ by removing $(L \cup \{v_1\}) \setminus \{v\}$, $(L \cup \{v_2\}) \setminus \{v\}$ and all their superfaces). \Box


Figure 4.1: Collapses from $K^{v \to v_1, v_2}$ to $K_1^{v \to v_1, v_2}$ if $v \in L$.

Density of $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$. Now, we will provide a formula for the density of $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$. In the following computations we also set

$$\delta_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) = p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) / \left| \binom{N}{\mathbf{k}} \right|$$

using the notation from the definition of $P_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$. We get

$$p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) = \left| \left\{ S \in \binom{N}{\mathbf{k}} : |S \cap \bar{R}| \le d \right\} \right|$$
$$= \sum_{\ell = (\ell_1, \dots, \ell_c) \in L_{\mathbf{k}}(d)} \left| \left\{ S \in \binom{N}{\mathbf{k}} : |S_i \cap \bar{R}_i| = l_i \right\} \right|$$
$$= \sum_{\ell = (\ell_1, \dots, \ell_c) \in L_{\mathbf{k}}(d)} \prod_{i=1}^c \binom{n_i - r_i}{l_i} \binom{r_i}{k_i - l_i}.$$

Then, using $(x)_m := x \cdot (x-1) \cdots (x-(m-1))$, the density is given by

$$\delta_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) = \frac{p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})}{\prod_{i=1}^{c} \binom{n_{i}}{k_{i}}} = \frac{\sum_{\ell = (\ell_{1}, \dots, \ell_{c}) \in L_{\mathbf{k}}(d)} \prod_{i=1}^{c} \binom{k_{i}}{\ell_{i}} (n_{i} - r_{i})_{\ell_{i}} (r_{i})_{k_{i} - \ell_{i}}}{\prod_{i=1}^{c} (n_{i})_{k_{i}}}.$$
 (4.5)

Proof of Proposition 29. For contradiction, let us assume that for every $i \in [c]$ we have that $\dim(K[V_i]) \leq \beta_i n_i - 1$. Let us set $r_i := \dim(K[V_i]) + 1 \leq \beta_i n_i$. Note that the conclusion of Theorem 26 can be restated as $\delta_{\mathbf{k}}(K) \leq \delta_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$.

Now we get

$$\begin{split} \delta_{\mathbf{k}}(K) &\leq \liminf_{m \to \infty} \delta_{\mathbf{k}}(K_{\langle m \rangle}) \text{ by Lemma 30(i)} \\ &\leq \liminf_{m \to \infty} \delta_{\mathbf{k}}(m\mathbf{n}, d, m\mathbf{r}) \text{ by Theorem 26 using Lemma 30(ii)} \\ &\leq \liminf_{m \to \infty} \delta_{\mathbf{k}}(m\mathbf{n}, d, \lfloor mn_i\beta_i \rfloor), r_i \leq \beta_i n_i \text{ and } p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r}) \text{ is monotone in } \mathbf{r} \\ &= \liminf_{m \to \infty} \frac{\sum_{\ell = (\ell_1, \dots, \ell_c) \in L_{\mathbf{k}}(d)} \prod_{i=1}^c \binom{k_i}{\ell_i} (mn_i - \lfloor mn_i\beta_i \rfloor)_{\ell_i} (\lfloor mn_i\beta_i \rfloor)_{k_i - \ell_i}}{\prod_{i=1}^c (mn_i)_{k_i}} \text{ by (4.5)} \\ &= \sum_{\ell = (\ell_1, \dots, \ell_c) \in L_{\mathbf{k}}(d)} \prod_{i=1}^c \binom{k_i}{\ell_i} (1 - \beta_i)^{\ell_i} (\beta_i)^{k_i - \ell_i} \\ &= \alpha_{\mathbf{k}}(d, \boldsymbol{\beta}) \end{split}$$

which is a contradiction with the assumptions.

Remark 31. It would be much more natural to try to avoid boosting the complex and show directly $\delta_k(K) \leq \delta_k(\mathbf{n}, d, \mathbf{r}) \leq \alpha_k(d, \boldsymbol{\beta})$ in the proof of Proposition 29. The former inequality follows from Theorem 26. However, the latter inequality turned out to be somewhat problematic for us when we attempted to show it directly from the definition of $\alpha_k(d, \boldsymbol{\beta})$ and from (4.5). Thus, in our computations, we take an advantage of the fact that the computations in the limit are easier.

Tightness of Theorem 27. We conclude this section by showing that the bound given in Theorem 27 is tight.

Let us fix $c, d \in \mathbb{N}$, $\mathbf{k} = (k_1, \ldots, k_c) \in \mathbb{N}_0^c$ with $k := k_1 + \cdots + k_c \geq d + 1$ and $\boldsymbol{\beta} = (\beta_1, \ldots, \beta_c) \in (0, 1]^c$ as in the statement of Theorem 27. Let $0 \leq \alpha' < \alpha_{\mathbf{k}}(d, \boldsymbol{\beta})$. We will find a complex K which contains at least $\alpha' | \binom{N}{\mathbf{k}} | \mathbf{k}$ -colorful faces while dim $K[N_i] < \beta_i n_i - 1$ for every $i \in [c]$ (using the notation from the statement of Theorem 27).

Similarly as in the proof of Theorem 27 let us consider $\varepsilon > 0$ such that $\beta - \varepsilon \in (0, 1]^c$ for $\varepsilon = (\varepsilon, \ldots, \varepsilon) \in (0, 1]^c$. In addition, because $\alpha_{\mathbf{k}}(d, \beta)$ is continuous in β due to its definition (4.4), we may pick ε such that $\alpha' < \alpha_{\mathbf{k}}(d, \beta - \varepsilon)$. For simplicity of notation, let $\beta' = (\beta'_1, \ldots, \beta'_c) := \beta - \varepsilon$.

Now we pick a positive integer m and set $\mathbf{n} = (m, \ldots, m) \in \mathbb{N}^c$, that is, $n_1 = \cdots = n_c = m$ and n = cm in our standard notation. We also set $\mathbf{r} = (r_1, \ldots, r_c)$ so that $r_i := \lfloor \beta'_i m \rfloor$.⁴ We assume that m is large enough so that $r_i \ge k_i$ for each $i \in [c]$. We define families N_i of convex sets in \mathbb{R}^d so that each N_i contains r_i copies of \mathbb{R}^d and $m - r_i$ hyperplanes in general position. We also assume that the collection of all hyperplanes in N_1, \ldots, N_c is in general position. We set K to be the nerve of the family $N = N_1 \sqcup \cdots \sqcup N_c$. In particular K is d-representable (therefore d-collapsible as well).

First, we check that dim $K[N_i] < \beta_i m - 1$ provided that m is large enough. A subfamily of N_i with nonempty intersection contains at most d hyperplanes from N_i . Therefore dim $K[N_i] < r_i + d = \lfloor \beta'_i m \rfloor + d < \beta_i m - 1$ for m large enough.

⁴This choice of **n** will yield a counterexample where each color class has equal size. It would be also possible to vary the sizes.

Next we check that K contains at least $\alpha' | \binom{N}{\mathbf{k}} | \mathbf{k}$ -colorful faces provided that m is large enough. Partitioning N_i so that R_i is the subfamily of the copies of \mathbb{R}^d and \bar{R}_i is the subfamily of hyperplanes, we get

$$f_{\mathbf{k}}(K) = p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$$

from the definition of $p_{\mathbf{k}}(\mathbf{n}, d, \mathbf{r})$. Therefore (4.5) gives

$$\delta_{\mathbf{k}}(K) = \frac{\sum_{\ell=(\ell_1,\dots,\ell_c)\in L_{\mathbf{k}}(d)} \prod_{i=1}^c \binom{k_i}{\ell_i} (m - \lfloor \beta'_i m \rfloor)_{\ell_i} (\lfloor \beta'_i m \rfloor)_{k_i - \ell_i}}{\prod_{i=1}^c (m)_{k_i}}.$$

Passing to the limit (considering the dependency of K on m), we get

$$\lim_{m \to \infty} \delta_{\mathbf{k}}(K) = \sum_{\boldsymbol{\ell} = (\ell_1, \dots, \ell_c) \in L_{\mathbf{k}}(d)} \prod_{i=1}^c \binom{k_i}{\ell_i} (1 - \beta_i')^{\ell_i} (\beta_i')^{k_i - \ell_i} = \alpha_{\mathbf{k}}(d, \boldsymbol{\beta}').$$

Therefore, for *m* large enough *K* contains at least $\alpha' | \binom{N}{\mathbf{k}} |$ **k**-colorful as $\alpha' < \alpha_{\mathbf{k}}(d, \beta')$.

4.4 A topological version?

A simplicial complex K is d-Leray if the *i*th reduced homology group $\tilde{H}_i(L)$ (over \mathbb{Q}) vanishes for every induced subcomplex $L \leq K$ and every $i \geq d$. As we already know, every d-representable complex is d-collapsible, and in addition every d-collapsible complex is d-Leray [87]. Helly-type theorems usually extend to d-Leray complexes and such extensions are interesting because they allow topological versions of Helly-type when collections of convex sets are replaced with good covers. We refer to several concrete examples [39, 51, 4] or to the survey [82].

We believe that it is possible to extend Theorem 23 to *d*-Leray complexes:

As we mentioned in the beginning of the chapter we conjecture that it should be possible to extend Theorem 23 to *d*-Leray complexes and probably Theorem 26 as well. Here we state the conjectured generalization of Theorem 23.

Conjecture 32 (The optimal colorful fractional Helly theorem for *d*-Leray complexes). Let *K* be a *d*-Leray simplicial complex with the set of vertices $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ divided into d+1 disjoint subsets. Let $n_i := |N_i|$ for $i \in [d+1]$ and assume that *K* contains at least $\alpha n_1 \cdots n_{d+1}$ colorful *d*-faces for some $\alpha \in (0, 1]$. Then there is $i \in [d+1]$ such that dim $K[N_i] \ge (1 - (1 - \alpha)^{1/(d+1)})n_i - 1$.

In fact, our original approach how to prove Theorem 23 was to prove directly Conjecture 32. Indications that this could be possible are that both the optimal fractional Helly theorem [4, 50] and the colorful Helly theorem [51] hold for d-Leray complexes. In addition, there is a powerful tool, algebraic shifting, developed by Kalai [50], which turned out to be very useful in attacking similar problems.

In the remainder of this section we briefly survey a possible approach towards Conjecture 32 but also the difficulty that we encountered. Because we do not really prove any new result in this section, our description is only sketchy.

Our starting point is the proof of the optimal fractional Helly theorem for d-Leray complexes. The key ingredient is the following theorem of Kalai [4, Theorem 13].

Theorem 33. Let K be a d-Leray complex and $f_0(K) = n$. Then $f_d(K) > \binom{n}{d+1} - \binom{n-r}{d+1}$ implies $f_{d+r}(K) > 0$ (where f(K) denotes the f-vector of K).

As far as we can judge, the only proof of Theorem 33 in the literature follows from the first and the third sentence in the following remark in [50]:

"It is not hard to see (although it has been overlooked for a long time) that the class of d-Leray complexes (for some d) with complete (d - 1)-dimensional skeletons is precisely the Alexander dual of the class of Cohen-Macaulay complexes. This observation implies that the fact that shifting preserves the Leray property easily follows from the fact that shifting preserves the Cohen-Macaulay property. Moreover, it shows that the characterization of face numbers of d-Leray complexes follows from the corresponding characterization for Cohen-Macaulay complexes."

For completeness we add that the characterization of face numbers of Cohen-Macaulay complexes has been done by Stanley [76]. Given a simplicial complex Kon vertex set N its Alexander dual is a simplicial complex defined as $K^* := \{T \subseteq N : N \setminus T \notin K\}$. We skip the definition of Cohen-Macaulay complex because we will only use it implicitly but we refer, for example, to [50, §4] for more details.

A simplicial complex K on vertex set [n] is called *shifted* if for all integers iand j with $1 \leq i < j \leq n$ and all faces A of K such that $j \in A$ and $i \notin A$, the set $(A \setminus \{j\}) \cup \{i\}$ is a face of K. *Exterior algebraic shifting* is a function that associates to a simplicial complex K a shifted complex $\Delta(K)$, while preserving many interesting invariants of K. Below we list some properties of exterior algebraic shifting that we will use. A simplicial complex is *pure* if all its inclusion-maximal faces have the same dimension.

- **Theorem 34.** (i) [50, Theorem 2.1] Exterior algebraic shifting preserves the *f*-vector.
- (ii) [50, Theorem 4.1] If K is Cohen-Macaulay, then $\Delta(K)$ is Cohen-Macaulay, in particular, pure.
- (iii) [50, 3.5.6] The following equality holds $\Delta(K^*) = \Delta(K)^*$.

The next lemma is a possible replacement of the third sentence in Kalai's remark how to prove Theorem 33. We prove it as motivation for the tools we would need in the colorful scenario.

Lemma 35. Let K be a d-Leray complex on [n] with complete (d-1)-skeleton and let $D = \dim(K) + 1$. Then $K^e \subseteq \Delta_{D-d-1} * \Delta_{n-D+d-1}^{(d-1)}$.

Proof. By the first sentence of Kalai's remark, the Alexander dual K^* of K is a Cohen-Macaulay complex. By the definition of Alexander dual, it has dimension n - d - 2 and contains complete (n - D - 2)-skeleton. Hence, properties (i) and (ii) of Theorem 34 imply that the exterior algebraic shifting $(K^*)^e$ of K^* is a pure shifted complex of dimension n - d - 2 with complete (n - D - 2)-skeleton. If we take any subset A of size n - D - 1 in $(K^*)^e$, then A is a face and by purity there must be a face of size n - d - 1 that contains A. Now, since $(K^*)^e$ is shifted we have that $\{1, 2, \ldots, D - d\} \cup A \in (K^*)^e$. This implies that $\Delta_{D-d-1} * \Delta_{n-D+d-1}^{(n-D-2)} \subseteq (K^*)^e = (K^e)^*$, by Theorem 34(iii). Taking Alexander dual

from both sides proves the first part of the statement. (Using that $(L^*)^* = L$; $L_1 \subseteq L_2 \Rightarrow L_2^* \subseteq L_1^*$; and $(\Delta_{D-d-1} * \Delta_{n-D+d-1}^{(n-D-2)})^* = \Delta_{D-d-1} * \Delta_{n-D+d-1}^{(d-1)}$. \Box

For completeness, Theorem 33 quickly follows from Lemma 35. Indeed, if K is d-Leray such that $f_{d+r}(K) = 0$, then $D := \dim K + 1 \leq d + r$. In addition, we can assume without loss of generality that K contains complete (d-1)-skeleton. Consequently, Lemma 35 gives $f_d(K) = f_d(K^e) \leq f_d(\Delta_{D-d-1} * \Delta_{n-D+d-1}^{(d-1)}) = \binom{n}{d+1} - \binom{n-D+d}{d+1} \leq \binom{n}{d+1} - \binom{n-r}{d+1}$. Now, in order to attack Conjecture 32, we would like to do something similar in

Now, in order to attack Conjecture 32, we would like to do something similar in colorful setting. In particular, we need to preserve the colorful f-vector. Babson and Novik [7] give a definition of colorful algebraic shifting which preserves the colorful f-vector. Nevertheless, the conjecture does not follow immediately from their result as the Alexander dual of a d-Leray complex is not in general balanced.

We show next why a shifting operator preserving d-Leray and colorful f-vector is enough. For it let us introduce the following definition. Let $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ with each N_i having a total order $<_i$. A simplicial complex K with vertex set N is color-shifted if for every $F \in K$, $i \in [d+1]$ and $v \in F \cap N_i$, if $w <_i v$ then $(F \setminus \{v\}) \cup \{w\} \in K$.

Proposition 36. Let K be a color-shifted d-Leray simplicial complex with vertex partition $N = N_1 \sqcup \cdots \sqcup N_{d+1}$ divided into d+1 disjoint sets. Let $n_i = |N_i|$ for $i \in [d+1]$ and assume that K contains at least $\alpha n_1 \cdots n_{d+1}$ for some $\alpha \in (0,1]$. Then there is $i \in [d+1]$ such that dim $K[N_i] \ge (1 - (1 - \alpha)^{1/d+1})n_i - 1$.

Proof. As observed previously it is enough to show if r_1, \ldots, r_{d+1} are positive integers such that dim $K[N_i] \leq r_i - 1$ then $f_1(K) \leq n_1 \cdots n_{d+1} - (n_1 - r_1) \cdots (n_{d+1} - r_{d+1})$. Let R_i denote the first r_i vertices in N_i for each $i \in [d+1]$. Then it is enough to verify that every colorful *d*-face intersects at least one of the R_i 's. Let us assume that this is not the case and let $T = (t_1, \ldots, t_{d+1})$ be such a face, with $t_i \in N_i \setminus R_i$ for every $i \in [d+1]$. Let T_i denote the first t_i element in N_i , in particular $t_i = |T_i| > |R_i| = r_i$. Since K is color-shifted and $T \in K$ we have that $T_1 \times \cdots \times T_{d+1} \subseteq K$. Then $K[\bigcup_{i \in [d+1]} T_i]$ is *d*-Leray and contains every colorful *d*-face. By the colorful Helly theorem of Kalai and Meshulam [51] we can conclude that there exists $i \in [d+1]$ such that $T_i \in K$. This is a contradiction since $r_i - 1 \ge \dim K[N_i] \ge \dim K[T_i] = t_i - 1 > r_i - 1$.

We close this chapter with the following question.

Problem 37. Is there a shifting operation preserving the d-Leray property as well as the colorful f-vector such that the resulting complex is color-shifted?

5. Volume rigidity

Let K be an n-vertex (d-1)-dimensional simplicial complex and $\mathbf{p} : N(K) \to \mathbb{R}^{d-1}$ be a generic mapping of its vertices, in the sense that its (d-1)n coordinates are algebraically independent over \mathbb{Q} . This chapter deals with the infinitesimal version of the following problem: is there a *non-trivial* continuous motion of the vertices starting at \mathbf{p} that preserves the volumes of all the (d-1)-simplices in K? By "non-trivial" we mean that, for some (d-1)-simplex on N(K) that is not in K, its volume would change along the motion. It is easy to show that the continuous and infinitesimal versions coincide for generic embeddings, as is the case for graph rigidity [5].

Volume Rigidity. The signed volume of a (d-1)-face $S = \{v_1, \ldots, v_d\} \in K$ with respect to **p** is given by the determinant of the $d \times d$ matrix

$$M_{\mathbf{p},S} = \begin{pmatrix} \mathbf{p}(v_1) & \dots & \mathbf{p}(v_d) \\ 1 & \dots & 1 \end{pmatrix}.$$

Observe that, for every $1 \leq i \leq d-1$, $1 \leq j \leq d$, the derivative of the signed volume det $M_{\mathbf{p},S}$ with respect to the *i*-th coordinate of $\mathbf{p}(v_j)$ is given by the cofactor $C_{i,j}(M_{\mathbf{p},S})$ — that is, the determinant of the submatrix obtained by removing the *i*-th row and *j*-th column multiplied by $(-1)^{i+j}$.

The volume-rigidity matrix $\mathfrak{V}(K, \mathbf{p})$ of the pair (K, \mathbf{p}) is a $(d-1)n \times f_{d-1}(K)$ matrix, where the columns are indexed by the (d-1)-faces of K, and every vertex is associated with a block of (d-1) rows. The column vector \mathbf{v}_S corresponding to a (d-1)-face $S = \{v_1, \ldots, v_d\} \in K$ is defined by

$$(\mathbf{v}_S)_{v_i,j} = C_{i,j}(M_{\mathbf{p},S}), \ i \in [d], \ j \in [d-1],$$

and 0 elsewhere. Here $(\mathbf{v}_S)_{v_i,j}$ denotes the *j*-th coordinate of \mathbf{v}_S in the block of v_i . In words, \mathfrak{V} is the Jacobian of the function $\mathbf{p} \mapsto (\det M_{\mathbf{p},S})_{S \in K}$, viewing \mathbf{p} as a (d-1)n-dimensional vector.

This matrix was introduced in [61, Appendix A] along with the description of a trivial (d^2-d-1) -subspace of the left kernel of $\mathfrak{V}(K, \mathbf{p})$, arising from the volumepreserving transformations of \mathbb{R}^{d-1} . Concretely, the trivial subspace consists of all (d-1)n-dimensional vectors z obtained by choosing a $(d-1) \times (d-1)$ matrix A whose trace is zero and a vector $u \in \mathbb{R}^{d-1}$, and letting $z_v = A \cdot \mathbf{p}(v) + u$ for every vertex v. The following definition suggests itself.

Definition 38. An *n*-vertex (d-1)-dimensional simplicial complex K is called volume-rigid if

$$\operatorname{rank}(\mathfrak{V}(K,\mathbf{p})) = (d-1)n - (d^2 - d - 1),$$

for a generic $\mathbf{p}: N(K) \to \mathbb{R}^{d-1}$.

Exterior shifting. Algebraic shifting was introduced by Kalai (see e.g. [49] and the survey [50]) and has been studied extensively in algebraic combinatorics. Here we present a variant of exterior shifting. The standard basis $(e_i)_{i \in [n]}$ of \mathbb{R}^n induces the basis $(e_S)_{S \subseteq [n]}$ of its exterior algebra $\bigwedge \mathbb{R}^n$. Consider a generic basis (f_1, \ldots, f_n) of \mathbb{R}^n , where without loss of generality we assume that $f_1 = \mathbf{1} \in \mathbb{R}^n$,

namely, the other $n^2 - n$ coordinates in this basis are algebraically independent over \mathbb{Q} . Consider the *exterior face ring* $\bigwedge K = \bigwedge \mathbb{R}^n / (e_S \colon S \notin K)$, and let qdenote the natural quotient map. Given a partial order < on the power set of [n], define

$$\Delta^{<}(K) = \{ S \subseteq [n] \colon q(f_S) \notin \operatorname{span}_{\mathbb{R}} \{ q(f_T) \colon T < S, \ |T| = |S| \} \}.$$
(5.1)

Of special importance in our case is the partial order $<_p$ defined by

$$S = \{s_1 < \dots < s_m\} \leq_p T = \{t_1 < \dots < t_{m'}\}$$

if m = m' and $s_i \leq t_i$, $\forall i \in [m]$. Corollary 44 asserts that $\Delta^p(K) := \Delta^{<_p}(K)$ is a shifted simplicial complex independent of the generic choice of f. (Note that $\Delta^p(K)$ may have more faces than K.)

5.0.1 Main results.

Our main result is a characterization of volume rigidity in the setting of Kalai's exterior shifting.

Theorem 39. Fix $d \ge 3$. An n-vertex (d-1)-dimensional simplicial complex K is volume-rigid if and only if $\{1, 3, 4, ..., d, n\} \in \Delta^p(K)$.

In the 2-dimensional case we are able to derive the volume rigidity of triangulations of the following surfaces.

Corollary 40. Every triangulation of the 2-sphere, the torus, the projective plane or the Klein bottle is volume rigid. In addition, every triangulation of the 2-sphere and the torus minus a single triangle is also volume-rigid. In particular, every simplicial disc with a 3-vertex boundary is minimally volume-rigid.

In the case of the 2-sphere we give a complete mathematical proof. For the other surfaces, we reduce — via edge contractions á la Whiteley [88] — to irreducible triangulations, whose volume-rigidity we verify numerically.

Hypergraph sparsity was introduced by Streinu and Theran [78], generalizing results on graph sparsity, prominently by White and Whiteley [89] who studied it from a matroid perspective. We say that a (d-1)-complex is $(d-1, d^2 - d - 1)$ -sparse (resp. tight) if every subset A of its vertices of cardinality at least d spans at most $(d-1)|A| - (d^2 - d - 1)$ simplices of dimensions d-1 (resp. and equality holds when A equals the entire vertex set).

Clearly, a vertex subset A spanning more (d-1)-simplices induces a non trivial linear dependence between the columns of $\mathfrak{V}(K, \mathbf{p})$, and it is natural to ask whether this characterizes all the linear dependencies in the volume rigidity matrix. Using Theorem 39, we show that the answer is negative, hence a Laman-type condition for volume-rigidity does not hold true ¹.

Corollary 41. For every $d \ge 3$, there exists a $(d-1, d^2 - d - 1)$ -tight (d-1)complex that is not volume-rigid.

¹Corollary 41 shows that Prop.1 in the preprint [77] from 2007 is a misstatement.

5.0.2 Relation to previous works.

The maximal independent sets of columns of $\mathfrak{V}(\binom{[n]}{d}, \mathbf{p})$, for all generic embeddings \mathbf{p} , form the bases of the same matroid. For d = 2 they correspond to spanning trees, namely the bases in the graphic matroid on $\binom{[n]}{2}$. Kalai [48] introduced for every integer $k \geq 1$ the k-hyperconnectivity matroid on $\binom{[n]}{2}^2$, where k = 1 corresponds to the graphic matroid, and identified its bases in terms of exterior shifting (w.r.t. the lexicographic order): G is a basis if and only if the edges of $\Delta^{\text{lex}}(G)$ form the initial segment that ends with $\{k, n\}$, w.r.t. the lex-order.

Here, in Theorem 39, rather then increasing the dimension of the embedding space and staying with graphs, we increase also the dimension of the pure complex, by the same number, and characterize the bases of the resulted *d*-volume-rigidity matroid in terms of exterior shifting w.r.t. the partial order $<_p$.

The fact that $(d-1, d^2 - d - 1)$ -sparse complexes form the independent sets of a matroid on $\binom{[n]}{d}$ was asserted in [58, 89]. Additional matroidal and algorithmic properties of sparsity matorids were studied by Streinu and Theran in [78, 79]. By Corollary 41, the $(d-1, d^2 - d - 1)$ -sparsity matroid strictly contains the (d-1)volume-rigidity matroid for all $d \geq 3$. It would be interesting to find further combinatorial conditions that once imposed on the bases of the sparsity-matroid would give the bases of the volume-rigidity matroid.

The remainder of the chapter is organized as follows. In Section 5.1 we establish the connection between volume rigidity and exterior shifting, and prove Theorem 39. Afterwards, in Section 5.2 we investigate the effect of local moves on volume rigidity and prove Corollary 40. In the following Section 5.3 we prove Corollary 41, and we conclude in Section 5.4 with some related open problems.

5.1 Volume rigidity and $\Delta^p(\cdot)$

This section is devoted to studying the basic properties of the shifted complex $\Delta^{p}(K)$, and to establishing the connection between $\Delta^{p}(K)$ and K's volume rigidity.

5.1.1 Basic properties of $\Delta^p(\cdot)$

We start by briefly exploring some useful properties of the complex $\Delta^p(K)$ that appears in Theorem 39. Given a partially ordered set (poset) (P, <) and an element $x \in P$ we denote by $P_{<,x}$ the prefix $\{y \in P : y \leq x\}$.

Claim 42. Let (P, <) be a poset and $x \in P$, then there exists a linear extension $<_l of <$ such that $P_{<_l,x} = P_{<,x}$.

Proof. View the sets $A = P_{<,x}$ and $B = P \setminus P_{<,x}$ as posets with the partial order induced by <. Extend each of these posets linearly, and concatenate the extensions such that the elements in A are smaller than those in B.

²The k-hyperconnectivity matroid is derived from an embedding of the vertex set into \mathbb{R}^k . Studying higher hyperconnectivity translates to increasing the dimension of the embedding space.

We will mainly work with the partial order $<_p$ on the power set of [n] and denote the set of its linear extensions by \mathcal{L} . We usually denote an element in \mathcal{L} by $<_l$ and the corresponding shifted complex by $\Delta^l(K) := \Delta^{<_l}(K)$.

Claim 43. $\Delta^p(K) = \bigcup_{\leq l \in \mathcal{L}} \Delta^l(K).$

Proof. On the one hand, if $S \in \Delta^{l}(K)$ for some $<_{l} \in \mathcal{L}$ then $q(f_{S})$ is not spanned by $B_{l,S} := \{q(f_{T}) : T <_{l} S\}$, which contains the vector set $B_{p,S}$. Therefore, by the definition of $\Delta^{<}$ in (5.1), we find that $\Delta^{p}(K) \supseteq \Delta^{l}(K)$. On the other hand, for every $S \in \Delta^{p}(K)$, there exists by the previous claim a linear extension $<_{l} \in \mathcal{L}$ satisfying $B_{p,S} = B_{l,S}$ hence $S \in \Delta^{l}(K)$.

Corollary 44. For every simplicial complex K there holds that $\Delta^p(K)$ is a shifted simplicial complex independent of the choice of the generic basis f. In addition, $\Delta^p(K) = K$ if K is shifted. \Box

That $\Delta^p(K)$ is downwards closed follows exactly as in the proof for $\Delta^{\text{lex}}(K)$. The rest of Corollary 44 follows immediately from the above decomposition of $\Delta^p(K)$ and the fact that the basic properties of algebraic shifting in [49] – being shifted, and independence the the generic f chosen– hold in every linear extensions of $<_p$, as remarked in [49, p.58].

5.1.2 Volume rigidity and $\Delta^p(\cdot)$

We are now ready to prove Theorem 39. We denote $S_0 = \{1, 3, ..., d, n\}$ and observe that the prefix $B := \{T \leq_p S_0 : |T| = d\}$ consists of the subsets [d] and $[d] \setminus \{i\} \cup \{v\}$ for $2 \leq i \leq d$ and $d+1 \leq v \leq n$. We define a linear transformation $\psi : \bigoplus_{i=2}^{d} \bigwedge^1 \mathbb{R}^n \to \bigwedge^d \mathbb{R}^n$ given by

$$\psi(m_2,\ldots,m_d) = \sum_{i=2}^d f_{[d]\setminus\{i\}} \wedge m_i.$$

Lemma 45. The image of ψ is spanned by $\{f_T : T \in B\}$, and its kernel is $(d^2 - d - 1)$ -dimensional.

Proof. The fact that $f_T \in \operatorname{im}(\psi)$ for every $T \in B$ can be shown directly. Indeed, $\psi(0, ..., 0, f_d) = f_{[d]}$ and by taking $m_i = f_v$, $m_{i'} = 0 \quad \forall i' \neq i$ we have that $\psi(0, ..., f_v, ..., 0) = f_{[d] \setminus \{i\} \cup \{v\}}$ for $2 \leq i \leq d$ and $d + 1 \leq v \leq n$. To show that these 1 + (n - d)(d - 1) linearly independent vectors span the image of ψ , we will construct $d^2 - d - 1$ linearly independent vectors in ker ψ which actually completes the proof by the rank-nullity theorem since $(1 + (n - d)(d - 1)) + (d^2 - d - 1) = n(d - 1)$.

First, for every $2 \leq i \leq d$ and $j \in [d] \setminus \{i\}$ consider the vector defined by setting $m_i = f_j$ and $m_{i'} = 0$ for every $i' \neq i$. Then,

$$\psi(m_2, ..., m_d) = f_{[d] \setminus \{i\}} \land f_j = 0$$

since $j \in [d] \setminus \{i\}$. This amounts to $(d-1)^2$ vectors in ker ψ , and the remaining d-2 are given by vectors of the form $m_i = a_i f_i$, $2 \leq i \leq d$, where the scalars $a_2, ..., a_d$ satisfy $\sum_{i=2}^d (-1)^i a_i = 0$. Indeed,

$$\psi(m_2, ..., m_d) = \sum_{i=2}^d a_i f_{[d] \setminus \{i\}} \wedge f_i = \sum_{i=2}^d a_i (-1)^{d-i} f_{[d]} = 0.$$

The linear independence of these $d^2 - d - 1$ vectors follows directly from the linear independence of $f_1, ..., f_d$.

Proof of Theorem 39. Identify the vertices of K with the set [n]. W.l.o.g. assume that $f_{d-1}(K) \geq (d-1)n - (d^2 - d - 1)$, as otherwise K is not volume-rigid and $\{1, 3, 4, ..., d, n\} \notin \Delta^p(K)$. In addition, suppose that the generic embedding $\mathbf{p}: N(K) \to \mathbb{R}^{d-1}$ is obtained from the vectors $f_2, ..., f_d$ in the generic basis of \mathbb{R}^n by taking $(f_i)_v = \mathbf{p}(v)_{i-1}$ for every $2 \leq i \leq d$ and $v \in [n]$.

Consider the (i, v)-unit vector $e_{i,v} = (m_2, ..., m_d)$ in the domain of ψ , for $2 \leq i \leq d$ and $v \in [n]$, defined by $m_i = e_v$ and $m_{i'} = 0$, $\forall i' \neq i$. Then,

$$\psi(e_{i,v}) = f_{[d] \setminus \{i\}} \wedge e_v$$

= $(-1)^{d-i} f_1 \wedge \cdots \wedge f_{i-1} \wedge e_v \wedge f_{i+1} \wedge \cdots \wedge f_d$
= $(-1)^{d-i+d-1} f_2 \wedge \cdots \wedge f_{i-1} \wedge e_v \wedge f_{i+1} \wedge \cdots \wedge f_d \wedge f_1.$

Let $S = \{v_1, ..., v_d\} \subset [n]$. Clearly, for the inner product on $\wedge \mathbb{R}^n$ with orthonormal basis $\{e_S : S \subset [n]\}$, we have $\langle e_S, \psi(e_{i,v}) \rangle = 0$ if $v \notin S$. Otherwise, by the identification of **p** with $f_2, ..., f_d$ above and $f_1 = \mathbf{1}$, if $v = v_j$ then $\langle e_S, \psi(e_{i,v}) \rangle$ is equal to $(-1)^{i-1}$ times the determinant of the matrix that is obtained from $M_{\mathbf{p},S}$ by replacing its (i - 1)-th row with the *j*-th *d*-dimensional all-ones row vector. Consequently,

$$\langle e_S, \psi(e_{i,v}) \rangle = (-1)^{i-1} C_{i-1,j}(M_{\mathbf{p},S}).$$

Thus, by letting $q : \bigwedge \mathbb{R}^n \longrightarrow \bigwedge K$ be the natural quotient map, and by choosing the basis $\{e_S : S \in K\}$ for $\bigwedge K$, we find that the $f_{d-1}(K) \times (d-1)n$ matrix representation of $q \circ \psi$ is equal — up to multiplying some of its columns by -1and reordering them — to the transpose of the volume-rigidity matrix $\mathfrak{V}(K, \mathbf{p})$. Therefore, K is volume-rigid if and only if dim ker $(q \circ \psi) = d^2 - d - 1$. In other words, K is not volume-rigid if and only if there exists a non-zero $f \in \mathrm{im}(\psi)$ such that q(f) = 0. By the characterization of ψ 's image in Lemma 45, f can be written as a non-trivial linear combination $f = \sum_{T \in B} \lambda_T f_T$.

To conclude the proof we claim that q(f) = 0 for some $f \in \text{span}\{f_T : T \in B\}$ if and only if $S_0 \notin \Delta^p(K)$. Indeed, on one direction, $q(\sum_{T \in B} \lambda_T f_T) = 0$ implies that for some $T \in B$, $q(f_T)$ is a linear combination of its predecessors in $<_p$. By (5.1), $T \notin \Delta^p(K)$ and since $\Delta^p(K)$ is shifted then $S_0 \notin \Delta^p(K)$. On the other hand, by manipulating the linear combination which asserts that $S_0 \notin \Delta^p(K)$, we obtain a non-zero vector $f = \sum_{T \in B} \lambda_T f_T$ satisfying q(f) = 0.

5.2 Volume rigidity, local moves and homology

We turn to study the effect of local combinatorial moves on volume rigidity. We start by proving a volume-rigidity analog of Whiteley's vertex splitting [88], by which he showed that every triangulation of the 2-sphere has a 3-rigid 1-skeleton.

Lemma 46 (Edge contraction). Let K be a pure (d-1)-dimensional simplicial complex, $e = \{u, w\} \in K$ such that at least (d-1) facets in K contain e. Let K' to be the simplicial complex obtained from K by contracting the edge e, i.e. by identifying the vertex u with w, and removing duplicates. If K' is volume rigid then so is K.

Proof. Without loss of generality assume that u < w are the first among the n vertices of K, as the vertex labels do not effect volume-rigidity. We will construct an auxiliary $(d-1)n \times f_{d-1}(K)$ matrix A such that

$$\operatorname{rank} \mathfrak{V}(K, \mathbf{p}) \ge \operatorname{rank}(A) = (d-1)n - (d^2 - d - 1).$$

First, we replace the position of the vertex w, i.e. $\mathbf{p}(w)$, by the position of the vertex u, i.e. $\mathbf{p}(u)$. Formally we define a new (non-generic) placement of vertices \mathbf{p}' that coincides with \mathbf{p} on all vertices except w on which we set it to equal to $\mathbf{p}(u)$. Clearly, since \mathbf{p} is generic, there holds rank $\mathfrak{V}(K, \mathbf{p}) \geq \operatorname{rank} \mathfrak{V}(K, \mathbf{p}')$. To obtain A, we add the rows in $\mathfrak{V}(K, \mathbf{p}')$ corresponding to the vertex u to the rows corresponding to the vertex w, an operation that does not change the rank.

We first claim that the submatrix of A which corresponds to the columns of the facets L containing $e = \{u, w\}$ is supported on the rows corresponding to u. Indeed, if $e \subseteq S = \{v_1, \ldots, v_d\} \in L$ then for $v_j \in S$ such that $v_j \neq \{u, w\}$ we have that each entry $A_{v_j,i;S} = C_{i,j}(M_{\mathbf{p}',S}) = 0$ because $M_{\mathbf{p}',S}$ has two identical columns as $\mathbf{p}'(u) = \mathbf{p}'(w)$. On the other hand, because we added the rows corresponding to vertex u to the rows corresponding to vertex w, we have that

$$A_{w,i;S} = C_{i,1}(M_{\mathbf{p}',S}) + C_{i,2}(M_{\mathbf{p}',S}) = 0.$$

This follows from our assumption that u and w are the first two vertices hence their cofactors in $M_{\mathbf{p}',S}$ have opposite signs, and they in fact cancel-out since and $\mathbf{p}'(u) = \mathbf{p}'(w)$.

Second, we claim that the submatrix $A_{u,L}$ of A corresponding to the d-1 rows of u and the columns of L has a full rank of d-1. We derive this claim by the assumption that $|L| \ge d-1$ and the fact that $(\mathbf{p}'(v) : v \ne u)$ is generic. Indeed, consider a vector $x \in \mathbb{R}^{d-1}$ in the left kernel of $A_{u,L}$. A brief calculation yields that the orthogonality of x and the column in $A_{u,L}$ corresponding to the facet $S = \{u, w, v_2, ..., v_d\}$ is equivalent to x being in the span of $\mathbf{p}'(v_i) - \mathbf{p}'(w)$, i =2, ..., d. By the assumption that \mathbf{p} is generic, such $|L| \ge d-1$ constraints are only satisfied by x = 0 hence

$$\operatorname{rank}(A_{u,L}) = d - 1. \tag{5.2}$$

Third, consider the complement submatrix $A' := A_{\{u\}^c, L^c}$ whose rows correspond to all the vertices except u, and columns to all the facets that are not in L. We observe that A' contains as a submatrix the generic volume rigidity matrix $\mathfrak{V}(K', \mathbf{p})$ — where \mathbf{p} is viewed here as a generic embedding of $N(K') = N(K) \setminus \{u\}$ into \mathbb{R}^{d-1} . Indeed, every facet S of K' arises from a facet \hat{S} of K.

- If u ∉ Ŝ then S = Ŝ and the columns in 𝔅(K', p) and A' corresponding to S are clearly equal.
- Otherwise, if u ∈ Ŝ then S = Ŝ ∪ {w} \ {u}, and by the construction of A
 in which p'(u) = p'(w) and the rows of u are added to the rows of w
 we have that the column in A' created from Ŝ is equal to the column of S
 in 𝔅(K', p).

Note that A' may contain some duplicate columns — in case there are two facets that differ only in the vertices of e — but, regardless, our observation that A' contains $\mathfrak{V}(K', \mathbf{p})$ as a submatrix implies that

$$\operatorname{rank}(A') = (n-1)(d-1) - (d^2 - d - 1).$$
(5.3)

In conclusion, A takes the form

$$A = \begin{array}{cc} L & L^c \\ u \\ \{u\}^c \begin{pmatrix} A_{u,L} & * \\ 0 & A' \end{pmatrix}, \end{array}$$

and by combining (5.2) and (5.3) we find that

$$\operatorname{rank}(\mathfrak{V}(K,\mathbf{p}) \ge \operatorname{rank}(A) = \operatorname{rank}(A_{u,L}) + \operatorname{rank}(A') = n(d-1) - (d^2 - d - 1),$$

as claimed.

The next two lemmas are direct analogs of basic results in graph rigidity [6] asserting that gluing preserves volume-rigidity. We include their proofs for completeness.

Lemma 47. Let K be (d-1)-volume-rigid, $v \notin N(K)$ and $S \subseteq N(K)$ such that $|S| \ge d$, then $K \cup (v * {S \choose d-1})$ is (d-1)-volume-rigid.

Proof. The vertex v is in at least d-1 facets of $L = K \cup (v * \binom{S}{d-1})$. The volume rigidity matrix of L is of the form

$$\begin{array}{ccc} K & S * v \\ V & \left(\mathfrak{V}(K, \mathbf{p}) & * \\ v & 0 & N \end{array} \right), \end{array}$$

where the matrix N has d-1 rows and at least d-1 columns. Because of general position N has full rank, i.e. rank(N) = d-1. Because K is (d-1)-volume-rigid we have that $rank(\mathfrak{V}(K,\mathbf{p})) = n(d-1) - d(d-1) + 1$. Then, $rank(\mathfrak{V}(L,\mathbf{p})) = rank(\mathfrak{V}(K,\mathbf{p})) + rank(N) = (n+1)(d-1) - d(d-1) + 1$ and consequently L is (d-1)-volume-rigid.

Lemma 48 (Union of volume-rigid complexes). Let K and L be (d-1)-volumerigid complexes such that $|N(K) \cap V(L)| \ge d$. Then $K \cup L$ is (d-1)-volume-rigid.

Proof. Because K and L are (d-1)-volume-rigid we can assume that each of them has a complete (d-1)-skeleton on its respective vertex set. Then $K \cup L$ contains the vertex spanning subcomplex Q obtained from K by adding one vertex v at a time, adding the facets $\binom{S}{d-1} * v$ at step v, where $S = N(L) \cap N(K)$. This subcomplex is (d-1)-volume-rigid at each step by application of the previous lemma. In particular, Q is (d-1)-volume-rigid, hence so is $K \cup L$.

5.2.1 Volume rigidity of surfaces

Barnette and Edelson [13, 14] proved that every compact surface without boundary admits only finitely many *irreducible* triangulations, namely, triangulations where every edge contraction would result in a simplicial complex not homeomorphic to the given surface. Thus, by Lemma 46, in order to conclude that for a given surface S every simplicial complex that triangulates it is volume-rigid, it is enough to verify if for the irreducible triangulations of S. Those are known for the surfaces indicated in Corollary 40: one for the 2-sphere (namely the boundary of a tetrahedron), two for the projective plane [12], 21 for the torus [55] and 29 for the Klein bottle [56, 80]. Clearly the boundary of the tetrahedron is volume-rigid, and we verified by computer that the irreducible triangulations K of the other surfaces mentioned above are volume-rigid – for this it was enough to find some embedding $\mathbf{p}_K : N(K) \longrightarrow \mathbb{R}^2$ such that $\operatorname{rank}(\mathfrak{V}(K, \mathbf{p}_K) = 2|N(K)| - 5$.

Remark 49. The fact that every triangulation K of the 2-sphere is volume-rigid follows also from combining the 3-hyperconnectivity of its graph with the Cohen-Macaulay property. Indeed, the first property says that $\{3, |N(K)|\} \in \Delta^{\text{lex}}(K)$, and as K is Cohen-Macaluay then $\Delta^{\text{lex}}(K)$ is pure and hence $\{1, 3, |N(K)|\} \in \Delta^{\text{lex}}(K)$ as $\Delta^{\text{lex}}(K)$ is shifted, which implies, by Claim 43, that $\{1, 3, |N(K)|\} \in \Delta^p(K)$, and we are done by Theorem 39.

To prove the second part of Corollary 40 we are left to show that removing one triangle from a triangulated 2-sphere or torus preserves volume-rigidity, done next. A pure simplicial complex is a *minimal cycle* (over some coefficients commutative ring \mathbb{F}) if there exists an \mathbb{F} -linear combination of its facets whose boundary vanishes, and no proper nonempty subset of its facets has this property. For example, every triangulation of a compact connected surface (resp. and orientable) is a minimal cycle over \mathbb{Z}_2 (resp. \mathbb{Z}).

Lemma 50. If K is a (d-1)-dimensional volume rigid minimal cycle over \mathbb{Z} , then $K \setminus S$ is volume rigid for every $S \in K$.

We first give short proof for the special case d = 3 and conclude the proof of Corollary 40. Afterwards, we give a more technical proof for the general case.

Proof of Lemma 50 (d = 3). As K is a minimal cycle over \mathbb{Z} , its 2-dimensional homology with \mathbb{R} -coefficients is one dimensional, and for each facet S of K, for $K \setminus S$ this homology vanishes. By the translation of homology in terms of algebraic shifting, $\Delta^{\text{lex}}(K) \ni \{2, 3, 4\} \notin \Delta^{\text{lex}}(K \setminus S)$, and as shifting preserves containment we conclude

$$\Delta^{\operatorname{lex}}(K \setminus S) = \Delta^{\operatorname{lex}}(K) \setminus \{\{2, 3, 4\}\}.$$

Note that in this dimension³ $T <_p \{1,3,n\}$ iff $T <_{\text{lex}} \{1,3,n\}$, and thus: $\{1,3,n\} \in \Delta^p(K)$ (by Theorem 39), hence $\{1,3,n\} \in \Delta^{\text{lex}}(K)$, and by the displayed equality above also $\{1,3,n\} \in \Delta^{\text{lex}}(K \setminus S)$, so finally $\{1,3,n\} \in \Delta^p(K \setminus S)$, equivalently, $K \setminus S$ is volume-rigid. \Box

Proof of Corollary 40. This is immediate from Lemma 46, Lemma 50 for the case d = 3, and the discussion in the beginning of §5.2.1.

We conclude this section with a more direct proof of Lemma 50 for all $d \ge 3$.

Proof of Lemma 50. For two subsets S and $T = S \setminus \{v\}$ of [n] that differ by one element, denote sign $(T, S) := (-1)^j$ if v is the *j*-th element in S. We prove the following stronger statement. Let $z \in \mathbb{Z}^{f_{d-1}(K)}$ be a (d-1)-dimensional chain in

³For d > 3, $\{T : T <_p \{1, 3, 4, \dots, d, n\}\}$ is smaller than $\{T : T <_{\text{lex}} \{1, 3, 4, \dots, d, n\}\}$.

K, and $\partial z \in \mathbb{Z}^{f_{d-2}(K)}$ be its boundary, i.e. $(\partial z)_T = \sum_{T \subset S} \operatorname{sign}(T, S) \cdot z_S$. Then, for every vertex v and every $i \in [d-1]$

$$\left(\mathfrak{V}(K,\mathbf{p})\cdot z\right)_{v,i} = (-1)^{d+i} \sum_{F} \operatorname{sign}(F, F \cup \{v\}) \cdot \det N_{\mathbf{p},F,i} \cdot (\partial z)_{F \cup \{v\}}, \quad (5.4)$$

where (i) the summation is over the (d-3)-faces F of K that belong to the link of v and, (ii) the $(d-2) \times (d-2)$ matrix $N_{\mathbf{p},F,i}$ is obtained from the $(d-1) \times (d-2)$ matrix $(\mathbf{p}(v) : v \in F)$ by removing its *i*-th row. In particular, if z is a generator of the (d-1)-homology of a minimal cycle K, then z is also a non-trivial linear dependence between the columns of $\mathfrak{V}(K, \mathbf{p})$. Therefore, removing a (d-1)-face from K does not change the rank of the volume rigidity matrix, hence if K is volume rigid then so is $K \setminus \{S\}$. To derive (5.4), note that

$$(\mathfrak{V}(K,\mathbf{p})\cdot z)_{v,i} = \sum_{v\in S\in K} z_S \cdot C_{i,j} M_{\mathbf{p},S}$$

= $\sum_S z_S \cdot (-1)^{i+j} \sum_{j'\in[d]\setminus\{j\}} (-1)^{d-1+j'-\mathbf{1}_{j'>j}} \det N_{\mathbf{p},S\setminus\{v_j,v_j'\},i}$ (5.5)

Indeed, suppose that $S = \{v_1, ..., v_d\}$ and $v = v_j$ and expand the (i, j)-th minor of $M_{\mathbf{p},S}$ by the last row (of ones). Denote $T := S \setminus \{v_{j'}\}$ and $F := T \setminus \{v_j\}$, and we easily observe that $\operatorname{sign}(F, T) = (-1)^{j-\mathbf{1}_{j'} < j}$. Therefore, by changing the order of summation in (5.5) we find that

$$(\mathfrak{V}(K,\mathbf{p})\cdot z)_{v,i} = (-1)^{d+i} \sum_{F} \operatorname{sign}(F,T) \det N_{\mathbf{p},F,i} \sum_{T \subset S} \operatorname{sign}(T,S) \cdot z_S,$$

as claimed.

5.3 Volume rigidity and sparsity

Proof of Corollary 41. Let $d \ge 3$ and K be the (d-1)-dimensional simplicial complex obtained from the graph $K_{3,3}$ by iterating the cone operation d-2times. Then K is $(d-1, d^2 - d - 1)$ -sparse. (Indeed, $K_{3,3}$ is (2,3)-sparse, and if a pure (k-1)-dimensional simplicial complex is $(k-1, k^2 - k - 1)$ -sparse then its cone is $(k, (k+1)^2 - (k+1) - 1)$ -sparse.) Thus, by completing it to a basis in the $(d-1, d^2 - d - 1)$ -sparsity matroid, we find a basis K' containing K, so K' is $(d-1, d^2 - d - 1)$ -tight.

In order to show that K' is not volume-rigid, by Theorem 39 it is enough to show that

$$\{1, 2, \dots, d-2, d+1, d+2\} \in \Delta^p(K'),\$$

as $\{1, 2, ..., d-2, d+1, d+2\} \leq_p \{1, 3, 4, ..., d, n\}$ and using tightness.

The displayed equation above follows from basic properties of this shifting operator, proved in the same way as for exterior shifting w.r.t. the lex-order:

- If K is a subcomplex of K' then $\Delta^p(K) \subseteq \Delta^p(K')$.
- Cone and Δ^p commute, namely, if K = v * L for a simplicial complex L then $\Delta^p(K) = 1 * (\Delta^p(L) + 1)$.

Here for a family F of subsets of [m], $F + 1 := \{B + 1 : B \in F\}$, and $B + 1 := \{i + 1 : i \in B\}$ (so $\emptyset + 1 = \emptyset$). To conclude the proof it is left to note that $\{3,4\} \in \Delta^{\text{lex}}(K_{3,3})$ and hence, by Claim 43, also $\{3,4\} \in \Delta^p(K_{3,3})$.

5.4 Concluding remarks

We end up with some related open problems. An obvious one is to extend Corollary 40 to include all surface triangulations.

Conjecture 51. Every triangulation of a compact connected surface without boundary, minus a single triangle, is volume-rigid.

The problem we face in applying Fogelsanger's decomposition [32] (see also [23, Sec.3.3]) to volume rigidity of surfaces is that the pieces in the decomposition include triangle faces not existing in the original triangulation, and thus the gluing lemmas we could prove, e.g. Lemma 48, are not strong enough to settle Conjecture 51.

It is known that for every triangulation K of the 2-sphere on n vertices, minus a single triangle, its exterior shifting $\Delta^{\text{lex}}(K) = \Delta^p(K)$ consists exactly of the triangle 13n and all the triangles that are smaller than it in the lex-order, and their subsets. This is a sufficient condition for volume rigidity by Theorem 39. The following conjecture deals with a higher-dimensional counterpart of this fact.

Conjecture 52. For every $d \ge 3$, every triangulation K of the (d-1)-sphere minus a single (d-1)-simplex is volume rigid.

It is also natural to ask whether the stronger property of $\{1, 3, 4, \ldots, d, n\} \in \Delta^{\text{lex}}(K)$ holds true. This is known, and tight, for stacked spheres [65] (also [67, Example 2.1.8]). Let us remark that the conclusion $\{1, 3, 4, \ldots, d, n\} \in \Delta^s(K)$ for Kalai's symmetric shifting operator $\Delta^s(\cdot)$ is equivalent to the hard-Lefschetz isomorphism from degree 1 to degree d - 1 in a generic Artinian reduction of the Stanley-Reisner ring of K over the field of reals; the later isomorphism was proved recently by Adiprasito [1].

Back to general complexes,

Problem 53. For every dimension, find a combinatorial characterization of the corresponding volume-rigidity matroid.

The combinatorial characterization problem is important for the *d*-rigidity matroid (and is open for $d \ge 3$). The *d*-rigidity of a graph *G* on *n* vertices is equivalent to $\{d, n\} \in \Delta^s(G)$. In view of this fact, we ask:

Problem 54. Define a version of symmetric shifting $\Delta^{sp}(\cdot)$ and find a matroid on $\binom{[n]}{d}$ such that its bases K are exactly those satisfying $\Delta^{sp}(K) = \{T : T \leq_p \{1, 3, 4, \ldots, d, n\}\}.$

An additional direction to explore is the volume rigidity of a (d-1) dimensional simplicial complex K in $\mathbb{R}^{d'}$ for $d' \geq d-1$. That is, let $\mathbf{p} : N(K) \to \mathbb{R}^{d'}$ be generic, and ask whether there is a non-trivial motion of the vertices that preserves all the volumes of K's (d-1)-dimensional simplices in $\mathbb{R}^{d'}$. The case d = 2 corresponds to the standard framework rigidity in $\mathbb{R}^{d'}$, and the case d' = d-1 is the volume rigidity notion we study in this chapter. Several natural questions on the remaining cases $2 < d \leq d'$ arise: what are the trivial motions in this setting? Is there a characterization of (d-1)-volume rigidity in $\mathbb{R}^{d'}$ in terms of algebraic shifting?

6. Erdős-Ko-Rado for Simplicial Complexes

A set system \mathcal{F} is said to be *pairwise-intersecting* if for every pair of its members $F, F' \in \mathcal{F}$ we have that $F \cap F' \neq \emptyset$. What is the largest cardinality of a family of pairwise-intersecting sets? A now-classic result of Erdős, Ko, and Rado answers this question if the sets all have the same number of elements, and are otherwise unrestricted.

Theorem 55 (Erdős, Ko, and Rado [28]). Let $k \leq n/2$. If \mathcal{F} is a family of pairwise-intersecting subsets of [n], each with k elements, then $|\mathcal{F}| \leq {n-1 \choose k-1}$.

If $|\mathcal{F}|$ achieves the upper bound and k < n/2, then \mathcal{F} consists of all the kelement subsets containing some fixed element.

That is, under the above hypotheses a family of maximal size of pairwiseintersecting objects is given by a family with a common intersection, i.e., $\bigcap \mathcal{F} = \bigcap_{F \in \mathcal{F}} F \neq \emptyset$. Moreover, under slightly stronger hypotheses, these are the only such families. Hilton and Milner [40] later gave an improved upper bound on the size of pairwise-intersecting families that do not all contain a common element.

Theorem 56 (Hilton and Milner [40]). Let $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a pairwise-intersecting family such that $\cap \mathcal{F} = \emptyset$. Then,

$$|\mathcal{F}| \le \binom{n-1}{k-1} - \binom{n-(k+1)}{k-1} + 1.$$

Hilton and Milner introduced the following generalization of pairwiseintersecting families. Two set systems \mathcal{F} and \mathcal{G} are said to be *cross-intersecting* if for every $F \in \mathcal{F}$ and $G \in \mathcal{G}$ we have that $F \cap G \neq \emptyset$. In the same article Hilton and Milner provided the following upper bound for cross-intersecting families of sets all of which members are of the same size.

Theorem 57. Let $k \leq n/2$ and $\mathcal{F}, \mathcal{G} \subseteq {\binom{[n]}{k}}$ be a pair of non-empty crossintersecting families. Then,

$$|\mathcal{F}| + |\mathcal{G}| \le {\binom{n}{k}} - {\binom{n-k}{k}} + 1.$$

There is a large number of generalizations of the above theorems. We focus on one in particular. Holroyd and Johnson asked at the 1997 British Combinatorial Conference [45] about whether an analogue of the Erdős-Ko-Rado property holds for independent sets in cyclic and similar graphs. Talbot showed the answer to be "yes" in a strong sense.

Theorem 58 (Talbot [81]). Let n, k, r be positive integers such that $k \leq n/(r+1)$. Let G be the undirected graph with vertex set \mathbb{Z}_n and edges consisting of those x, ysuch that either $x - y \mod (n)$ or $y - x \mod (n)$ is in $\{1, \ldots, r\}$.

If \mathcal{F} is a family of pairwise-intersecting independent sets of G, each with k elements, then $|\mathcal{F}|$ is at most the size of the family \mathcal{B} of all independent sets with k elements containing 0. If $|\mathcal{F}|$ achieves the upper bound and $n \neq 2k + 2$, then \mathcal{F} is \mathcal{B} up to relabeling the vertices.

Holroyd and Talbot asked whether similar results hold for independent sets in other graphs G. Since the independent sets of a graph form a simplicial complex, Borg extended this question to arbitrary simplicial complexes. Before stating the conjecture, let us restate Theorems 55, 56 and 57 in the language of simplicial complexes. For this purpose, we will first recall some basic definitions for simplicial complexes. Let K be a simplicial complex and v a vertex in K, the link of v in K is the simplicial complex $lk(v, K) = \{F \in K : v \notin F, F \cup \{v\} \in K\}$. For example, let $\Delta_{n-1} = \{F \subseteq [n]\}$ denote the (n-1)-simplex, then

$$lk(1, \Delta_{n-1}) = \{F \subseteq [n] \colon 1 \notin F, F \cup 1 \subseteq [n]\}.$$

This last set system is in bijection with Δ_{n-2} . Consequently,

$$f_{k-2}(\operatorname{lk}(1,\Delta_{n-1})) = f_{k-2}(\Delta_{n-2}) = \binom{n-1}{k-1}.$$

Where for a simplicial complex K, $f_k(K)$ denotes the number of k-faces of K. Now we can restate Theorem 55.

Theorem 59 (Restatement of Theorem 55). Let $k \leq n/2$. If $\mathcal{F} \subseteq K = \Delta_{n-1}$ is a family of pairwise-intersecting (k-1)-faces, then $|\mathcal{F}| \leq f_{k-2}(\operatorname{lk}(1,K)) = \binom{n-1}{k-1}$.

If $|\mathcal{F}|$ achieves the upper bound and k < n/2, then \mathcal{F} consists of all the (k-1)-faces containing some fixed vertex. That is, $\mathcal{F} = \{F \cup \{v\} : F \in lk(v, K), |F| = k-1\}$ for some $v \in N$.

Now we need to recall the notion of an induced simplicial complex. Let K be a simplicial complex with vertex set N and let $S \subseteq N$. The induced simplicial complex of K on S is $K[S] = \{F \in K : F \subseteq S\}$. Finally, we can restate Theorems 56 and 57.

Theorem 60 (Restatement of Theorem 56). Let $\mathcal{F} \subseteq K = \Delta_{n-1}$ be a pairwiseintersecting family of (k-1)-faces such that $\bigcap \mathcal{F} = \emptyset$. Then,

$$|\mathcal{F}| \le f_{k-2}(\operatorname{lk}(1,K)) - f_{k-2}(\operatorname{lk}(1,K[[n] \setminus [k+1]])) + 1.$$

Theorem 61 (Restatement of Theorem 57). Let $\mathcal{F}, \mathcal{G} \subseteq K = \Delta_{n-1}$ be a pair of non-empty cross-intersecting families of (k-1)-faces with $k \leq n/2$. Then,

$$|\mathcal{F}| + |\mathcal{G}| \le f_{k-1}(K) - f_{k-1}(K[[n] \setminus [k]]) + 1.$$

It is natural to ask if one can replace in the above reformulations the underlying (n-1)-simplex K by any arbitrary simplicial complex. There are counterexamples for k around the size of the minimal facet cardinality, but not for somewhat smaller k. We recall that a facet is an inclusion-wise maximal face. The following conjecture states this requirement precisely.

Conjecture 62 (Holroyd and Talbot [42], extended by Borg to arbitrary simplicial complexes [16]). Let K be a simplicial complex whose smallest facet has d vertices, and let $k \leq d/2$. If \mathcal{F} is a family of pairwise-intersecting faces of K, each with k elements, then there is some vertex v of K so that $|\mathcal{F}| \leq f_{k-2}(\operatorname{lk}(v, K))$. If k < d/2 and $|\mathcal{F}|$ achieves the upper bound, then \mathcal{F} consists of the faces containing some vertex v.

If a simplicial complex K satisfies the upper bound of Conjecture 62 at a specified value of k, then we say that K is k-EKR. If every pairwise-intersecting family of maximum size has a common intersection, then we say that K is *strictly* k-EKR. Since originally this conjecture was posed for the independence complex of a graph, i.e., the simplicial complex whose faces are the independence sets of a graph, we extend these definitions to graphs as follows. We abuse terminology to say that a graph is (strictly) k-EKR if its independence complex has the same property. For a graph G, we will denote its independence complex by Ind(G).

There has been considerable work on Conjecture 62. Hurlbert and Kamat showed [43] that any chordal graph with an isolated vertex satisfies the upper bound of Conjecture 62. Holroyd, Spencer and Talbot showed [41] that the following families of graphs satisfy the above upper bound: the disjoint union of $n \geq k$ complete graphs, each of order at least 2; the r-th power of a path on n vertices; the disjoint union of n > 2k complete graphs, cycles and paths. Holroyd and Talbot showed [42] that the independence complex of a disjoint union of cliques and disjoint union of a pair of complete multipartite graphs satisfy the above upper bound as well. Borg showed [16] that the conjecture is true for shifted simplicial complexes. Moreover, he showed that if the minimal facet cardinality of a simplicial complex K is at least $(k-1)\binom{3k-3}{2}+k$, then K satisfies k-EKR and strict k-EKR. See also [68] for EKR type property on facets of flag complexes. Woodroofe showed more generally [91] that any sequentially Cohen-Macaulay near-cone satisfies the upper bound of Conjecture 62. Regarding this last result we want to remark that the class of sequentially Cohen-Macaulay simplicial complexes is a broad class that includes the independence complexes of chordal graphs and many others [24, 62, 90]. In particular the independence complex of a chordal graph with an isolated vertex is a sequentially Cohen-Macaulay near-cone. These notions will be defined in subsections 6.2.2 and 6.2.3. Neither Hurlbert and Kamat nor Woodroofe addressed the strict k-EKR property for this type of simplicial complexes.

The main purpose of the current chapter is to fill in this gap. We show:

Theorem 63. Let $2 \le k < d/2$. If the simplicial complex K is a sequentially Cohen-Macaulay near-cone with minimal facet cardinality d, then K is strictly k-EKR. That is, the pairwise-intersecting families of maximum size consist of all (k-1)-faces containing an apex vertex.

Under stronger hypothesis on the underlying simplicial complex, so called t-fold near-cone, we are able to generalize Theorem 56 and Theorem 57. For example, the independence complex of a chordal graph with t isolated vertices $\{w_1, \ldots, w_t\}$ is a sequentially Cohen-Macaulay t-fold near-cone w.r.t. (w_1, \ldots, w_t) .

Theorem 64. Let $2 \leq k < d/2$. Let K be a sequentially Cohen-Macaulay (k+1)-fold near-cone w.r.t. $W = (w_1, \ldots, w_{k+1})$ and with minimal facet cardinality d. Let $\mathcal{F} \subseteq K$ be a pairwise-intersecting family of (k-1)-faces with $\cap \mathcal{F} = \emptyset$, then

 $|\mathcal{F}| \le f_{k-2}(\operatorname{lk}(w_1, K)) - f_{k-2}(\operatorname{lk}(w_1, K)[N \setminus W]) + 1.$

Theorem 65. Let $k \leq d/2$. Let K be a sequentially Cohen-Macaulay k-fold near-cone w.r.t. W and with minimal facet cardinality d. Let $\mathcal{F}, \mathcal{G} \subseteq K$ be a pair of non-empty cross-intersecting families of (k-1)-faces, then

$$|\mathcal{F}| + |\mathcal{G}| \le f_{k-1}(K) - f_{k-1}(K[N \setminus W]) + 1.$$

In fact, we are able to show a stronger version of these theorems using an algebraic notion to lower bound the minimal facet cardinality, so called depth of a simplicial complex. This notion will be defined in subsection 6.2.3.

Theorem 66. Let $k \ge 2$. If K is a near-cone with apex vertex w and $k < \frac{\operatorname{depth}(K)+1}{2}$, then K is strictly k-EKR. That is, the pairwise-intersecting families of maximum size consist of all (k-1)-faces containing an apex vertex.

Theorem 67. Let K be a (k + 1)-fold near-cone w.r.t. $W = (w_1, \ldots, w_{k+1})$ and $2 \le k \le \frac{\operatorname{depth}(K)+1}{2}$. Let $\mathcal{F} \subseteq K$ be a pairwise-intersecting family of (k - 1)-faces such that $\bigcap \mathcal{F} = \emptyset$. Then,

$$|\mathcal{F}| \leq f_{k-2}(\operatorname{lk}(w_1, K)) - f_{k-2}(\operatorname{lk}(w_1, K)[N \setminus W]) + 1.$$

Theorem 68. Let K be a k-fold near-cone w.r.t. W and $k \leq \frac{\operatorname{depth}(K)+1}{2}$. Let $\mathcal{F}, \mathcal{G} \subseteq K$ be a pair of non-empty cross-intersecting families of (k-1)-faces, then

$$|\mathcal{F}| + |\mathcal{G}| \le f_{k-1}(K) - f_{k-1}(K[N \setminus W]) + 1.$$

The novelty of our techniques is to combine algebraic and combinatorial shifting operations. We also make use of some of the ideas behind recent proofs of the Hilton-Milner theorem [36, 44].

This chapter is organized as follows. In Sections 6.1 we review graph classes which independence complex falls into the class of sequentially Cohen-Macaulay complexes, in particular proving Conjecture 62 for such graph classes. In Section 6.2 we recall the necessary background for the current chapter. In Section 6.3 we give the proofs of Theorems 64 and 67. In Section 6.4 we present the proofs of Theorems 63 and 66. In Section 6.5 we present the proofs of Theorems 65 and 68.

Notation. Throughout this chapter we will be using N to denote the vertex set of a simplicial complex when an order on the vertices is not required. On the other hand, when a total order on the vertices is needed we will use $[n] = \{1, \ldots, n\}$. We will usually use letters such as v, w to refer to vertices coming from N and i, j for those coming from [n].

6.1 Applications

In this section we recall some families of simplicial complexes that belong to the class of sequentially Cohen-Macaulay simplicial complexes. By applying our main results of this chapter we are able to derive the strict k-EKR property in the cases for which only the k-EKR property was previously known. In particular, proving completely the conjecture for such families. Under additional hypothesis we are able to derive a Hilton-Milner type result as well as an upper bound on the size of cross-intersecting families for such families of simplicial complexes.

Let us first recall the definition of a join for simplicial complexes. Given a pair of simplicial complexes L and K, their *join*, denoted by L * K, is the simplicial complex given by $\{F \sqcup T : F \in L, T \in K\}$ where \sqcup denotes the disjoint union.

We would like to emphasize that even though the vertices of K and L might not be disjoint, we make them disjoint when considering their join. For example, by taking $N(K) \times \{1\}$ and $N(L) \times \{2\}$ to be the new vertex sets for the simplicial complexes K and L. Here we have denoted the vertex set of a simplicial complex K by N(K). If L is only a single vertex, then the join $\{v\} * K$ is called a *cone* with apex vertex v and we will usually denote it by v * K. Through this subsection we will use the fact that a cone is a near-cone, in spite of the fact that we have not introduced near-cones yet. Similarly, we will use the fact that $\Delta_{t-1} * K$ is a t-fold near-cone and postpone the definition until subsection 6.2.2. Let us recall that Δ_{t-1} stands for the (t-1)-simplex.

Let us now recall several families of simplicial complexes that belong to the class of sequentially Cohen-Macaulay simplicial complexes. One might argue that the graph class of trees is the simplest class of graph. One definition of a tree is that of a graph with every subgraph having a leaf. Faridi [29] generalized this to simplicial complexes, so called simplicial trees, and showed that these are sequentially Cohen-Macaulay. Further generalization were obtained to clutters with the free vertex property [86]. Yet another example of a sequentially Cohen-Macaulay simplicial complex is the simplicial complex given by the independent sets of a matroid. In fact, this last one satisfies an even stronger property, so called vertex decomposability [71].

Several classes of graphs whose independence complex is sequentially Cohen-Macaulay have been found. For example: chordal graphs [33]; graphs with no induced cycles of length other than 3 or 5 [90]; bipartite graphs satisfying the following recursive condition: there exist a degree 1 vertex v with its unique neighbor w such that $G \setminus N[v]$ and $G \setminus N[w]$ satisfy the same condition [86]. Here we have denotes by N[v] the closed neighborhood of v in G, i.e., N[v] = $\{v\} \cup \{w \in N : vw \in G\}$.

We notice that for two graphs G_1 and G_2 it follows from the definition that $\operatorname{Ind}(G_1 \sqcup G_2) = \operatorname{Ind}(G_1) * \operatorname{Ind}(G_2)$. In particular, $\operatorname{Ind}(\{w\} \sqcup G\}) = w * \operatorname{Ind}(G)$ and this is a near-cone. In other words, if G has an isolated vertex we have that $\operatorname{Ind}(G)$ is a near-cone. We notice that adding an isolated vertex to any of the above graph classes will create a graph whose independence complex is a sequentially Cohen-Macaulay near-cone.

Hurlbert and Kamat [43] showed that chordal graphs with an isolated vertex satisfy the k-EKR property. We can now conclude, using Theorem 63, that chordal graphs with an isolated vertex completely satisfy Conjecture 62.

Corollary 69. Let G be a chordal graph with an isolated vertex. Let d denote the minimal facet cardinality of Ind(G). Then, for k < d/2 we have that G satisfies strict k-EKR.

Proof. Since G is a chordal graph with an isolated vertex, by the previous discussion, we have that Ind(G) is a sequentially Cohen-Macaulay near-cone. The conclusion follows now by applying Theorem 63 to Ind(G).

Next we derive a Hilton-Milner type result and an upper bound on the size of cross-intersecting families for chordal graphs with several isolated vertices. Before presenting the corollary let us compute the independence complex of a graph with several isolated vertices. Let $W = \{w_1, \ldots, w_t\}$ denote t isolated vertices. Then, by the previous discussion, $\operatorname{Ind}(W \sqcup G) = \operatorname{Ind}(W) * \operatorname{Ind}(G) =$ $\Delta_{t-1} * \operatorname{Ind}(G)$ is a *t*-fold near-cone w.r.t. (w_1, \ldots, w_t) . In other words, if *G* has *t* isolated vertices $W = \{w_1, \ldots, w_t\}$ we have that $\operatorname{Ind}(G)$ is a *t*-fold near-cone w.r.t. (w_1, \ldots, w_t) . Moreover, if *G* is a chordal graph with *t* isolated vertices $\{w_1, \ldots, w_t\}$ then $\operatorname{Ind}(G)$ is a sequentially Cohen-Macaulay *t*-fold near-cone w.r.t. (w_1, \ldots, w_t) . The following results follow from combining this observation with Theorem 64 and Theorem 65, respectively.

Corollary 70. Let G be a chordal graph with vertex set N having k + 1 isolated vertices denoted by $W = \{w_1, \ldots, w_{k+1}\}$. Set K = Ind(G) and denote by d its minimal facet cardinality. Then, for $k \leq 2d$ and any pairwise-intersecting family $\mathcal{F} \subseteq K$ of (k-1)-faces such that $\cap \mathcal{F} = \emptyset$, we have that

 $|\mathcal{F}| \le f_{k-2}(\operatorname{lk}(w_1, K)) - f_{k-2}(\operatorname{lk}(w_1, K)[N \setminus W]) + 1.$

Proof. By the above discussion we have that K = Ind(G) is a sequentially Cohen-Macaulay (k+1)-fold near-cone w.r.t. (w_1, \ldots, w_{k+1}) . The conclusion now follows by applying Theorem 64 to K.

Corollary 71. Let G be a chordal graph with vertex set N having k isolated vertices denoted by $W = \{w_1, \ldots, w_k\}$. Set K = Ind(G) and denote by d its minimal facet cardinality. Then, for $k \leq 2d$ and any pair of non-empty crossintersecting families $\mathcal{F}, \mathcal{G} \subseteq K$ of (k-1)-faces, we have that

$$|\mathcal{F}| + |\mathcal{G}| \le f_{k-1}(K) - f_{k-1}(K[N \setminus W]) + 1.$$

Proof. By the above discussion we have that K = Ind(G) is a sequentially Cohen-Macaulay k-fold near-cone w.r.t. (w_1, \ldots, w_k) . The conclusion now follows by applying Theorem 65 to K.

Holroyd, Spencer and Talbot showed [41, Theorem 8] that a disjoint union of $n \ge 2k$ graphs given by paths, cliques and cycles including an isolated vertex satisfies k-EKR.

Theorem 72. [41, Theorem 8] Let $2k \leq n$. If $G = \bigsqcup_{i=1}^{n} G_i$ is a disjoint union of non-empty graphs with each G_i being a path, a clique or a cycle and at least one of the G_i being an isolated vertex, then G satisfies k-EKR.

Woodroofe generalized this result by showing that a disjoint union of $n \ge 2k$ graphs containing an isolated vertex satisfies k-EKR [91, Proposition 4.3].

Theorem 73. [91, Proposition 4.3] Let $2k \leq n$. If $G = \bigsqcup_{i=1}^{n} G_i$ is a disjoint union of non-empty graphs such that G has an isolated vertex, then G satisfies k-EKR.

We follow Woodroofe's strategy to show that the strict k-EKR also holds for this type of graphs. For it, first we need to recall how depth behaves w.r.t. disjoint union, see subsection 6.2.3 for the definition of depth of a simplicial complex.

Lemma 74. [91, Lemma 2.12] Let K_1 and K_2 be simplicial complexes. Then $\operatorname{depth}(K_1 * K_2) = \operatorname{depth}(K_1) + \operatorname{depth}(K_2) + 1$.

Combining this lemma with Theorem 66 we can deduce the following corollary.

Corollary 75. Let 2k < n. If $G = \bigsqcup_{i=1}^{n} G_i$ is a disjoint union of non-empty graphs and G has an isolated vertex, then G satisfies strict k-EKR.

Proof. On the one hand, since G has an isolated vertex, then by the previous discussion Ind(G) is a near-cone. On the other hand, by Lemma 74 we have that

$$depth(Ind(G)) = depth(Ind(G_1 \sqcup \cdots \sqcup G_n))$$
$$= depth(Ind(G_1) * \cdots * Ind(G_n)) \ge n - 1.$$

That is, $\operatorname{Ind}(G)$ is a near-cone with depth $(\operatorname{Ind}(G)) + 1 \ge n > 2k$. The conclusion now follows by applying Theorem 66 to $\operatorname{Ind}(G)$.

We can deduce a Hilton-Milner type upper bound as well as an upper bound on the size of cross-intersecting families for independence complexes of union of graphs. The following corollaries follow from combining Lemma 74 with Theorems 67 and 68, respectively.

Corollary 76. Let $2k \leq n$ and $G = \bigsqcup_{i=1}^{n} G_i$ be a disjoint union of non-empty graphs with vertex set N such that G contains k + 1 isolated vertices given by $W = \{w_1, \ldots, w_{k+1}\}$. Let $\mathcal{F} \subseteq \operatorname{Ind}(G)$ be a pairwise-intersecting family of (k-1)-faces such that $\bigcap \mathcal{F} = \emptyset$, then

$$|\mathcal{F}| \le f_{k-2}(\operatorname{lk}(w, \operatorname{Ind}(G))) - f_{k-2}(\operatorname{lk}(w, \operatorname{Ind}(G))[N \setminus W]) + 1,$$

where w is any of the elements in W.

Proof. On the one hand, since G has k + 1 isolated vertices given by $W = \{w_1, \ldots, w_{k+1}\}$, by the above discussion, we can conclude that Ind(G) is a (k+1)-fold near-cone w.r.t. (w_1, \ldots, w_{k+1}) . On the other hand, by Lemma 74 we have that

$$depth(Ind(G)) = depth(Ind(G_1 \sqcup \cdots \sqcup G_n))$$
$$= depth(Ind(G_1) * \cdots * Ind(G_n)) \ge n - 1.$$

That is, $\operatorname{Ind}(G)$ is a (k+1)-fold near-cone with depth $(\operatorname{Ind}(G)) + 1 \ge n \ge 2k$. The conclusion now follows by applying Theorem 67 to $\operatorname{Ind}(G)$.

Corollary 77. Let $2k \leq n$ and $G = \bigsqcup_{i=1}^{n} G_i$ be a disjoint union of non-empty graphs with vertex set N such that G contains k isolated vertices given by $W = \{w_1, \ldots, w_k\}$. Let $\mathcal{F}, \mathcal{G} \subseteq \operatorname{Ind}(G)$ a pair of non-empty cross-intersecting families of (k-1)-faces, then

$$|\mathcal{F}| + |\mathcal{G}| \le f_{k-1}(\operatorname{Ind}(G)) - f_{k-1}(\operatorname{Ind}(G)[N \setminus W]) + 1.$$

Proof. On the one hand, since G has k isolated vertices given by $W = \{w_1, \ldots, w_k\}$, by the above discussion, we can conclude that Ind(G) is a k-fold near-cone w.r.t. (w_1, \ldots, w_k) . On the other hand, by Lemma 74 we have that

$$depth(Ind(G)) = depth(Ind(G_1 \sqcup \cdots \sqcup G_n))$$
$$= depth(Ind(G_1) * \cdots * Ind(G_n)) \ge n - 1.$$

That is, $\operatorname{Ind}(G)$ is a k-fold near-cone with depth $(\operatorname{Ind}(G)) + 1 \ge n \ge 2k$. The conclusion now follows by applying Theorem 68 to $\operatorname{Ind}(G)$.

6.2 Preliminaries

6.2.1 Combinatorial shifting

Let 2^N denote the set of all subsets of N. Given a set system $\mathcal{F} \subseteq 2^N$, and $F \in \mathcal{F}$. Let $v, w \in N$, the *combinatorial shift* shift_{w,v} is defined by

shift_{w,v}(F,
$$\mathcal{F}$$
) =
$$\begin{cases} (F \setminus \{v\}) \cup \{w\} & \text{if } v \in F, w \notin F \text{ and} \\ & (F \setminus \{v\}) \cup \{w\} \notin \mathcal{F}, \\ F & \text{otherwise.} \end{cases}$$

 $\operatorname{shift}_{w,v}(\mathcal{F}) = {\operatorname{shift}_{w,v}(F, \mathcal{F}) \colon F \in \mathcal{F}}.$

Combinatorial shifting has been a successful technique to prove upper bounds on the size of cross-intersecting and pairwise-intersecting set systems. We will be using the following properties [35].

Theorem 78. Let \mathcal{F} , $\mathcal{G} \subseteq 2^N$ and $v, w \in N$.

- 1. $|\operatorname{shift}_{w,v}(\mathcal{F})| = |\mathcal{F}|.$
- 2. If $\mathcal{G} \subseteq \mathcal{F}$, then shift_{w,v}(\mathcal{G}) \subseteq shift_{w,v}(\mathcal{F}).
- 3. If \mathcal{F} and \mathcal{G} are cross-intersecting, then $\operatorname{shift}_{w,v}(\mathcal{F})$ and $\operatorname{shift}_{w,v}(\mathcal{G})$ are crossintersecting. In particular, if \mathcal{F} is pairwise-intersecting, then $\operatorname{shift}_{w,v}(\mathcal{F})$ is pairwise-intersecting.

We will be interested in applying repeatedly combinatorial shifting to a set system until we obtain a set system that is invariant under further aplication. First, let us recall that such a sequence of applications exists. For it, let us assume that the vertex set N is equipped with a total order, i.e., N = [n]. A set system $\mathcal{F} \subseteq 2^{[n]}$ is said to be *shifted* if for every $F \in \mathcal{F}$ and $i, j \in [n]$ such that $i < j, j \in F$ and $i \notin F$ we have that $(F \setminus \{j\}) \cup \{i\} \in \mathcal{F}$. The property of being shifted can be stated in terms of a partial order $<_p$ defined as follows. Let $S = \{s_1 < \cdots < s_k\}, T = \{t_1 < \cdots t_k\} \in {[n] \choose k}$, then $S \leq_p T$ if $s_i \leq t_i$ for every $i = 1, \ldots, k$. Then, \mathcal{F} is shifted if and only if for every $F \in \mathcal{F}$ and $G \leq_p F$ we have that $G \in \mathcal{F}$. Moreover, we have the following characterization in terms of combinatorial shifting [35].

Theorem 79. Let $\mathcal{F} \subseteq 2^{[n]}$. Then, \mathcal{F} is shifted if and only if $\operatorname{shift}_{i,j}(\mathcal{F}) = \mathcal{F}$ for every $i, j \in [n]$ with i < j.

By iterating the combinatorial shifting operation, $\operatorname{shift}_{i,j}$ with $1 \leq i < j \leq n$, we will eventually obtain a set system that is shifted [35, Proposition 2.2].

Theorem 80. [35, Proposition 2.2] Let $\mathcal{F} \subseteq 2^N$ and apply $\operatorname{shift}_{i,j}$ to \mathcal{F} for $1 \leq i < j \leq n$ exactly once in an order such that $\operatorname{shift}_{i,j}$ is applied before $\operatorname{shift}_{i',j'}$ if j' < j. The resulting set system is shifted.

In general we will not have available the entire range of parameters, i.e., $1 \le i < j \le n$, to perform combinatorial shifts in order to arrive to a shifted family. In the following we show how we shift a set system in a limited range of parameters

in order to obtain another set system invariant under further applications of combinatorial shift in the same range of parameters. Despite this restriction, the same argument as in [35, Proposition 2.2] works in this setting as well. For completeness we present the proof.

Lemma 81. Let $\mathcal{F} \subseteq 2^N$ and let $W = \{w_1, \ldots, w_t\} \subseteq N$. Apply $\operatorname{shift}_{w_i,v}$ to \mathcal{F} for $i \in [t]$ and $v \in N \setminus \{w_1, \ldots, w_i\}$ exactly once in an order such that $\operatorname{shift}_{w_i,v}$ is applied before $\operatorname{shift}_{w_i',w_{j'}}$ for every $i \in [t]$ and $v \in N \setminus W$ and $\operatorname{shift}_{w_i,w_j}$ is applied before $\operatorname{shift}_{w_{j'},w_{j'}}$ if j' < j.

Proof. We proceed by induction on $|N \setminus W|$. If N = W then the order is the same as the one in the statement of Theorem 80 and the conclusion follows by this theorem. Let $v \in N \setminus W$. First, we apply $\operatorname{shift}_{w_i,v}$ to \mathcal{F} once for every $i \in [t]$ and denote the resulting set system by \mathcal{H} . Then

$$\mathcal{H}(v) = \{F \setminus \{v\} \colon F \in \mathcal{H}, \ v \in F\} \subseteq \mathcal{H}(w_i)$$

for every $i \in [t]$ and this inclusion is preserved during later shifts. Set $\mathcal{H}(\bar{v}) = \{F \in \mathcal{H} : v \notin F\}$. We can apply the inductive hypothesis to $\mathcal{H}(v)$ and $\mathcal{H}(\bar{v})$ since $|(N \setminus \{v\}) \setminus W| < |N \setminus W|$ from which the conclusion follows. \Box

The same argument yields the following variation of the previous lemma. It will be instrumental to prove Theorems 64 and 67. We state it now for future reference.

Lemma 82. Let $\mathcal{F} \subseteq 2^N$, let $W = \{w_1, \ldots, w_t\} \subseteq N$ and fix some $w_s \in W$. Apply $\operatorname{shift}_{w_i,v}$ to \mathcal{F} for $i \in [t] \setminus \{1, s\}$ and $v \in N \setminus \{w_1, \ldots, w_i, w_s\}$ exactly once in an order such that $\operatorname{shift}_{w_i,v}$ is applied before $\operatorname{shift}_{w_{i'},w_{j'}}$ for every $i \in [t] \setminus \{1, s\}$ and $v \in N \setminus W$ and $\operatorname{shift}_{w_i,w_j}$ is applied before $\operatorname{shift}_{w_{i'},w_{i'}}$ if j' < j.

Let K be a simplicial complex on vertex set N and let u be a vertex of K. We will denote by del(w, K) the simplicial complex $K = [N \setminus \{w\}] = \{T \in K : w \notin T\}$. Let us now describe the behavior of combinatorial shifting when applied to del(u, K) and lk(u, K).

Lemma 83. Let K be a simplicial complex and u, v and w be different vertices of K. Then,

$$\operatorname{shift}_{w,v}(\operatorname{del}(u, K)) = \operatorname{del}(u, \operatorname{shift}_{w,v}(K)) \text{ and}$$
$$\operatorname{shift}_{w,v}(\operatorname{lk}(u, K)) = \operatorname{lk}(u, \operatorname{shift}_{w,v}(K)).$$

Proof. Let us start by showing that the first equality holds. Let $F \in del(u, K)$, we want to verify that $shift_{w,v}(F, del(u, K)) \in del(u, shift_{w,v}(K))$.

If $\operatorname{shift}_{w,v}(F, \operatorname{del}(u, K)) = F$, then by definition $v \notin F$, $w \in F$ or $(F \setminus \{v\}) \cup \{w\} \in \operatorname{del}(u, K)$. Since $\operatorname{del}(u, K) \subseteq K$, in each of the three cases we can conclude that $\operatorname{shift}_{w,v}(F, K) = F$. Because $u \notin F$ we have that $F \in \operatorname{del}(u, \operatorname{shift}_{w,v}(K))$.

If $\operatorname{shift}_{w,v}(F, \operatorname{del}(u, K)) = (F \setminus \{v\}) \cup \{w\}$, then $(F \setminus \{v\}) \cup \{w\} \notin \operatorname{del}(u, K)$ and, since it does not contain u, it is also not in K. Then, $\operatorname{shift}_{w,v}(F, K) = (F \setminus \{v\}) \cup \{w\} \in \operatorname{del}(u, \operatorname{shift}_{w,v}(K))$ as wanted.

Now, let $F \in K$ with $\operatorname{shift}_{w,v}(F,K) \in \operatorname{del}(u,\operatorname{shift}_{w,v}(K))$. We will verify that $\operatorname{shift}_{w,v}(F,K) \in \operatorname{shift}_{w,v}(\operatorname{del}(u,K))$ by showing that $\operatorname{shift}_{w,v}(F,K) = \operatorname{shift}_{w,v}(F,\operatorname{del}(u,K))$. If $\operatorname{shift}_{w,v}(F,K) = F$ then $v \notin F$, $w \in F$ or $(F \setminus \{v\}) \cup$

 $\{w\} \in K$. If $(F \setminus \{v\}) \cup \{w\} \in K$ then, since it does not contain the vertex u, it is also in del(u, K). Consequently in all three cases it follows that $\operatorname{shift}_{w,v}(F, \operatorname{del}(u, K)) = F$.

If $\operatorname{shift}_{w,v}(F,K) = (F \setminus \{v\}) \cup \{w\}$ then $(F \setminus \{v\}) \cup \{w\} \notin K$ and consequently not in $\operatorname{del}(u,K)$. Then, $\operatorname{shift}_{w,v}(F,\operatorname{del}(u,K)) = (F \setminus \{v\}) \cup \{w\} \in \operatorname{shift}_{w,v}(\operatorname{del}(u,K))$ as wanted.

Let us show next that the second equality holds. Let $F \in \operatorname{lk}(u, K)$, we want to verify that $\operatorname{shift}_{w,v}(F, \operatorname{lk}(u, K)) \in \operatorname{lk}(u, \operatorname{shift}_{w,v}(K))$. If $\operatorname{shift}_{w,v}(F, \operatorname{lk}(u, K)) = F$ then $v \notin F$, $w \in F$ or $(F \setminus \{v\}) \cup \{w\} \in \operatorname{lk}(u, K)$. If $(F \setminus \{v\}) \cup \{w\} \in \operatorname{lk}(u, K)$ then $(F \setminus \{v\}) \cup \{w, u\} \in K$. Consequently in all three cases we have that $\operatorname{shift}_{w,v}(F \cup \{u\}, K) = F \cup \{u\}$. That is, $F \in \operatorname{lk}(u, \operatorname{shift}_{w,v}(K))$.

If $\operatorname{shift}_{w,v}(F, \operatorname{lk}(u, K)) = (F \setminus \{v\}) \cup \{w\}$ then $(F \setminus \{v\}) \cup \{w\} \notin \operatorname{lk}(u, K)$ and consequently $(F \setminus \{v\}) \cup \{w, u\} \notin K$. Then, $\operatorname{shift}_{w,v}(F \cup \{u\}, K) = (F \setminus \{v\}) \cup \{w, u\} \in \operatorname{shift}_{w,v}(K)$ and consequently $(F \setminus \{v\}) \cup \{w\} \in \operatorname{lk}(u, \operatorname{shift}_{w,v}(K))$ as wanted.

Now, let $F \in \operatorname{lk}(u, \operatorname{shift}_{w,v}(K))$ then $F \cup \{u\} \in \operatorname{shift}_{w,v}(K)$. Let $F' \in \operatorname{lk}(u, K)$ such that $\operatorname{shift}_{w,v}(F' \cup \{u\}, K) = F \cup \{u\}$. If F = F' then $v \notin F'$, $w \in F'$ or $(F' \setminus \{v\}) \cup \{w, u\} \in K$. If $(F' \setminus \{v\}) \cup \{w, u\} \in K$ then $(F' \setminus \{v\}) \cup \{w\} \in$ $\operatorname{lk}(u, K)$. Then, in all three cases we have that $\operatorname{shift}_{w,v}(F', \operatorname{lk}(u, K)) = F' = F \in$ $\operatorname{shift}_{w,v}(\operatorname{lk}(u, K))$ as wanted.

If $F = (F' \setminus \{v\}) \cup \{w\}$ then $(F' \setminus \{v\}) \cup \{w, u\} \notin K$ and consequently $(F' \setminus \{v\}) \cup \{w\} \notin \operatorname{lk}(u, K)$. Therefore, $\operatorname{shift}_{w,v}(F', \operatorname{lk}(u, K)) = (F' \setminus \{v\}) \cup \{w\} = F \in \operatorname{shift}_{w,v}(\operatorname{lk}(u, K))$ as wanted. \Box

6.2.2 *t*-fold near-cone

A simplicial complex K is a *near-cone* with apex vertex w if for every face $F \in K$ we have that $(F \setminus \{v\}) \cup \{w\} \in K$ for every $v \in F$. We notice that K is a near-cone with apex vertex w if and only if $\operatorname{shift}_{w,v}(K) = K$ for every $v \in N$. That is, when we apply combinatorial $\operatorname{shift}_{w,v}$ to the set system K, this one does not change.

Let K be a simplicial complex with vertex set N and $W = (w_1, \ldots, w_t)$ a sequence of different vertices of K, we say that K is a t-fold near-cone w.r.t W if K is a near-cone with apex vertex w_1 and, $del(w_1, K)$ and $lk(w_1, K)$ are (t - 1)fold near-cones w.r.t. (w_2, \ldots, w_t) . The following lemma provides an alternative characterization of t-fold near-cones.

Lemma 84. Let K be a simplicial complex and $W = (w_1, \ldots, w_t) \subseteq N$ a sequence of different vertices. Then, K is a t-fold near-cone w.r.t. W if and only if $\text{shift}_{w_i,v}(K) = K$ for every $w_i \in W$ and $v \in N \setminus \{w_1, \ldots, w_i\}$.

Proof. On the one hand, let $w_i \in W$ and $v \in N \setminus \{w_1, \ldots, w_i\}$. We want to show that $\operatorname{shift}_{w_i,v}(K) = K$. For it, let $F \in K$ and assume that $v \in F$, otherwise the combinatorial shift does not affect F. We want to verify that $(F \setminus \{v\}) \cup \{w_i\} \in K$. We will show that $F' = F \setminus \{w_j \in W \cap F : j < i\}$ is in a subcomplex K'of $\operatorname{lk}(\{w_j \in W \cap F : j < i\}, K)$ that is a near-cone with apex vertex w_i . We build K' as follows. First, we initialize $K_0 = K$, then for $j = 1, \ldots, i - 1$ we set $K_j = \operatorname{del}(w_j, K_{j-1})$ if $w_j \notin F \cap W$ and $K_j = \operatorname{lk}(w_j, K_{j-1})$ otherwise. By hypothesis it follows that for every $j = 0, \ldots, i - 1$ the simplicial complex K_j is a



Figure 6.1: Example of a 3-near-cone that it is not a 3-fold near-cone.

near-cone with apex vertex w_{j+1} and $F \setminus \{w_t \in W \cap F : t < j\} \in K_j$. In the end we have that $F' \in K_{i-1} = K'$. Then, since K' is a near-cone with apex vertex w_i we have that

$$(F' \setminus \{v\}) \cup \{w_i\} \in K' \subseteq \operatorname{lk}(\{w_t \in W \cap F \colon t < i\}, K).$$

Then, $(F \setminus \{v\}) \cup \{w_i\} \in K$.

On the other hand, since $\operatorname{shift}_{w_1,v}(K) = K$ for every $v \in N \setminus \{w_1\}$ we have that K is a near-cone with apex vertex w_1 . Now, we want to verify that $\operatorname{del}(w_1, K)$ and $\operatorname{lk}(w_1, K)$ are near-cones with apex vertex w_2 . To see this we will show that $\operatorname{shift}_{w_2,v}(\operatorname{del}(w_1, K)) = \operatorname{del}(w_1, K)$ and $\operatorname{shift}_{w_2,v}(\operatorname{lk}(w_1, K)) = \operatorname{lk}(w_1, K)$ for every $v \in N \setminus \{w_1, w_2\}$. By Lemma 83 we have that

$$shift_{w_2,v}(del(w_1, K)) = del(w_1, shift_{w_2,v}(K)) = del(w_1, K),$$

and

$$shift_{w_2,v}(lk(w_1, K)) = lk(w_1, shift_{w_2,v}(K)) = lk(w_1, K),$$

where in both cases the last equality follows from the assumption $\operatorname{shift}_{w_2,v}(K) = K$ for every $v \in N \setminus \{w_1, w_2\}$. Then, $\operatorname{del}(w_1, K)$ and $\operatorname{lk}(w_1, K)$ are near-cones with apex vertex w_2 . By repeating the argument with $\operatorname{del}(w_1, K)$ and $\operatorname{lk}(w_1, K)$ in place of K we can conclude that K is a t-fold near-cone w.r.t. W.

Example 85. Let G be a graph with t isolated vertices $\{w_1, \ldots, w_t\}$, then Ind(G) is a t-fold near-cone w.r.t. $W = (w_1, \ldots, w_t)$. In fact, in this case the order of W is not important.

The following notion is closely related to t-fold near-cones. A simplicial complex K is a t-near-cone w.r.t. $W = (w_1, \ldots, w_t)$ if there exists a sequence of simplicial complexes

$$K = K(0) \supseteq K(1) \supseteq \cdots \supseteq K(t)$$

such that for every $1 \leq j \leq t$, $K(j) = \{T \in K(j-1) : w_j \notin T\}$ and K(j-1) is a near-cone with apex vertex w_j .

Lemma 86. If K is a t-fold near-cone w.r.t. $W = (w_1, \ldots, w_t)$, then it is a t-near-cone w.r.t. W.

Proof. Let K(0) = K and for j = 1, ..., t set $K(j) = del(w_j, K(j-1))$. By induction K(j-1) is a near-cone with apex vertex w_j .

Remark 87. In general a t-near-cone is not a t-fold near-cone. Consider the simplicial complex K with facets $\{1, 2, 4\}, \{1, 3\}, \{2, 3\}$ depicted in Figure 6.1. It is a 3-near-cone w.r.t. (1, 2, 3) since K is a near-cone with apex 1. The simplicial complex K(1) has facets $\{2, 3\}$ and $\{2, 4\}$ and it is a near-cone with apex vertex 2, see Figure 6.1. The complex K(2) has facets $\{3\}$ and $\{4\}$ and it is a near-cone with apex vertex 3. Finally, K(3) has a unique facet $\{4\}$. But $\{1, 2, 3\} \in \text{shift}_{3,4}(K)$ while $\{1, 2, 3\} \notin K$. By Lemma 84 we can conclude that K is not a 3-fold near-cone w.r.t. (1, 2, 3) since shift $_{3,4}(K) \neq K$.

6.2.3 Depth and the sequentially Cohen-Macaulay property

Let \mathbb{F} denote a field. The *i*-th Betti number $\beta_i(K, \mathbb{F}) = \dim_{\mathbb{F}} H_i(K, \mathbb{F})$ is the dimension of the *i*-th homology group. Since we are considering coefficients over a field, this last one is in fact a vector space. One can read-off the Betti number from the near-cone combinatorially as follows.

Theorem 88. [50, Lemma 3.1] Let K be a near-cone with apex vertex w, then

$$\beta_{k-1}(K) = |\{S \in K \colon |S| = k, \ S \cup \{w\} \notin K\}|.$$

A simplicial complex K is called Cohen-Macaulay over \mathbb{F} if for every face $F \in K$ we have that $\tilde{H}_i(\operatorname{lk}(F, K), \mathbb{F}) = 0$ for $i < \operatorname{dim}(\operatorname{lk}(F, K))$. That is, the reduced homology of every link vanishes on every dimension except possibly the top one. The pure k-skeleton of K is the simplicial complex generated by the k-faces of K. A simplicial complex is said to be sequentially Cohen-Macaulay over \mathbb{F} if for every k, the pure k-skeleton of K is Cohen-Macaulay over \mathbb{F} . From now on we will assume that the field has characteristic 0 and drop it from the notation.

We need to be able to control the behavior of the minimal facet cardinality of a simplicial complex because it plays a key role in Conjecture 62. For this purpose we will use the following definition of *depth of a simplicial complex* K

$$depth(K) = \max\{d \colon K^{(d)} \text{ is Cohen-Macaulay}\}.$$

We can restate this definition in a more convenient way as follows. The simplicial complex $K^{(d)}$ is Cohen-Macaulay if and only if $\tilde{H}_i(\operatorname{lk}(F, K^{(d)}), \mathbb{F}) = 0$ for every $F \in K^{(d)}$ and $i < \dim \operatorname{lk}(F, K^{(d)})$. Then, for $i < \dim \operatorname{lk}(F, K^{(d)})$ we have that

$$\begin{split} \tilde{H}_i(\mathrm{lk}(F, K^{(d)}), \mathbb{F}) &= \tilde{H}_i(\mathrm{lk}(F, K^{(d)})^{(i+1)}, \mathbb{F}) = \tilde{H}_i(\mathrm{lk}(F, K)^{(i+1)}, \mathbb{F}) \\ &= \tilde{H}_i(\mathrm{lk}(F, K), \mathbb{F}), \end{split}$$

where we have used that the k-th homology group only depends on the (k+1)-th skeleton, i.e., $\tilde{H}_k(K^{(k+1)}, \mathbb{F}) = \tilde{H}_k(K, \mathbb{F})$, and that $\operatorname{lk}(F, K^{(d)})^{(i)} = \operatorname{lk}(F, K)^{(i)}$ for $i \leq \dim \operatorname{lk}(F, K^{(d)})$. Since dim $\operatorname{lk}(F, K^{(d)}) \leq d - |F|$ we can conclude the following characterization

$$depth(K) = \max\{d \colon H_i(lk(F, K), \mathbb{F}) = 0 \text{ for every } F \in K \text{ and } i < d - |F|\}.$$

Next, we observe that depth(K) + 1 is at most the minimal facet cardinality. To see this let F be a facet of K, then $lk(F, K) = \{\emptyset\}$ and consequently $\tilde{H}_{-1}(lk(F, K)) = \tilde{H}_{-1}(\{\emptyset\}) = \mathbb{F}$. That is, depth $(K) + 1 \leq |F|$ for any facet of K, in particular depth(K) + 1 is a lower bound on the minimal facet cardinality. The depth of a simplicial complex is one less than the depth of its Stanley-Reisner ring [75].

6.2.4 Exterior algebraic shifting

Algebraic shifting was introduced by Kalai (see e.g. [49] and the survey [50]) and has been studied extensively in algebraic combinatorics. Here we review the definition of exterior algebraic shifting as well as some of its properties. Let us denote by $(e_i)_{i \in [n]}$ the standard basis of $V = \mathbb{R}^{[n]}$ and consider $(f_i)_{i \in [n]}$ a generic change of basis given by $f_i = \sum_{j \in [n]} a_{ij}e_j$ with the coefficients $(a_{ij})_{i,j \in [n]}$ being algebraically independent. Consider the exterior face ring

$$\bigwedge K = \bigwedge V / (e_T \colon T \notin K),$$

and let q denote the natural quotient map $q: \wedge V \to \wedge K$. The exterior algebraic shift of K, denoted by $\Delta(K)$, is defined as

$$\Delta(K) = \{T \subseteq [n] \colon q(f_T) \notin \operatorname{span}_{\mathbb{R}} \{q(f_S) \colon S <_{lex} T, |S| = |T|\}\},\$$

where $<_{lex}$ denotes the lexicographical order defined as $S <_{lex} T$ if |S| < |T| or |S| = |T| and $\min((S \cup T) \setminus (S \cap T)) \in S$. Here we merely state the properties we will be using.

Theorem 89. Let K be a simplicial complex.

- 1. [50, Theorem 2.1.1] Exterior algebraic shifting preserves the f-vector, i.e., $f_k(K) = f_k(\Delta(K)).$
- 2. [50, Theorem 2.1.2] The simplicial complex $\Delta(K)$ is shifted.
- 3. [50, Theorem 2.1.4] If K is shifted, then $\Delta(K) = K$.
- 4. [50, Theorem 2.2.7] If $K \subseteq L$, then $\Delta(K) \subseteq \Delta(L)$.
- 5. [50, Theorem 3.2] Exterior algebraic shifting preserves the Betti numbers, i.e., $\beta_k(K) = \beta_k(\Delta(K))$ for every $k \ge 0$.
- 6. [50, Theorem 4.1] If K is Cohen-Macaulay then $\Delta(K)$ is Cohen-Macaulay.
- 7. [50, Theorem 6.2] If $\mathcal{F} \subseteq {\binom{[n]}{a}}$ and $\mathcal{G} \subseteq {\binom{[n]}{b}}$ are cross-intersecting, then $\Delta(\mathcal{F})$ and $\Delta(\mathcal{G})$ are cross-intersecting. In particular, if \mathcal{F} is pairwise-intersecting, then $\Delta(\mathcal{F})$ is pairwise-intersecting.

We would like to point out that in contrast to combinatorial shifting, exterior algebraic shifting produces a shifted complex in one step rather than after a sequence of applications.



Figure 6.2: Exterior algebraic shifting of a tree and a cycle.

Example 90. Let G be a tree on n vertices, then $\Delta(G)$ is a connected, acyclic and shifted graph. We claim that the only edges are $\{1, j\} \in \Delta(G)$ for j = 2, ..., n. On the one hand, these edges must be there: since $\Delta(G)$ is connected for every j there is an edge $\{i, j\} \in \Delta(G)$, then $\{1, j\} \in \Delta(G)$ since the complex is shifted. Since algebraic shifting preserves the number of edges, these are the only ones because |E(G)| = n - 1. See the left side of Figure 6.2.

Now, let G be a cycle on n vertices, then $\Delta(G)$ contains a star on the vertex 1 and the edge $\{2,3\}$. Since it is connected, it contains the star. On the other hand, since algebraic shifting preserves the first Betti number then $\{2,3\}$ is in the exterior algebraic shift. By counting the number of edges these must be all of them. See the right side of Figure 6.2.

The following corollary generalizes the above observation regarding the algebraic shift of a cycle. It follows from Theorems 88 and 89.

Corollary 91. Let K be a simplicial complex with $\beta_{k-1}(K) > 0$. Then,

$$\binom{[k+1]}{k} \subseteq \Delta(K).$$

Proof. Let $S \in \Delta(K)$ be a k-face with $1 \notin S$. Such a face exists since

$$|\{T \in \Delta(K) \colon |T| = k, \{1\} \cup S \notin \Delta(K)\}| = \beta_{r-1}(\Delta(K)) = \beta_{r-1}(K) > 0.$$

Because $\{2, \ldots, k+1\} \leq_p S$ and $\Delta(K)$ is shifted we have that $\{2, \ldots, k+1\} \in \Delta(K)$. Using that $[k+1] \setminus \{i\} <_p \{2, \ldots, k+1\}$ for all $i = 2, \ldots, k+1$ and that $\Delta(K)$ is shifted we can conclude that $\binom{[k+1]}{k} \subseteq \Delta(K)$.

Before proceeding let us mention the importance of the preceding corollary. We will use it to guarantee that the algebraic shifting of a pairwise-intersecting family \mathcal{F} satisfies the empty intersection condition, i.e., $\bigcap \Delta(\mathcal{F}) = \emptyset$. Concretely, if $\beta_{k-1}(\langle \mathcal{F} \rangle) > 0$, then $\binom{[k+1]}{k} \subseteq \Delta(\mathcal{F})$. Here we have used the notation $\langle \mathcal{F} \rangle$ to denote the simplicial complex generated by \mathcal{F} . In particular, $\bigcap \Delta(\mathcal{F}) \subseteq \bigcap \binom{[k+1]}{k} = \emptyset$. We will use this in several steps when proving Theorem 63 and Theorem 64.

Next we recall how the structure of a *t*-near-cone behaves w.r.t. algebraic shifting. For it, first let us set a bit of notation. Let $j \in [n]$. For a subset $T \subseteq [n]$ we denote by T+j the set $\{t+j: t \in T\}$. For a simplicial complex K with vertex set [n] we denote by K+j the simplicial complex $\{T+j: T \in K\}$.

Theorem 92. [66, Corollary 5.6] Let K be a t-near-cone w.r.t. $W = (w_1, \ldots, w_t)$. Then

$$\Delta(K) = B \cup \bigsqcup_{j=1}^{t} \{\{j\} \sqcup T \colon T \in (\Delta(\operatorname{lk}(w_j, K(j-1))) + j)\}$$

where \sqcup denotes disjoint union and $B = \{F \in \Delta(K) : F \cap [t] = \emptyset\}.$

In Example 90 we saw that the degree of a vertex can increase after performing algebraic shifting. The next corollary tells us that if we have a near-cone, the number of faces containing the apex vertex is preserved under algebraic shifting. This is of crucial importance when proving Theorem 63 since we want to transfer the upper bound from the shifted simplicial complex back to the original one.

Corollary 93. Let K be a near-cone with apex vertex w, then

$$f_k(\operatorname{lk}(w, K)) = f_k(\Delta(\operatorname{lk}(w, K))) = f_k(\operatorname{lk}(1, \Delta(K))).$$

Proof. Since a near-cone with apex vertex w is a 1-near-cone w.r.t. W = (w), by applying Theorem 92 we obtain

$$\Delta(K) = B \sqcup \{\{1\} \sqcup T \colon T \in (\Delta(\operatorname{lk}(w, K)) + 1)\}$$

with $B = \{F \in \Delta(K) : 1 \notin F\}$. Then $lk(1, \Delta(K)) = \Delta(lk(w, K)) + 1$ and the desired equality of f-vectors follows.

For Theorems 64 and 65 we need to relate the number of faces of K disjoint from W with those of $\Delta(K)$ disjoint from [t]. This is the purpose of the following proposition.

Proposition 94. Let K be a t-near-cone w.r.t. $W = (w_1, \ldots, w_t)$. Then

$$f_k(K[N \setminus W]) = f_k(\Delta(K)[[n] \setminus [t]]).$$

Proof. It is enough to verify the equality for the complement; that is, to verify that

$$f_k(K) - f_k(K[N \setminus W]) = |\{T \in K \colon |T| = k+1, \ T \cap W \neq \emptyset\}$$

coincides with

$$f_k(\Delta(K)) - f_k(\Delta(K)[[n] \setminus [t]]) = |\{T \in \Delta(K) \colon |T| = k+1, \ T \cap [t] \neq \emptyset\}|$$

because $f_k(K) = f_k(\Delta(K))$. Let $T \in K$ be such that $T \cap W = \{w_{i_1}, \dots, w_{i_k}\}$ with $i_1 < \dots < i_k$, then $T \in K(i_1 - 1) \setminus (\bigcup_{j > i_1} K(j - 1))$. To count such faces only once, we order them by the lowest member of W they contain, that is

$$\begin{split} |\{T \in K \colon |T| = k+1, \ T \cap W \neq \emptyset\}| &= \sum_{j=1}^{t} |\{T \in K \colon |T| = k+1, \\ \min\{j \colon w_j \in T \cap W\} = j\}| \\ &= \sum_{j=1}^{t} f_{k-1}(\operatorname{lk}(w_j, K(j-1))). \end{split}$$

Similarly, and using Theorem 92, we have that

$$|\{T \in \Delta(K) \colon |T| = k+1, \ T \cap [t] \neq \emptyset\}| = \sum_{j=1}^{t} |\{T \in \Delta(K) \colon |T| = k+1, \\ \min T \cap [t] = j\}|$$
$$= \sum_{j=1}^{t} f_{k-1}(\Delta(\operatorname{lk}(w_j, K(j-1))))$$
$$= \sum_{j=1}^{t} f_{k-1}(\operatorname{lk}(w_j, K(j-1))),$$

where the last equality follows from Theorem 89.1.

Now we will exhibit the interplay between depth and the minimal facet cardinality after performing algebraic shifting.

Theorem 95 ([25]). The minimum facet dimension of $\Delta(K)$ is at least d if and only if $K^{(d)}$ is Cohen-Macaulay.

On the one hand, this theorem implies that depth(K) + 1 coincides with the minimal facet cardinality of $\Delta(K)$. On the other hand, we have seen at the end of subsection 6.2.3 that depth(K) + 1 is at most the minimal facet cardinality of K. Summarizing, we have that

minimal facet size of $\Delta(K) = \operatorname{depth}(K) + 1 \leq \min$ facet size of K.

If K is sequentially Cohen-Macaulay with minimal facet cardinality d, then its pure (d-1)-skeleton K' is Cohen-Macaulay. By applying algebraic shifting to such skeleton we obtain a Cohen-Macaulay (d-1)-dimensional subcomplex $\Delta(K') \subseteq \Delta(K)$. We notice that for every $k \leq (d-1)$ we have that

$$f_k(\Delta(K')) = f_k(K') = f_k(K) = f_k(\Delta(K))$$

and consequently the shifted subcomplex $\Delta(K')$ contains every face of $\Delta(K)$ with dimension at most (d-1). We can conclude that the minimal facet cardinality of $\Delta(K)$ is $d = \operatorname{depth}(K) + 1$ since $\Delta(K')$ is pure of dimension (d-1).

6.3 Hilton-Milner type upper bound

In this section we prove Theorems 64 and 67. For it we adapt the injective proof from [36, 44] to the simplicial complex setting. The first step for this is to show that we can always assume that the pairwise-intersecting family \mathcal{F} with the empty intersection property, i.e., $\bigcap \mathcal{F} = \emptyset$, is shifted. This is the content of the following lemma.

Lemma 96. Let K be a (k + 1)-fold near-cone w.r.t. $W = (w_1, \ldots, w_{k+1})$ and $2 \le k \le \frac{\operatorname{depth}(K)+1}{2}$. Let $\mathcal{F} \subseteq K$ be a pairwise-intersecting family of (k - 1)-faces such that $\bigcap \mathcal{F} = \emptyset$ and $|\mathcal{F}|$ is maximal. Then, there exists a pairwise-intersecting shifted family of (k - 1)-faces $\mathcal{F}' \subseteq \Delta(K)$ with $\bigcap \mathcal{F}' = \emptyset$ and $|\mathcal{F}'| = |\mathcal{F}|$.

Proof. First, because K is a (k + 1)-fold near-cone w.r.t. W, by Lemma 84, we have that $\operatorname{shift}_{w_i,v}(K) = K$ for every $w_i \in W$ and $v \in N \setminus \{w_1, \ldots, w_i\}$. In this proof we apply combinatorial shift to \mathcal{F} as well as to K in order to preserve the inclusion. Since we only apply combinatorial shift that does not modify K we avoid mentioning at each step that the result of shifting \mathcal{F} is contained in K

Now, we iteratively apply $\operatorname{shift}_{w_1,v}$ to \mathcal{F} for $v \in N \setminus \{w_1\}$ according to Lemma 81¹ while the resulting family satisfies the empty intersection condition in the statement, otherwise we stop. Let us denote the resulting family by \mathcal{H} . We proceed by case analysis.

Case 1: shift_{w1,v}(\mathcal{H}) = \mathcal{H} for all $v \in N$. Since \mathcal{H} satisfies the empty intersection condition, there exists $F \in \mathcal{H}$ such that $w_1 \notin F$ and for every $v \in F$ we have that $(F \setminus \{v\}) \cup \{w_1\} \in \mathcal{H}$. That is, $\binom{F \cup \{w_1\}}{k} \subseteq \mathcal{H}$ and consequently $\beta_{k-1}(\langle \mathcal{H} \rangle) > 0$, where by $\langle \mathcal{H} \rangle$ we denote the simplicial complex with facets given by \mathcal{H} . Then by Corollary 91 we have that $\binom{[k+1]}{k} \subseteq \Delta(\mathcal{H})$ and we can conclude that $\mathcal{F}' = \Delta(\mathcal{H}) \subseteq \Delta(K)$ is the desired family.

Case 2: there exists $w_s \in W$ such that $\bigcap \operatorname{shift}_{w_1,w_s}(\mathcal{H}) \neq \emptyset$. Then $F \cap \{w_1, w_s\} \neq \emptyset$ for all $F \in \mathcal{H}$. By maximality of $|\mathcal{H}|$ we have that

$$\mathcal{T} = \{ T \cup \{ w_1, w_s \} \colon T \in \text{lk}(\{ w_1, w_s \}, K), |T| = k - 2 \} \subseteq \mathcal{H}.$$

Since $\cap \mathcal{H} = \emptyset$ there exist $F_1, F_s \in \mathcal{H}$ such that $w_1 \in F_1, w_s \notin F_1$ and $w_s \in F_s, w_s \notin F_s$

Next, we iteratively apply shift_{w_j,v}, with $w_j \in W \setminus \{w_1, w_s\}$ and $v \in N \setminus \{w_1, \ldots, w_j, w_s\}$ to \mathcal{H} until obtaining a stable family as shown in Lemma 82. Denote the resulting family by \mathcal{G} . Because $F_1, F_s \in \mathcal{H}$, there are $G_1, G_s \in \mathcal{G}$ such that $w_1 \in G_1, w_s \notin G_1$ and $w_s \in G_s, w_1 \notin G_s$. Since shift_{w,v}(\mathcal{G}) = \mathcal{G} for w and v in the above range, we have that $W \setminus w_1, W \setminus w_s \in \mathcal{G}$. Moreover, we also have that shift_{w,v}(\mathcal{T}) = \mathcal{T} , for the same range of parameters as above since shift_{w,v}(K) = K. Consequently $\binom{W}{k} \subseteq \mathcal{G}$. The conclusion follows as in the previous case.

Case 3: there exist $v \in N \setminus W$ and $w_s \in W$ such that $\bigcap \operatorname{shift}_{w_1,v}(\mathcal{H}) \neq \emptyset$ and $\bigcap \operatorname{shift}_{w_s,v}(\mathcal{H}) = \emptyset$. Let $\mathcal{G} = \operatorname{shift}_{w_s,v}(\mathcal{H})$. We claim that every $F \in \mathcal{G}$ intersects $\{w_1, w_s\}$. If $F \in \mathcal{H}$, then either $w_s \in F$, $v \notin F$, or $(F \setminus \{v\}) \cup \{w_s\} \in \mathcal{H}$. In the last two cases it follows that $w_1 \in F$ since $\bigcap \operatorname{shift}_{w_1,v}(\mathcal{H}) \neq \emptyset$. If $F \notin \mathcal{H}$ then $w_s \in F$. We proceed as in the previous case with \mathcal{G} in place of \mathcal{H} .

Case 4: there exists $v \in N \setminus W$ such that $\cap \operatorname{shift}_{w,v}(\mathcal{H}) \neq \emptyset$ for all $w \in W$. On the one hand, since $\cap \mathcal{H} = \emptyset$ there exists $F \in \mathcal{H}$ such that $v \notin F$. Because $F \cap \{w, v\} \neq \emptyset$ for all $w \in W$ we can conclude that $w \in F$ for all $w \in W$. This is a contradiction with the size of F.

The next step is to build an injection from the shifted pairwise-intersecting family to a proper subcomplex of the link of the apex vertex. For it we first recall the injection used in the unrestricted case, that is if the underlying simplicial complex is the simplex. We will need the following technical lemma.

Lemma 97. [36, 44] Let $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a pairwise-intersecting shifted family. For every $F \in \mathcal{F}$ there exists $l \geq 1$ such that $|F \cap [2l-1]| \geq l$. Moreover, the maximum such l = l(F) satisfies $|F \cap [2l(F)]| = l(F)$.

¹We apply the lemma with $W = \{w_1\}$. Here W denotes the vertex set in the statement of Lemma 81.

Let $2k \leq n$ and $\mathcal{F} \subseteq {\binom{[n]}{k}}$ be a shifted pairwise-intersecting family with $\bigcap \mathcal{F} = \emptyset$. The following function was previously defined by Frankl [36], see also [44].

$$\alpha \colon \mathcal{F} \to \left\{ F \in \binom{[n]}{k} \colon 1 \in F, \ F \cap [2, k+1] \neq \emptyset \right\} \cup \{[2, k+1]\}$$
$$F \mapsto \begin{cases} F & \text{if } 1 \in F \text{ or } [2, k+1] \subseteq F, \\ (F \cup [2l(F)]) \setminus (F \cap [2l(F)]) & \text{otherwise.} \end{cases}$$

In the same article it was shown that α is well defined and the following properties hold.

Lemma 98. For $F \in \mathcal{F}$ such that $\alpha(F) \neq F$ we have that:

- $1 \in \alpha(F)$.
- $\alpha(F) \notin \mathcal{F}$.
- $\alpha(F) \cap [2, k+1] \neq \emptyset$.
- α is injective.
- $|\alpha(F)| = |F|$.

Next, we show that α still works in the simplicial complex setting. Before stating the lemma, let us recall the notion of the star of a vertex. Let K be a simplicial complex and v be a vertex in K, then the *star of* v *in* K is the set $st(v, K) = \{F \in K : v \in F\}.$

Lemma 99. Let $k \leq d/2$ and K be a shifted simplicial complex with vertex set [n] and minimal facet cardinality d. Let $\mathcal{F} \subseteq K$ be shifted pairwise-intersecting family of (k-1)-faces with $\bigcap \mathcal{F} = \emptyset$. Then, for every $F \in \mathcal{F}$ we have that

$$\alpha(F) \in (\operatorname{st}(1,K) \setminus \operatorname{st}(1,K[[n] \setminus [2,k+1]])) \cup \{[2,k+1]\}$$

Proof. It is enough to verify that $\alpha(F) \in K$. First, we notice that $d/2 \ge k \ge |F \cap [2l(F)]| = l(F)$. In particular,

$$|F \cup [2l(F)]| = |F| + |[2l(F)]| - |F \cap [2l(F)]| = k + l(F) \le 2k \le d.$$

Since K has minimal facet cardinality d there exist a face T with $k + l(F) \leq d$ vertices containing F. Let $T \setminus F = \{t_1 < \cdots < t_{l(F)}\}$ and $[2l(F)] \setminus F = \{s_1 < \cdots < s_{l(F)}\}$. Then, $s_i \leq t_i$ for every $i \in [l(F)]$. Since K is shifted, it is not affected when applying shift_{si,ti} to it. On the other hand, if we apply iteratively shift_{si,ti} to T exactly once for each $i \in [l(F)]$, we will obtain $F \cup [2l(F)]$. From this we can conclude that the later one is in K. Since $F \subseteq F \cup [2l(F)]$, $(F \cup [2l(F)]) \setminus (F \cap [2l(F)]) \subseteq F \cup [2l(F)]$ and $[2, k + 1] \in K$ we conclude that $\alpha(F) \in K$.

Combining Lemma 98 and Lemma 99 we obtain the following corollary.

Corollary 100. Let $2 \leq k \leq d/2$ and K be a shifted simplicial complex with vertex set [n] and minimal facet cardinality d. Let $\mathcal{F} \subseteq K$ be a shifted pairwise-intersecting family of (k-1)-faces with $\cap \mathcal{F} = \emptyset$. Then,

$$\alpha \colon \mathcal{F} \to (\mathrm{st}(1, K) \setminus \mathrm{st}(1, K[[n] \setminus [2, k+1]])) \cup \{[2, k+1]\}$$

is a well defined injection. In particular,

$$|\mathcal{F}| \leq f_{k-2}(\operatorname{lk}(1,K)) - f_{k-2}(\operatorname{lk}(1,K[[n] \setminus [k+1]])) + 1.$$

Proof of Theorem 67. On the one hand, since $lk(w_1, K)$ is an k-fold near-cone w.r.t. (w_2, \ldots, w_{k+1}) we have by Corollary 93 and Proposition 94 that

$$f_{k-2}(\mathrm{lk}(1,\Delta(K))[[n] \setminus [k+1]]) = f_{k-2}(\Delta(\mathrm{lk}(w_1,K))[[n] \setminus [k+1]]) = f_{k-2}(\mathrm{lk}(w_1,K)[N \setminus W]).$$

On the other hand, since the minimal facet cardinality of $\Delta(K)$ is depth(K) + 1, by Lemma 96 there exists $\mathcal{F}' \subseteq \Delta(K)$ shifted pairwise-intersecting family such that $|\mathcal{F}'| = |\mathcal{F}|$ and $\cap \mathcal{F}' = \emptyset$. Then

$$\begin{aligned} |\mathcal{F}| &= |\mathcal{F}'| \le f_{k-2}(\mathrm{lk}(1,\Delta(K))) - f_{k-2}(\mathrm{lk}(1,\Delta(K))[[n] \setminus [k+1]]) + 1 \\ &= f_{k-2}(\mathrm{lk}(w_1,K)) - f_{k-2}(\mathrm{lk}(w_1,K)[N \setminus W]) + 1, \end{aligned}$$

where we combined Corollary 93 and Corollary 100.

As a consequence we obtain a proof for Theorem 64.

Proof of Theorem 64. Since K is a sequentially Cohen-Macaulay simplicial complex with minimal facet cardinality d, it follows from the discussion at the end of subsection 6.2.3 that depth(K) + 1 = d. Since $k \leq d/2 = \frac{\text{depth}(K)+1}{2}$, we can conclude the proof by applying Theorem 67.

6.4 Strict Erdős-Ko-Rado

In this section we prove Theorems 63 and 66. We do this by reducing the general case to the case where the underlying simplicial complex is shifted. The following corollary for shifted simplicial complexes follows from Corollary 100.

Corollary 101. Let $2 \leq k < d/2$ and K be a shifted simplicial complex with vertex set [n] and minimal facet cardinality d. Let $\mathcal{F} \subseteq K$ be a shifted pairwise-intersecting family of (k-1)-faces with $\cap \mathcal{F} = \emptyset$. Then, $|\mathcal{F}| < f_{k-2}(\operatorname{lk}(1,K))$.

The following proof carries on the reduction mentioned above. When this is not possible we restrict the analysis to a facet. Once we have restricted the situation to a facet, we will exhibit a set that is in the star of an apex vertex but not in the pairwise-intersecting family. This will show that the star of the apex vertex is stricly larger. Proof of Theorem 66. Let $\mathcal{F} \subseteq K$ be the pairwise-intersecting family of maximal size satisfying the empty intersection condition, i.e., $\bigcap \mathcal{F} = \emptyset$. We iteratively apply shift_{w,v} to \mathcal{F} for $v \in N \setminus \{w\}$ as shown in Lemma 81² while the resulting family satisfies the empty intersection condition, otherwise we interrupt the process and output the previous family. Let \mathcal{H} be the resulting family from this process. Since K is a near-cone with apex vertex w we have that shift_{w,v}(K) = K for every $v \in N$, see subsection 6.2.2, and consequently $\mathcal{H} \subseteq K$. We proceed by case analysis.

Case 1: shift_{w,v}(\mathcal{H}) = \mathcal{H} for all $v \in N \setminus \{w\}$. Since $\cap \mathcal{H} = \emptyset$, there exists $F \in \mathcal{H}$ such that $w \notin F$. Because shift_{w,v}(\mathcal{H}) = \mathcal{H} for every $v \in N \setminus \{w\}$ we have that $(F \setminus \{v\}) \cup \{w\} \in \mathcal{H}$ for every $v \in F$. That is $\binom{F \cup \{w\}}{k} \subseteq \mathcal{H}$, and consequently $\beta_{k-1}(\langle \mathcal{H} \rangle) > 0$. By Corollary 91 we can conclude that $\binom{[k+1]}{k} \subseteq \Delta(\mathcal{H}) \subseteq \Delta(\mathcal{H})$. This implies that $\Delta(\mathcal{H})$ is a shifted pairwise-intersecting family that satisfies the empty intersection condition and it is contained in a shifted simplicial complex $\Delta(K)$ with minimal facet cardinality depth(K) + 1, see subsection 6.2.3. Since $k < \frac{\operatorname{depth}(K)+1}{2}$, by Corollary 101 and Corollary 93 we have that

$$|\mathcal{F}| = |\mathcal{H}| = |\Delta(\mathcal{H})| < f_{k-2}(\operatorname{lk}(1, \Delta(K))) = f_{k-2}(\operatorname{lk}(w, K)).$$

Case 2: there exist $v \in N \setminus \{w\}$ and $F \in \binom{N \setminus \{w,v\}}{k-1}$ such that $\cap \operatorname{shift}_{w,v}(\mathcal{H}) \neq \emptyset$ and $F \cup \{w\}, F \cup \{v\} \in \mathcal{H}$. On the one hand, since $\cap \operatorname{shift}_{w,v}(\mathcal{H}) \neq \emptyset$ we have that $T \cap \{w,v\} \neq \emptyset$ for every $T \in \mathcal{H}$. On the other hand, because $|\mathcal{H}|$ is maximal we have that

$$\{T \cup \{w, v\} \colon T \in \operatorname{lk}(\{w, v\}, K), \ |T| = k - 2\} \subseteq \mathcal{H}.$$

Then $\binom{F \cup \{w,v\}}{k} \subseteq \mathcal{H}$ because $F \cup \{v\}$ is contained in a facet of size strictly lager than k and consequently we can exchange any of the remaining vertices in this facet for w. Since K is a near-cone with apex vertex w, after this exchange we still have a face. The conclusion follows as in the previous case.

Case 3: there exist $v \in N \setminus \{w\}$ such that $\bigcap \operatorname{shift}_{w,v}(\mathcal{H}) \neq \emptyset$ and for every $F \in \binom{N \setminus \{v,w\}}{k-1}$ if $F \cup \{v\} \in \mathcal{H}$ then $F \cup \{w\} \notin \mathcal{H}$. In this case, we define the function

$$\phi \colon \mathcal{H} \to \{T \in \mathrm{lk}(w, K) \colon |T| = k - 1\}$$
$$F \mapsto \begin{cases} F \setminus \{w\} & \text{if } w \in F, \\ F \setminus \{v\} & \text{if } w \notin F. \end{cases}$$

The map ϕ is injective since if $F = F_1 \setminus \{v\} = F_2 \setminus \{w\}$ for some $F_1, F_2 \in \mathcal{H}$, then F contradicts the assumption of this case. We claim that ϕ is not surjective. The theorem will follow from this since $|\mathcal{H}| = |\operatorname{im} \phi| < f_{k-2}(\operatorname{lk}(w, K))$. Let us assume that ϕ is surjective. We split the analysis into two subcases.

Subcase a: there exist $T \in K$ and $F_v, F_w \in \mathcal{H}$ such that $w \in F_w, v \notin F_w, v \in F_v, w \notin F_v$ and $F_v, F_w \subseteq T$. Without loss of generality we can assume that T is a facet and let d = |T|. Then

$$H_T(w\bar{v}) = \{F \setminus \{w\} \subseteq T \colon F \in \mathcal{H}, w \in F, v \notin F\}$$

²We apply the lemma with $W = \{w\}$. Here W denotes the vertex set in the statement of Lemma 81.

and $H_T(\bar{w}v)$ are a pair of non-empty cross-intersecting families of (k-2)-faces in $T \setminus \{w, v\}$ and $|T \setminus \{w, v\}| = d - 2 \ge \operatorname{depth}(K) - 1$. Since $k < \frac{\operatorname{depth}(K) + 1}{2}$, then $(k-1) < \frac{\operatorname{depth}(K) - 1}{2} \le \frac{d-2}{2}$ and consequently by Theorem 57 we have that

$$\begin{aligned} |\mathcal{H}_T(w\bar{v})| + |\mathcal{H}_T(\bar{w}v)| &\leq \binom{d-2}{k-1} - \binom{(d-2) - (k-1)}{k-1} + 1 \\ &= \binom{d-2}{k-1} - \binom{d-k-1}{k-1} + 1 < \binom{d-2}{k-1}, \end{aligned}$$

where the last inequality follows since (d-2) > 2(k-1) and consequently $\binom{(d-2)-(k-1)}{k-1} > 1$. On the other hand, the total number of (k-2)-faces in $T \setminus \{w, v\}$ is $\binom{d-2}{k-1}$. Then, there exists some $F \in \binom{T \setminus \{w, v\}}{k-1} \setminus (\mathcal{H}_T(v\bar{w}) \cup \mathcal{H}_T(w\bar{v}))$ and consequently $F \notin \operatorname{im} \phi$. This is a contradiction with ϕ being surjective.

Subcase b: for every $T \in K$ containing v and w either $\mathcal{H}_T(w\bar{v}) \neq \emptyset$ or $\mathcal{H}_T(\bar{w}v) \neq \emptyset$, but not both simultaneously. Since $\cap \mathcal{H} = \emptyset$ there exist $F_w, F_v \in \mathcal{H}$ such that $w \in F_w, v \notin F_w$ and $v \in F_v, w \notin F_v$. Since $k < \frac{\operatorname{depth}(K)+1}{2}$, there exist a facet T containing F_v and w and set d = |T|. Since $F_v \subseteq T$ then $\mathcal{H}_T(w\bar{v}) = \emptyset$ and consequently $F_w \notin T$. Since

$$|(T \setminus \{w,v\}) \setminus (F_w \setminus \{w\})| \ge (d-2) - (k-1) = d-k-1 \ge \operatorname{depth}(K) - k > k-1$$

there exists $G \subseteq (T \setminus \{w, v\}) \setminus F_w$ with |G| = k - 1. On the one hand, since ϕ is surjective, then $G \in \operatorname{im} \phi$. On the other hand, since $\mathcal{H}_T(w\bar{v}) = \emptyset$, then $G \cup \{v\} \in \mathcal{H}$. But $(G \cup \{v\}) \cap F_w = \emptyset$ which contradicts \mathcal{H} being pairwise-intersecting.

As a corollary we obtain Theorem 63.

Proof of Theorem 63. Since K is a sequentially Cohen-Macaulay simplicial complex with minimal facet cardinality d, it follows from the discussion at the end of subsection 6.2.3 that depth(K) + 1 = d. Since $k < d/2 = \frac{\text{depth}(K)+1}{2}$, we can conclude the proof by applying Theorem 66.

6.5 Cross-intersecting families

In this section we extend Theorem 57 to simplicial complexes. First we show that Theorem 65 holds for shifted simplicial complexes. In particular we give a different proof from the one by Borg [17].

Proposition 102. Let $k \leq d/2$. Let K be a shifted simplicial complex with vertex set [n] and minimal facet cardinality d. Let $\mathcal{F}, \mathcal{G} \subseteq K$ be a pair of non-empty cross-intersecting families of (k-1)-faces, then

$$|\mathcal{F}| + |\mathcal{G}| \le f_{k-1}(K) - f_{k-1}(K[[n] \setminus [k]]) + 1.$$

Proof. By Theorem 89 we have that $\Delta(\mathcal{F})$, $\Delta(\mathcal{G}) \subseteq \Delta(K) = K$ are a pair of non-empty shifted cross-intersecting families. We proceed by induction on n and k using that the claim holds if the underlying simplicial complex is the simplex. Set $\mathcal{F}(n) = \{F \setminus \{n\} \colon F \in \Delta(\mathcal{F}), n \in F\}$ and $\mathcal{F}(\bar{n}) = \{F \in \Delta(\mathcal{F}) \colon n \notin F\}$.
We define analogously $\mathcal{G}(n)$ and $\mathcal{G}(\bar{n})$. First we claim that $\mathcal{F}(n)$ and $\mathcal{G}(n)$ are cross-intersecting families of (k-2)-faces contained in $\mathrm{lk}(n, K)$. If not, then there exist $F \in \Delta(\mathcal{F})$ and $G \in \Delta(\mathcal{G})$ such that $F \cap G = \{n\}$. Since $2k \leq d$ there exist a facet $T \in K$ strictly containing F. Since $|F \cup G| < 2k \leq d$, then there exist $i \in T \setminus (F \cup G)$. Because $\Delta(\mathcal{F})$ is shifted we have that $(F \setminus \{n\}) \cup \{i\} \in \Delta(\mathcal{F})$. But, the face $(F \setminus \{n\}) \cup \{i\}$ is disjoint from G, which contradicts the crossintersecting assumption. Then $\mathcal{F}(n)$ and $\mathcal{G}(n)$ are cross-intersecting families of (k-2)-faces in $\mathrm{lk}(n, K)$. By induction we can conclude that

$$|\mathcal{F}(n)| + |\mathcal{G}(n)| \le f_{k-2}(\operatorname{lk}(n,K)) - f_{k-2}(\operatorname{lk}(n,K)[[n-1] \setminus [k-1]]) + 1.$$

Since $\mathcal{F}(\bar{n})$ and $\mathcal{G}(\bar{n})$ are cross-intersecting families of (k-1)-faces in del(n, K) we can conclude by induction that

$$|\mathcal{F}(\bar{n})| + |\mathcal{G}(\bar{n})| \le f_{k-1}(\operatorname{del}(n,K)) - f_{k-1}(\operatorname{del}(n,K)[[n-1] \setminus [k]]) + 1.$$

The conclusion will follow from the fact that $f_{k-2}(\operatorname{lk}(n, K)) + f_{k-1}(\operatorname{del}(n, K)) = f_{k-1}(K)$ and the following inequality

$$f_{k-1}(K[[n] \setminus [k]]) \le f_{k-2}(\operatorname{lk}(n, K)[[n-1] \setminus [k-1]]) + f_{k-1}(\operatorname{del}(n, K)[[n-1] \setminus [k]]) - 1.$$

To verify this last inequality consider a (k-1)-face $F \in K[[n] \setminus [k]]$, if $n \in F$, then

$$F \setminus \{n\} \in \operatorname{lk}(n, K)[[n-1] \setminus [k]] \subseteq \operatorname{lk}(n, K)[[n-1] \setminus [k-1]].$$
(6.1)

If $n \notin F$, then $F \in del(n, K)[[n-1] \setminus [k]]$. Moreover, the right hand side of (6.1) also contains the (k-2)-face $[k, 2k-2] \in lk(n, K)[[n-1] \setminus [k-1]]$. This is because $[k, 2k-2] \subseteq [d-1] \cup \{n\} \in K$ since K is shifted with minimal facet cardinality d. Substracting this face gives us the desired inequality. \Box

Proof of Theorem 68. By Theorem 89 we know that $\Delta(\mathcal{F})$ and $\Delta(\mathcal{G})$ are a pair of non-empty cross-intersecting families of (k-1)-faces in $\Delta(K)$. This last one is a shifted simplicial complex with minimal facet cardinality depth(K) + 1. By Proposition 102 we have that

$$\mathcal{F}|+|\mathcal{G}| = |\Delta(\mathcal{F})|+|\Delta(\mathcal{G})|$$

$$\leq f_{k-1}(\Delta(K)) - f_{k-1}(\Delta(K)[[n] \setminus [r]]) + 1$$

$$= f_{k-1}(K) - f_{k-1}(K[N \setminus W]) + 1,$$

where the last equality follows from Proposition 94.

As an application we obtain a proof of Theorem 65.

Proof of Theorem 65. Since K is a sequentially Cohen-Macaulay simplicial complex with minimal facet cardinality d, it follows from the discussion at the end of subsection 6.2.3 that depth(K) + 1 = d. Since $k \leq d/2 = \frac{\text{depth}(K)+1}{2}$, we can conclude the proof by applying Theorem 68.

Bibliography

- [1] K. A. Adiprasito. Combinatorial lefschetz theorems beyond positivity. arXiv preprint arXiv:1812.10454, 2018.
- [2] N. Alon. An extremal problem for sets with applications to graph theory. J. Combin. Theory Ser. A, 40(1):82–89, 1985.
- [3] N. Alon and G. Kalai. A simple proof of the upper bound theorem. *European J. Combin.*, 6(3):211–214, 1985.
- [4] N. Alon, G. Kalai, J. Matoušek, and R. Meshulam. Transversal numbers for hypergraphs arising in geometry. Adv. in Appl. Math., 29(1):79–101, 2002.
- [5] L. Asimow and B. Roth. The rigidity of graphs. Trans. Amer. Math. Soc., 245:279–289, 1978.
- [6] L. Asimow and B. Roth. The rigidity of graphs. II. J. Math. Anal. Appl., 68(1):171–190, 1979.
- [7] E. Babson and I. Novik. Face numbers and nongeneric initial ideals. *Electron. J. Combin.*, 11(2):Research Paper 25, 23, 2004/06.
- [8] J. Balogh, B. Bollobás, R. Morris, and O. Riordan. Linear algebra and bootstrap percolation. J. Combin. Theory Ser. A, 119(6):1328–1335, 2012.
- [9] J. Balogh and G. Pete. Random disease on the square grid. In Proceedings of the Eighth International Conference "Random Structures and Algorithms" (Poznan, 1997), volume 13, pages 409–422, 1998.
- [10] I. Bárány. A generalization of Carathéodory's theorem. Discrete Math., 40(2-3):141–152, 1982.
- [11] I. Bárány, F. Fodor, L. Montejano, D. Oliveros, and A. Pór. Colourful and fractional (p, q)-theorems. Discrete Comput. Geom., 51(3):628–642, 2014.
- [12] D. Barnette. Generating the triangulations of the projective plane. J. Combin. Theory Ser. B, 33(3):222–230, 1982.
- [13] D. W. Barnette and A. Edelson. All orientable 2-manifolds have finitely many minimal triangulations. *Israel J. Math.*, 62(1):90–98, 1988.
- [14] D. W. Barnette and A. L. Edelson. All 2-manifolds have finitely many minimal triangulations. *Israel J. Math.*, 67(1):123–128, 1989.
- [15] B. Bollobás. Weakly k-saturated graphs. In Beiträge zur Graphentheorie (Kolloquium, Manebach, 1967), pages 25–31. Teubner, Leipzig, 1968.
- [16] P. Borg. Extremal t-intersecting sub-families of hereditary families. J. Lond. Math. Soc. (2), 79(1):167–185, 2009.
- [17] P. Borg. On cross-intersecting uniform sub-families of hereditary families. *Electron. J. Combin.*, 17(1):Research Paper 60, 10, 2010.

- [18] M. Borowiecki and E. Sidorowicz. Weakly P-saturated graphs. volume 22, pages 17–29. 2002. Conference on Graph Theory (Elgersburg, 2000).
- [19] D. Bulavka, A. Goodarzi, and M. Tancer. Optimal bounds for the colorful fractional Helly theorem. In K. Buchin and É. Colin de Verdière, editors, 37th International Symposium on Computational Geometry (SoCG 2021), volume 189 of Leibniz International Proceedings in Informatics (LIPIcs), pages 19:1–19:14, Dagstuhl, Germany, 2021. Schloss Dagstuhl – Leibniz-Zentrum für Informatik.
- [20] D. Bulavka, E. Nevo, and Y. Peled. Volume rigidity and algebraic shifting. In 35th Conference on Formal Power Series and Algebraic Combinatorics (FPSAC23), volume 89B of Séminaire Lotharingien de Combinatorie, 2023.
- [21] D. Bulavka and R. Woodroofe. Strict Erdős-Ko-Rado for simplicial complexes. In European Conference on Combinatorics, Graph Theory and Applications 2023 (Eurocomb'23), Proceedings of the 12th European Conference on Combinatorics, Graph Theory and Applications, pages 245–253, 2023.
- [22] Denys Bulavka, Martin Tancer, and Mykhaylo Tyomkyn. Weak Saturation of Multipartite Hypergraphs. *Combinatorica*, July 2023.
- [23] J. Cruickshank, B. Jackson, and S. Tanigawa. Global rigidity of triangulated manifolds. arXiv preprint arXiv:2204.02503, 2022.
- [24] A. Dochtermann and A. Engström. Algebraic properties of edge ideals via combinatorial topology. *Electron. J. Combin.*, 16(2, Special volume in honor of Anders Björner):Research Paper 2, 24, 2009.
- [25] A. M. Duval. Algebraic shifting and sequentially Cohen-Macaulay simplicial complexes. *Electron. J. Combin.*, 3(1):Research Paper 21, approx. 14, 1996.
- [26] J. Eckhoff. An upper-bound theorem for families of convex sets. Geom. Dedicata, 19(2):217–227, 1985.
- [27] P. Erdős, Z. Füredi, and Z. Tuza. Saturated r-uniform hypergraphs. Discrete Math., 98(2):95–104, 1991.
- [28] P. Erdős, Chao Ko, and R. Rado. Intersection theorems for systems of finite sets. Quart. J. Math. Oxford Ser. (2), 12:313–320, 1961.
- [29] Sara Faridi. Simplicial trees are sequentially Cohen-Macaulay. J. Pure Appl. Algebra, 190(1-3):121–136, 2004.
- [30] R. J. Faudree and R. J. Gould. Weak saturation numbers for multiple copies. Discrete Math., 336:1–6, 2014.
- [31] R. J. Faudree, R. J. Gould, and M. S. Jacobson. Weak saturation numbers for sparse graphs. *Discuss. Math. Graph Theory*, 33(4):677–693, 2013.
- [32] A. L. Fogelsanger. The generic rigidity of minimal cycles. ProQuest LLC, Ann Arbor, MI, 1988. Thesis (Ph.D.)–Cornell University.

- [33] Christopher A. Francisco and Adam Van Tuyl. Sequentially Cohen-Macaulay edge ideals. *Proc. Amer. Math. Soc.*, 135(8):2327–2337, 2007.
- [34] P. Frankl. An extremal problem for two families of sets. European J. Combin., 3(2):125–127, 1982.
- [35] P. Frankl. The shifting technique in extremal set theory. In Surveys in combinatorics 1987 (New Cross, 1987), volume 123 of London Math. Soc. Lecture Note Ser., pages 81–110. Cambridge Univ. Press, Cambridge, 1987.
- [36] P. Frankl. A simple proof of the Hilton-Milner theorem. Mosc. J. Comb. Number Theory, 8(2):97–101, 2019.
- [37] F. R. Gantmacher. The theory of matrices. Vol. 1. AMS Chelsea Publishing, Providence, RI, 1998. Translated from the Russian by K. A. Hirsch, Reprint of the 1959 translation.
- [38] E. Helly. Uber mengen konvexer Körper mit gemeinschaftlichen Punkten. Jahresber. Deustch. Math.-Verein., 32:175–176, 1923.
- [39] E. Helly. Uber Systeme von abgeschlossenen Mengen mit gemeinschaftlichen Punkten. Monatsh. Math. Phys., 37(1):281–302, 1930.
- [40] A. J. W. Hilton and E. C. Milner. Some intersection theorems for systems of finite sets. Quart. J. Math. Oxford Ser. (2), 18:369–384, 1967.
- [41] F. Holroyd, C. Spencer, and J. Talbot. Compression and Erdős-Ko-Rado graphs. Discrete Math., 293(1-3):155–164, 2005.
- [42] F. Holroyd and J. Talbot. Graphs with the Erdős-Ko-Rado property. Discrete Math., 293(1-3):165–176, 2005.
- [43] G. Hurlbert and V. Kamat. Erdős-Ko-Rado theorems for chordal graphs and trees. J. Combin. Theory Ser. A, 118(3):829–841, 2011.
- [44] G. Hurlbert and V. Kamat. New injective proofs of the Erdős-Ko-Rado and Hilton-Milner theorems. *Discrete Math.*, 341(6):1749–1754, 2018.
- [45] A. Johnson and F. C. Holroyd. 'Problem 25', Problems from the Sixteenth British Combinatorial Conference. P.J. Cameron, 1997.
- [46] G. Kalai. Intersection patterns of convex sets. Israel J. Math., 48(2-3):161– 174, 1984.
- [47] G. Kalai. Weakly saturated graphs are rigid. In Convexity and graph theory (Jerusalem, 1981), volume 87 of North-Holland Math. Stud., pages 189–190. North-Holland, Amsterdam, 1984.
- [48] G. Kalai. Hyperconnectivity of graphs. Graphs Combin., 1(1):65–79, 1985.
- [49] G. Kalai. Symmetric matroids. J. Combin. Theory Ser. B, 50(1):54–64, 1990.

- [50] G. Kalai. Algebraic shifting. In Computational commutative algebra and combinatorics (Osaka, 1999), volume 33 of Adv. Stud. Pure Math., pages 121–163. Math. Soc. Japan, Tokyo, 2002.
- [51] G. Kalai and R. Meshulam. A topological colorful Helly theorem. Adv. Math., 191(2):305–311, 2005.
- [52] M. Katchalski and A. Liu. A problem of geometry in \mathbb{R}^n . Proc. Amer. Math. Soc., 75(2):284–288, 1979.
- [53] M. Kim. A note on the colorful fractional Helly theorem. *Discrete Math.*, 340(1):3167–3170, 2017.
- [54] G. Kronenberg, T. Martins, and N. Morrison. Weak saturation numbers of complete bipartite graphs in the clique. J. Combin. Theory Ser. A, 178:105357, 15, 2021.
- [55] S. A. Lavrenchenko. Irreducible triangulations of the torus. Journal of Soviet Mathematics, 51(5):2537–2543, October 1990.
- [56] S. Lawrencenko and S. Negami. Irreducible triangulations of the Klein bottle. J. Combin. Theory Ser. B, 70(2):265–291, 1997.
- [57] C. W. Lee. Generalized stress and motions. In Polytopes: abstract, convex and computational (Scarborough, ON, 1993), volume 440 of NATO Adv. Sci. Inst. Ser. C: Math. Phys. Sci., pages 249–271. Kluwer Acad. Publ., Dordrecht, 1994.
- [58] M. Loréa. On matroidal families. *Discrete Math.*, 28(1):103–106, 1979.
- [59] L Lovász. Problem 206. Matematikai Lapok, 25:181, 1974.
- [60] L. Lovász. Flats in matroids and geometric graphs. In Combinatorial surveys (Proc. Sixth British Combinatorial Conf., Royal Holloway Coll., Egham, 1977), pages 45–86, 1977.
- [61] E. Lubetzky and Y. Peled. The Threshold for Stacked Triangulations. *International Mathematics Research Notices*, 2022. https://doi.org/10.1093/imrn/rnac276.
- [62] S. Morey and R. H. Villarreal. Edge ideals: algebraic and combinatorial properties. In *Progress in commutative algebra 1*, pages 85–126. de Gruyter, Berlin, 2012.
- [63] N. Morrison and J. A. Noel. Extremal bounds for bootstrap percolation in the hypercube. J. Combin. Theory Ser. A, 156:61–84, 2018.
- [64] G. Moshkovitz and A. Shapira. Exact bounds for some hypergraph saturation problems. J. Combin. Theory Ser. B, 111:242–248, 2015.
- [65] S. Murai. Algebraic shifting of cyclic polytopes and stacked polytopes. Discrete Math., 307(14):1707–1721, 2007.

- [66] E. Nevo. Algebraic shifting and basic constructions on simplicial complexes. J. Algebraic Combin., 22(4):411–433, 2005.
- [67] E. Nevo. Algebraic shifting and f-vector theory. *PhD thesis, arXiv preprint* arXiv:0709.3265, 2007.
- [68] J. A. Olarte, F. Santos, J. Spreer, and C. Stump. The EKR property for flag pure simplicial complexes without boundary. J. Combin. Theory Ser. A, 172:105205, 29, 2020.
- [69] O. Pikhurko. Uniform families and count matroids. *Graphs Combin.*, 17(4):729–740, 2001.
- [70] O. Pikhurko. Weakly saturated hypergraphs and exterior algebra. Combin. Probab. Comput., 10(5):435–451, 2001.
- [71] J. Scott Provan and Louis J. Billera. Decompositions of simplicial complexes related to diameters of convex polyhedra. *Math. Oper. Res.*, 5(4):576–594, 1980.
- [72] A. Rosén. Geometric multivector analysis. Birkhäuser Advanced Texts: Basler Lehrbücher. Birkhäuser/Springer, Cham, 2019. From Grassmann to Dirac.
- [73] G. Semanišin. On some variations of extremal graph problems. Discuss. Math. Graph Theory, 17(1):67–76, 1997.
- [74] E. Sidorowicz. Size of weakly saturated graphs. Discrete Math., 307(11-12):1486–1492, 2007.
- [75] D. E. Smith. On the Cohen-Macaulay property in commutative algebra and simplicial topology. *Pacific J. Math.*, 141(1):165–196, 1990.
- [76] R. P. Stanley. The upper bound conjecture and Cohen-Macaulay rings. Studies in Appl. Math., 54(2):135–142, 1975.
- [77] I. Streinu and L. Theran. Algorithms for area and volume rigidity. In *Proceedings of the 17th Fall Workshop on Computational Geometry*, 2007.
- [78] I. Streinu and L. Theran. Sparse hypergraphs and pebble game algorithms. European J. Combin., 30(8):1944–1964, 2009.
- [79] I. Streinu and L. Theran. Natural realizations of sparsity matroids. Ars Math. Contemp., 4(1):141–151, 2011.
- [80] T. Sulanke. Note on the irreducible triangulations of the Klein bottle. J. Combin. Theory Ser. B, 96(6):964–972, 2006.
- [81] J. Talbot. Intersecting families of separated sets. J. London Math. Soc. (2), 68(1):37–51, 2003.
- [82] M. Tancer. Intersection patterns of convex sets via simplicial complexes: A survey. *Thirty essays on geometric graph theory*, pages 521–540, 2013.

- [83] Z. Tuza. A generalization of saturated graphs for finite languages. In Proceedings of the 4th international meeting of young computer scientists, IMYCS '86 (Smolenice Castle, 1986), number 185, pages 287–293, 1986.
- [84] Z. Tuza. Extremal problems on saturated graphs and hypergraphs. volume 25, pages 105–113. 1988. Eleventh British Combinatorial Conference (London, 1987).
- [85] Z. Tuza. Asymptotic growth of sparse saturated structures is locally determined. volume 108, pages 397–402. 1992. Topological, algebraical and combinatorial structures. Frolik's memorial volume.
- [86] Adam Van Tuyl and Rafael H. Villarreal. Shellable graphs and sequentially Cohen-Macaulay bipartite graphs. J. Combin. Theory Ser. A, 115(5):799– 814, 2008.
- [87] G. Wegner. d-collapsing and nerves of families of convex sets. Arch. Math. (Basel), 26:317–321, 1975.
- [88] W. Whiteley. La division de sommet dans les charpentes isostatiques. Structural Topology, (16):23–30, 1990. Dual French-English text.
- [89] W. Whiteley. Some matroids from discrete applied geometry. In Matroid theory (Seattle, WA, 1995), volume 197 of Contemp. Math., pages 171–311. Amer. Math. Soc., Providence, RI, 1996.
- [90] R. Woodroofe. Vertex decomposable graphs and obstructions to shellability. Proc. Amer. Math. Soc., 137(10):3235–3246, 2009.
- [91] R. Woodroofe. Erdős-Ko-Rado theorems for simplicial complexes. J. Combin. Theory Ser. A, 118(4):1218–1227, 2011.

List of publications

- [19] D. Bulavka, A. Goodarzi, and M. Tancer. Optimal bounds for the colorful fractional Helly theorem. In K. Buchin and É. Colin de Verdière, editors, 37th International Symposium on Computational Geometry (SoCG 2021), volume 189 of Leibniz International Proceedings in Informatics (LIPIcs), pages 19:1–19:14, Dagstuhl, Germany, 2021. Schloss Dagstuhl – Leibniz-Zentrum für Informatik.
- [20] D. Bulavka, E. Nevo, and Y. Peled. Volume rigidity and algebraic shifting. In 35th Conference on Formal Power Series and Algebraic Combinatorics (FPSAC23), volume 89B of Séminaire Lotharingien de Combinatoire, 2023.
- [21] D. Bulavka and R. Woodroofe. Strict Erdős-Ko-Rado for simplicial complexes. In *European Conference on Combinatorics, Graph Theory and Applications 2023 (Eurocomb'23)*, Proceedings of the 12th European Conference on Combinatorics, Graph Theory and Applications, pages 245–253, 2023.
- [22] D. Bulavka, M. Tancer, and M. Tyomkyn. Weak Saturation of Multipartite Hypergraphs. *Combinatorica*, July 2023.